# CHEMICAL PARTITIONING OF TRACE METALS IN SEDIMENTS OF A TROPICAL ESTUARY

A THESIS

SUBMITTED TO THE COCHIN UNIVERSITY OF SCIENCE AND TECHNOLOGY IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF

**DOCTOR OF PHILOSOPHY** 

IN

ENVIRONMENTAL CHEMISTRY UNDER THE FACULTY OF MARINE SCIENCES

C. KUNHIKRISHNAN NAIR

CHEMICAL OCEANOGRAPHY DIVISION - SCHOOL OF MARINE SCIENCES COCHIN UNIVERSITY OF SCIENCE AND TECHNOLOGY COCHIN - 682 016

January 1992

COCHIN UNIVERSITY OF SCIENCE AND TECHNOLOGY

CHEMICAL OCEANOGRAPHY DIVISION. SCHOOL OF MARINE SCIENCES FINE ARTS AVENUE



COCHIN 682 016, INDIA

Dr. JACOB CHACKO READER Tel. 0484-366478 (Off.) 0484-855699 (Res.) Telex: (0885) 5019 CU-IN

### CERTIFICATE

This is to certify that this thesis is a bonafide record of the research carried out by Shri.C.Kunhikrishnan Nair, initially under the guidance of the late Prof. (Dr) P.N.K. Nambisan, and presently under my guidance, in partial fulfilment of the requirements for the degree of **Philosophiae Doctor** of the Cochin University of Science and Technology.

Cochin - 682 016 January, 1992

Jacob Chacko

Supervising Teacher

### PREFACE

Chemical analysis of sediments is one of the reliable techniques employed for the monitoring of metals in aquatic To-date a number of reports are available on environments. the baseline values of `total metals' from different environments. Concurrently consistent efforts have been directed at developing techniques that could yield better information. Sequential chemical extractions that provide better insight into the environmental behaviour of trace metals by quantifying sediment bound trace metals susceptible to short-term processes, have caught the attention of many scientists. This thesis reports the attempts made by the author to explore the possibilities of a sequential chemical extraction scheme in delineating the behaviour of sediment bound trace metals, in the highly dynamic environments of a tropical estuary.

The thesis is presented in five chapters.

CHAPTER 1, the introduction, reviews the historical development of chemical partitioning technique as an effective tool in trace metal monitoring programmes. An update on scientific information about trace metals in the Cochin estuary is also provided together with the aims and objectives of the present investigation.

CHAPTER 2 describes the study area and details the materials and methods employed in this study.

CHAPTER 3 presents the data recorded on the physical and chemical parameters of the sediments such as moisture content, texture, organic carbon content and total trace metal concentration along with results and discussion. The seasonal and spatial variations observed have been critically analysed in relation to the varying estuarine conditions.

CHAPTER 4 presents the results and discussion on chemical partitioning of trace metals. The seasonal and spatial variations in distribution of the various fractions of iron, manganese, zinc, copper, cadmium, lead, cobalt, nickel and chromium have been discussed individually and the variations explained.

CHAPTER 5 deals with the interrelationship between metals, metal fractions and environmental parameters. Metal enrichment factors have also been worked out phase-wise. A critical commentary on the role of chemical partitioning of trace metals in the tropical estuarine environment is also presented. The results of the present investigation have been/ are being published, as given below.

 i) Heavy metal speciation in sediments of Cochin estuary determined using chemical extraction techniques. Sci. Total Environ., 102 (2), 113 - 128.

ii) Speciation of Mn, Fe, Cr, Ni, Zn and Co in the sediments of a tropical estuary (communicated to Environmental Geology and Water Science).

iii) Seasonal and spatial distribution of chemically determinable species of trace metals in sediments of Cochin estuary. (under preparation)

iv) An investigation on total trace metals in sediments of the Cochin estuary.(under preparation)

v) State of art in metal speciation - new perspectives. (under preparation)

vi) Exchangeable and organically bound trace metal levels in sediments in relation to estuarine chemical reactivity.(under preparation)

vii) Implications of metal fractionation in sediments strategies in assessing trace metal levels in aquatic
environments. (under preparation)

## CONTENTS

•

CHAPTER 1	
INTRODUCTION	1
CHAPTER 2	
MATERIALS AND METHODS	17
CHARTER A	
PHYSICO-CHEMICAL CHARACTERISTICS OF SEDIMENTS	23
CHAPTER 4	
TRACE METAL PARTITIONING IN SEDIMENTS	44
CHAPTER 5	
CHEMICAL PARTITIONING -	
IMPLICATIONS IN A TROPICAL ESTUARY	<b>1</b> 01
DEEEDENGEG	121
	IGno i

## CHAPTER 1 INTRODUCTION

### 1.1 TRACE METALS

- **1.2 METAL ASSOCIATION IN SEDIMENTS**
- **1.3 CHEMICAL PARTITIONING**
- **1.4 ESTUARINE VARIABILITY**
- 1.5 THE COCHIN ESTUARY
- 1.6 SCOPE OF THE PRESENT STUDY

1.1 TRACE METALS

The term "trace metal" is widely recognised and used to denote those metals that are distributed in earth's crust in very low concentrations. The popular elements of this group are cadmium, chromium, cobalt, copper, lead, nickel and zinc. Often, iron and manganese also find a place in the study of trace metals, as the environmental chemistry of the trace metals is closely related to that of iron and manganese (Jenne, 1968). Industrially important, these metals and their compounds find a variety of uses in all fields of human endeavour, like agriculture, engineering, health, hygiene and so on. Nriagu and Pacyna (1988) have recently reviewed the trends in the production of several of such industrially important metals and have concluded that there has been a significant increase, during the past 55 year period (1930-85), in the amount of the different metals produced. The increase in production and supply of trace metals is generally viewed with great concern as it has been established that most of these trace metals are potentially toxic to living organisms.

Trace metals are known to play a dual role in living organisms. Many of these elements are essential for the sustenance of life, although required only in very small quantities. At higher levels of uptake, most of these metals The difference between the nutritive function turn toxic. and toxicity is only a matter of variations in concentration of the metals in the organisms and generally the difference is large enough to cause any practical concern (Waldichuk, 1985). However, there are some metals like cadmium and lead whose essentiality is not yet known but toxicity is well established even at very low concentrations.

One oft-quoted property of the trace metals is their persistence in aquatic environments. Once the metal is released into a system, it remains there for relatively long periods as it cannot be destroyed or disintegrated. Although the form in which it remains in the system may vary depending on the biogeochemical processes prevalent in the system, the total metal content would be the same unless and until it is transported to another regime.

The tragedies of Minamata and Niigata in Japan [where over 50 mortalities and over 100 permanently disabled victims of mercury poisoning caused by the consumption of mercury

contaminated shellfish and finfish (Goldwater, 1971) were reported] served to arouse serious concern over the `health' of oceans and in particular over metal pollution in the aquatic environment. Public concern over metal-toxicity further aggravated with the reports that the affliction known as "itai-itai", again reported from Japan, was caused by the consumption of food contaminated by cadmium (Ui, 1972). There were reports on the toxic effects of other metals also, especially that of lead (Chisolm, 1971), which entered the atmosphere from smelters and through the use of tetraethyl lead gasoline. These reports have provided a stimulus for in initiating more comprehensive studies aimed at identifying known or suspected areas of pollution and for complementary investigations of the chemical behaviour of trace metals in soils, sediments and water (Wittmann, 1979; Fowler, 1990).

### 1.2 METAL ASSOCIATION IN SEDIMENTS

Processing of ores and production of metals lead to the generation of metal containing wastes that get redistributed in the environment under the influence of the natural agents -Eventually, under the influence of wind and water. the hydrological cycles these metals find their way to the surface waters (Forstner, 1979a). Transport and fate of trace metals depend greatly on the physical and chemical factors controlling Trace metals from a number of non-point diffusive the system. sources, such as, geologically weathered products, urban and rural runoff, domestic sewage, etc. are transported to surface the unconsolidated accumulation waters. The sediment. of particles covering the floor area of the waterbody usually acts as a repository for these trace metals. In sediments, trace metals are associated within a variety of accumulative phases. Gibbs (1973) suggested four groups of metal associations; later workers (Forstner and Patchineelam, 1976; Engler et al. 1977; Jenne and Luoma, 1977; Salomons and Forstner; 1980; Harrison et

al. 1981) expanded this categorisation to include all main types of metal associations as may occur in sediments, soils and deposited dusts. Broadly, five categories of accumulative phases are recognised:

- i) adsorptive and exchangeable phase
- ii) carbonate bound phase
- iii) easily reducible phase (Fe/Mn oxide phase)
- iv) organic/ sulfide bound phase and
- v) detrital or residual phase.

Characterisation of the distribution of trace metals among the various accumulative phases of the sedimentary particles is of great interest to sediment and water chemists for a number of reasons (Bernhard et al. 1986). A quantitative knowledge of the selective distribution of trace metals in sediments can be of value in assessing the potential impact of sediment re-suspension on water quality (Brannon et al. 1977). The selective distribution of trace metals may also be helpful in evaluating the relative influence of these metals on biological communities (Luoma and Jenne, 1976). By studying the distribution of metals between the different sediment phases, their contamination risks can be ascertained because the more mobile fractions introduced by man's activities remain bound to the exchangeable, the carbonate and the easily reducible phases (Calmano and Forstner, 1983; Forstner, 1985; Chester et al. 1985; Samanidou and Fytianos, 1987; Yoshimura et al.1988; Elsokkary and Muller, 1990). The metal levels in the inert fraction, being detrital and of lattice origin, can be taken as a measure of the geochemical contamination of the sediment sample as proposed by Salomons and Forstner (1980). The differentiation of the accumulative phases has been particularly useful in recognising the diagenetic reactions in deep sea sediments (Goldberg and Arrhenius, 1958; Arrhenius and Korkish, 1959; Chester and Hughes, 1967 & 1969; Heath and Dymond, 1977; Rapin, 1983; Kumar and Roonwal, 1987; Chester et al. 1988).

Though the significance of identifying the accumulative phases has been well recognised, physical methods of suffer serious limitations because of identification the extremely low quantities present in the matrix. Toxic metals generally exist, even in polluted sediments and soils. at concentrations rarely exceeding 1000 mg/kg. Their existence as a discrete mineral phase is unlikely except when they result from dispersion from an area of mineralization and even if present in this form their abundance is too low to identify by presently available techniques (Harrison, 1987). Therefore, the best approach to obtain information on those fractions that are relevant for geochemical/biological processes is to separate the various chemically extractable fractions (Nissenbaum, 1972; Agemian and Chau, 1977; Waldichuk, 1985: Tessier and Campbell, 1987; Luoma, 1989, Ure, 1990; Dudka and Chlopecka, 1990).

### 1.3 CHEMICAL PARTITIONING

The use of chemical extractants to quantify the element in a particular solid phase was originally attempted in soil studies (Jackson, 1958; Jenne et al. 1986; Bermond and Sommer, 1989), where the "pool" of nutrients of different solubilities and mobilities were selectively sampled and determined using extractants of different strengths. Because of the similarities between sediments and soil, extraction procedures used in soils have often been borrowed or adapted for sediment analyses (Tessier et al. 1979). A number of extraction procedures, varying in manipulative complexity, have been proposed for the partitioning of the metal phase. Some of these techniques (Agemian and Chau, 1976; Loring, 1976; Malo, 1977) employ a single extractant and are designed to effect separation between residual and non-residual metals. These methods are rapid and simple and afford a better contrast between the anomalous and normal samples than the determination of total

metal concentrations. The sequential extraction procedures (Nissenbaum, 1972; Presley et al. 1972; Gibbs, 1973; Gupta and Chen, 1975; Luoma and Jenne, 1976; Engler et al. 1977; Tessier et al. 1979; Harrison et al. 1981) employ a series of extractions with reagents of increasing power, under specified conditions, so that the individual phases can be more precisely isolated. The scheme of sequential extractions, although more time-consuming, provides detailed information on the origin, mode of occurrence, biological and physico-chemical availability, mobilization and transport of trace metals (Tessier et al. 1979; Campbell and Tessier, 1989; Gunn et al. 1989: Teraoka and Nakashima, 1990). The various extractants employed hitherto have been compiled and presented in Table 1.1.

Though the differentiated analysis in a sequential chemical extraction provides much more information than that from single extraction the extraction schemes are not free of flaws (Van Valin and Morse, 1982; Etcheber and Bourg, 1983; Rapin et al. 1986; Kheboian and Bauer, 1987). Transformation of phases during sample preparation (Jenne et al. 1980; Thompson et al. 1980; Trefry and Metz, 1984; Kersten and Forstner, 1987), re-adsorption and sometimes precipitation of metals (Rendell et al. 1980; Belzile et al. 1989), variability introduced by the duration, temperature, sequence of extraction and solid extractant ratio (Calmano and Forstner, 1983; Miller et al. 1986), failure to validate the result with well defined samples (Nirel and Morel, 1990) etc. are some of the problems cited in the literature. Selectivity of some of the reagents being used is also in doubt (Engler et al. 1977; Nirel and Morel, 1990; Ure, 1990). The use of barium chloride - triethanolamine buffer or magnesium chloride for the determination of exchangeable metals is prone to interferences arising out of the formation of organic chelates in the former case and of chloro-complexes (Calmano and Forstner, 1983) in the latter case. Substantial precipitation of trace metals, presumably due to the formation

5		
	Phase/Reagent	References
A.	EXCHANGEABLE PHASE	
a)	Ammonium acetate	Calmano and Forstner, 1983; Chapman, 1965; Engler et al. 1977; Gupta and Chen, 1975; Hong and Forstner, 1983; Jackson, 1958; Lee, 1985; Rapin, 1983; Samanidou and Fytianos, 1987; Stenekar et al. 1988; Teraoka and Nakashima, 1990; Yoshimura et al. 1988.
( q	Sodium acetate	Chapman, 1965
c)	Magnesium chloride	Bradley and Cox, 1990; Elliot et al. 1990; Gibbs, 1973; Lum et al. 1982; Tessier et al. 1979.
(p	Barium chloride- Triethanolamine	Forstner and Stoffers, 1981; Jackson, 1958
в.	CARBONATE PHASE	
a)	Sodium acetate- acetic acid	Bradley and Cox, 1990; Elliot et al. 1990; Lum et al. 1982; Nissenbaum, 1972; Samanidou and Fytianos, 1987; Stenekar et al. 1988; Tessier et al. 1979.
( q	Carbon dioxide treatment	Patchineelam, 1975
c)	Cation exchanger	Deurer et al. 1978; Forstner and Stoffers, 1981.
ļ		

Table 1.1 Reagents for the extraction of metal phases

Contd....

•
•
•
τ
ŭ.
<b>C</b>
ň.
ŏ.
<u> </u>
<del>, 1</del>
٠
<b></b>
• •
~
Ð
le
ble
able
ſable

	Phase/Reagent	References
с.	Fe/Mn PHASE	
a)	Hydroxylamine- hydrochloride/ acetic acid	Chester and Hughes, 1967; Kumar and Roonawal, 1987; Lee, 1985; Samanidou and Fytianos, 1987; Stenekar et al. 1988; Teraoka and Nakashima, 1990; Tessier et al. 1979; Yoshimura et al. 1988.
( q	Hydroxylamine- hydrochloride/ nitric acid	Bradley and Cox, 1990; Calmano and Forstner, 1983; Chao, 1972; Lum et al. 1982; Forstner and Stoffers, 1981.
( )	Ammonium oxalate- oxalic acid	Calmano and Forstner, 1983; Elliot et al. 1990; Patchineelam and Forstner, 1983; Schwertmann, 1964.
(p	Sodium dithionite- sodium citrate	Aguilera and Jackson, 1953; Engler et al. 1977; Holmgren, 1967.

Contd...

	Dhace/Reagent	References
D.	ORGANIC/SULFIDE PHASE	
a)	Hydrogen peroxide- ammonium acetate	Bradley and Cox, 1990; Calmano and Forstner, 1983; Engler et al. 1977; Hong and Forstner, 1983; Jackson, 1958; Lee, 1985; Lum et al. 1982; Patchi- neelam and Forstner, 1983; Rapin, 1983; Samanidou and Fytianos, 1987; Stenekar et al. 1988; Teraoka and Nakashima, 1990; Tessier et al. 1979; Yoshimura et al. 1988.
( q	Hydrogen peroxide- nitric acid	Gupta and Chen, 1975
(c)	Organic solvents	Cooper and Harris, 1974
(p	Sodium hydroxide/ sulfuric acid	Volkov and Fomina, 1974
е)	Sodium hypochlorite	Gibbs, 1973

Table 1.1 contd....

Contd....

Table 1.1 contd.....

.

.

e/Reagent References	UAL PHASE	line fusion	c/perchloric/ de Groot et al. 1982; ofluoric acids Jackson, 1958; Lee, 1985; Teraoka and Nakashima, 1990.	c/hydrochloric Bradley and Cox, 1990; s Forstner, 1979b; Lum et al. 1982.	c/perchloric Feijtel et al. 1988; Lithnor, 1975; Samanidou and Fytianos, 1987; Tiller et al. 1989.	c acid Calmano and Forstner, 1983; Elsokkary and Muller, 1990; Knudtsen and O'Connor, 1987; Rapin 1983.
Phase/Reagent	RESIDUAL PHAG	Alkaline fusio	Nitric/perchle hydrofluoric	Nitric/hydroc acids	Nitric/perchle acids	Nitric acid
	ш	a)	( q	( )	( p	e)

of sulfide as a result of the disproportionation of dithionite, were observed during the use of dithionite-citrate couple for the determination of moderately reducible phases (Tessier et al. 1979).

Most of the limitations mentioned above could be rectified by careful selection of the extractants and the extraction conditions. However, the non-specificity of the extractants makes the procedure operational in character and the extracted "phases" are often operationally defined. Despite these drawbacks, sequential extraction procedures have yielded valuable insight into the geochemical processes occurring in the sediments (Chester and Hughes, 1969; Heath and Dymond, 1977; Singh and Subramanian, 1984; Kumar and Roonwal, 1987; Chester et al. 1988) as well as into the complexities of geochemical association of trace elements (Tessier et al. 1980; Lion et al. 1982; Allen et al. 1990; Bradley and Cox, 1990). These methods have been widely used in environmental studies to quantify those phases of the sediments that constitute the reservoir for potential release of contaminants into the water column (Lee. Chester et al. 1985; Carruesco and Lapaquellerie, 1985; 1985; Samanidou and Fytianos, 1987; Pardo et al. 1990; Allen et al. 1990; Clevenger, 1990; Elsokkary and Muller, 1990; Zhang et al. Sequential extraction procedures have also proved 1990). to be valuable in the description and prediction of biological trace metal uptake by sediment dwelling organisms and aquatic plants (Luoma and Bryan 1978; Luoma, 1983 & 1989; Langston and Bryan, 1984; Tessier et al. 1984; Gunn et al. 1989)

In spite of the proven advantages in elucidating the environmental effects of trace metals, the sequential extraction techniques have been used, only rarely, in estuarine sediment studies. The information on chemical partitioning of trace metals in sediments is particularly important for estuaries, where because of the rapidly changing hydrochemical

conditions of water column, а steady state vertical distribution of elements cannot be used to understand their reactivities (Hart and Davies, 1981; Amdurer et al. 1983). Further, these waters receive large amounts of natural and anthropogenic inputs of potentially harmful substances (Yeats et al. 1978; Satyanarayana et al. 1985a; Qasim and Sen Gupta, 1988; Grigg, 1990). With their often restricted circulation, it is imperative to know what controls the movement of these chemicals through the system in order to predict whether they will be concentrated locally and in what form or if they will be diluted to the ocean (Duinker, 1986; Chester, 1990).

#### **1.4** ESTUARINE VARIABILITY

The term `estuarine variability' encompasses the spatial diversity and temporal changeability (Wolfe and Kjerfve, 1986) recorded in physical, chemical and biological properties of the estuary in response to exogenous inputs of materials or energy that may themselves vary over wide ranges of magnitude and exhibit different periodicities over a broad range of time scales. Estuaries are mixing zones between fresh river water and saline coastal water. Physical, chemical and biological processes within an estuary vary over a broad spectrum, both in space and time (Morris, 1990; Dyer, 1991). Spatial variations depend on the topography of the estuary, the on gradients in salinities and on the fine sediment concentration (Uncles et al. 1988; Wright and Phillips, 1988; van Leussan and Dronkers, 1990). With respect to temporal variations, several time scales can be distinguished (Bero and Gibbs, 1990): long-term (yearly), seasonal (monthly/fortnightly), intermediate (daily/diurnal) and short-term (hours/minutes). While annual and seasonal variations depend on the climatological factors, fortnightly and diurnal variations are caused by tidal actions (Holmes, 1986; Vale, 1990).

Within an estuary, the amount and direction of the flow and thus the water level and river width vary periodically with tides (Bowden, 1980; river inputs and Dyer, 1988). Sedimentation of suspended particulate matter, resuspension of this material and possible erosion of consolidated sediments change with the flow dynamics and influence the particulate matter concentration in the water column (Michaelis, 1990). These processes, thus leave significant impressions on the trace metal distribution within the estuary, since major part of trace metals in estuaries are bound to particulate matter (Luoma and Phillips, 1988; Vollmer et al. 1990).

It may be remarked that an estuary provides a particularly varied chemical environment for element cycling (Burton and Liss, 1976; Elderfield, 1978; Morris et al. 1982; Martin and Whitfield, 1983). Suspended matter carried by rivers is subjected to gradually altering salinities (and pH) within an estuary. This often causes precipitation of suspended material by flocculation (Sholkovitz, 1976 & 1978; Sholkovitz and Copland, 1981) and chemical exchange between the suspended sediment and estuarine waters that changes the chemistry of both (Van der Weijden et al. 1977). Gradients in the chemical environment, e.g., in the salt or oxygen content, may change the partitioning of trace metals with the possible consequence that for instance, the transport mechanism vary with time and location (Carey, 1990).

The major nutrient elements as well as some of the essential trace metals are much more abundant in river water than in sea water (Martin and Whitfield, 1983). The uptake and release of metals between the organisms and water body regulate the metal fluxes within the estuary (Salomons and Mook, 1980; Windom et al. 1991). In general, estuaries have a high biological productivity but the species composition vary greatly with the changing salinity pattern. While the fresh water plankton die

off with increasing salinity, the marine plankton die with the growing influence of fresh water. The detritus thus formed contribute substantially to the suspended load, especially in the maximum turbidity zone (Michaelis, 1990). The organic rich particulate matter settles to the bottom and leads to the formation of anoxic interstitial waters which strongly influence the partitioning of trace metals between the various components of the system.

Estuaries, adjacent to centres of urbanization and industrialisation, are likely to receive bio- degradable and stable compounds of anthropogenic origin, containing trace metals (Allan, 1990). Early interest in the study of trace metals in estuaries arose mainly from the fact that the estuaries represented an important stage in the supply of trace metals to the oceans (Aston, 1978). The behaviour of trace metals, besides showing distinct seasonal and diurnal variation within an estuary (Bewers and Yeast, 1980; Windom et al. 1991), varies from one estuary to the other (Turner et al. 1990). Seasonal and annual variations in river flow affect the transportation processes that distribute the metals within the system (Nichols et al. 1986; Ackroyd et al. 1987; Kappenberg et al. 1990). The seasonal and annual changes in salinity pattern also influence the adsorption, desorption and chemical partitioning of trace metals. The biological and geochemical activity of a particular trace metal is strongly influenced by its chemical partitioning within the sedimentary phases. Trace metals would not remain permanently burfied in sediments (Yen and Tang, 1977). A matter of particular concern is that of the decaying organic matter which may eventually set free many of the toxic metals that get resuspended by tides, storms and dredging. It is quite possible that the toxic metals released in this way could damage the estuarine and nearshore aquatic life (Dyrssen and Wedborg¢, 1980) and hence intensive studies are needed to determine the release rates of trace metals from estuarine sediments. The

chemical partitioning of sediment bound metals can offer much information in this direction (Hart and Davies, 1981).

The multi-component heterogeneous nature of sediments provides little scope in predicting the partitioning of metals from fundamental theories and hence the quantification has to be made based on empirical observations. Most of the studies reported in literature on chemical partitioning relate to Europe and North America. As far as tropical systems are concerned, very few authentic data are available (Windom et al. 1988). In this respect, the Cochin estuary situated on the west coast of India constitutes an excellent model, under tropical climate for the study of the partitioning of trace metals, especially in view of increasing evidences on the toxic effects of metal contamination in this coastal water body.

### 1.5 COCHIN ESTUARY

The Cochin estuary extending between  $09^{\circ}40' - 10^{\circ} 12'N$ and 76°10′-76°30′E is a typical tropical positive estuary as per Pritchard's classification (Pritchard, 1967; Fairbridge, 1980), as the runoff plus precipitation exceeds evaporation. The climate of the region is tropical maritime with a 1979). characteristic monsoonal season (Ananthakrishnan et al. Two major rivers discharge fresh water into the estuarine system; the Periyar river flows into the northern parts and the Muvattupuzha river into the southern parts. The estuary is connected to the Lakshadweep sea (a part of the Arabian sea) through a permanent opening, the barmouth at Cochin. The Cochin barmouth is 450 m wide and is dredged to a depth of 10-13 m through which tides act within the estuary. Cochin is a major port on the south west coast of India.

The considerable research effort expended in elucidating the physical, biological and chemical characteristics of the Cochin

estuarine system during the last three decades, is indicative of the economic and social importance enjoyed by the Cochin harbour area and its adjoining backwater system. The hydrography of the estuary was investigated by several investigators including, George and Kartha (1963), Cherian (1967); Qasim and Gopinathan (1969), Josanto (1971), Haridas et al. (1973), Shynamma and Balakrishnan (1973), Narayana Pillai et al. (1973), Balakrishnan and Shynamma (1976), Anto et al. (1977), Rama Raju et al. (1979), Udaya Varma et al. (1981 δ. 1987) Lakshmanan et al. (1982), Sankaranarayanan et al. (1986), Anirudhan et al. (1987), Pillai (1989), Sarala Devi et al. (1979), Anirudhan and Nambisan (1990) and Pylee et al. (1990). The information available includes spatial and seasonal distributions of temperature, salinity, dissolved oxygen, pH, alkalinity, and suspended solids of surface and bottom waters of the region as a whole or a part. Nair et al. (1975), Kurian (1975), Gopalakrishna Pillai (1979), al. Anwar et Batcha (1985), Madhusoodana Kurup (1983), Joseph (1989) and several others investigated the biological characteristics of the estuary as the fishery resources from this extensive water body formed the back-bone of the regional economy. The greater awareness on the potential health hazards due to the industrial and domestic effluents added to this estuary prompted several research workers to take up toxicological studies (Lakshmanan, 1982; Sivadasan, 1987; Baby, 1987; Latha Thampuran, 1987; Krishnakumar, 1987; Prabhudeva, 1988) and considerable data is now available on the toxic effects of several of the pollutants, especially of trace metals on several of the native fauna of the Cochin estuary and adjoining marine environments (Lakshmanan and Nambisan, 1977, 1979, 1980, 1983, 1985a, 1985b & 1989; Nambisan et al. 1977; Abraham et al. 1986; Menon, 1986; Baby and Menon, 1986; Sivadasan et al. 1986; Prabhudeva and Menon, 1987; Suresh and Mohandas, 1987; Sathyanathan et al. 1988; Sivadasan and Nambisan, 1988; Krishnakumar et al. 1990).

Like any other estuary located in the vicinity of metropolitan cities and industrial conglomerates, Cochin estuary also is subjected to increasing human interferences (Gopalan et al. 1983; Balchand, 1984; Balchand and Nambisan, 1986; Lakshmanan et al. 1987; Paul and Pillai, 1983a; Joy et al. Nearly 70 percent of the chemical industries of Kerala 1990). State is situated on the banks of river Periyar and the effluents from these industries (fertiliser. pesticide, fungicide, metallurgical, rare earth and catalyst plants) form an important source of pollutants to this estuary. The point source of pollutants identified on the Muvattupuzha river is a newsprint factory, but its effluents seldom reach the estuary proper (Balchand and Nambisan, 1986). In addition to the industrial effluents, the estuary receives organic wastes also, from domestic sewage works, coconut husk retting yards, fish processing units etc. Storm water channels also contribute substantially to the pollutant load of this waterbody. The net effect of these discharges have caused irreparable damage to the flora and fauna of this waterbody (Jayapalan et al. 1976, Unnithan et al. 1977; Vijayan et al. 1976; Sarala Devi et al. 1979; Remani et al. 1980; Venugopal et al. 1980; Remani et al.1981; Gopalan et al. 1983; Remani et al. 1983; Joseph et al. 1988; Sarala Devi and Venugopal, 1989; Ouseph, 1990) and the productivity of this region has been seriously affected.

Control and improvement of water standards in an estuarine system requires an insight into the various physico-chemical parameters that define the water quality. Several workers have recently addressed this problem in Cochin estuary and data is now available on the status of nutrients (Sankaranarayanan and Qasim, 1969; Manikoth and Salih, 1974; Sankaranarayanan and Punampunnayil, 1979; Sankaranarayanan et al. 1984; Lakshmanan et al. 1987; Anirudhan, 1988; Anirudhan and Nambisan, 1990; Sarala Devi et al. 1991) and organic compounds (Nair et al. 1989; Balchand et al. 1990) in waters and sediments of Cochin

estuary. Perhaps the most widely studied toxicants in the Cochin estuary are the trace metals. Earlier reports on the trace metal distribution in water, and biota sediments of Cochin estuary were due to Sankaranarayanan and Rosamma (1978), Sankaranarayanan et al. (1978), Murty and Veeraya (1981)Venugopal et al. (1982), Jayalakshmy and Sankaranarayanan (1983) and Paul and Pillai (1983a; 1983b). Sankaranarayanan and Rosamma (1978) observed high levels of zinc, chromium etc. the particulate matter of estuarine waters. Sankaranarayanan et al. (1978) recorded high concentrations of zinc, copper, iron etc. in oysters of this region. Murty and Veeraya (1981)reported high spatial variability in the distribution of trace metals (iron, manganese, nickel, cobalt, copper and titanium) in the sediments of the southern parts of the estuary and brought out its dependency on the particle size distribution. Venugopal et al. (1982) studied the levels of copper, manganese, cobalt, nickel and zinc in the sediments from the northern part of the estuary and reported noticeable spatial and seasonal variability. The level of trace metals in water, suspended solids and sediments from the industrial zone and backwater zone was studied by Paul and Pillai (1983a). They have also attempted (Paul and Pillai, 1983b) to study the human trace metals through contaminated exposure to food and concluded that the human population in this region was not exposed to high levels of toxic metals in spite of high levels of contaminants in the environmental samples.

Recently Sivadasan (1987) elaborated on the seasonal distribution of toxic metals, copper, zinc and mercury in an estuarine crustacean. Ouseph (1987) measured the levels of zinc, mercury, chromium, lead and cadmium in the sediments of Cochin estuary and concluded that high content of mercury, zinc and cadmium reach the estuary through industrial wastes.

The speciation studies on the trace metals cadmium, copper and zinc in estuarine waters (Shibu et al. 1990) revealed the influence of environmental factors such as, influx of riverine inputs of metals, introduction of industrial effluents and sewage, modification arising out of anthropogenic activities and hydrographic changes related to complexity of water use on metal speciation and fate in the estuary. Levels of iron, manganese, zinc, copper, cadmium, lead, chromium, cobalt and nickel in surficial sediments of the estuary were studied by Nair et al. (1990). These studies revealed that the metal behaviour in the surficial sediments is influenced by natural as well as anthropogenic factors. Most of the metals considered were found to be significantly enhanced in the lower reaches of the northern parts of the estuary. These high values were attributed to industrial effluents from the factories on the banks of the Periyar river. The studies have also indicated the predominant role of seasonal hydrographic changes in regulating metal levels in the sediments.

### 1.6 SCOPE OF THE PRESENT STUDY

Although there are a number of references on the trace metal distributions in the sediments of Cochin estuary, no attempt has so far been made to assess chemical partitioning of trace metals in sediments. Indeed, such information is not available for any estuary on the Indian coast, although a few reports on partitioning of heavy metals in deep sea sediments are available (Murty et al. 1973, 1977 & 1980; Rao et al. 1976 & 1978). Initial studies by the author (Nair et al. 1991) have indicated significant spatial variability in the distribution of the different metal forms in the sediments of this region and therefore, an attempt has been made to improve the data base by following a scheme of intensive sampling and analytical determinations. Since the characteristics of estuarine systems are greatly influenced by the monsoonal cycle, a seasonal study

is expected to throw much light on the behaviour of metals within the estuarine zone. It would also help to establish the background levels of toxic metals in their different chemical forms for assessment of the extent of environmental pollution. Although total elemental concentrations are rarely sufficient to describe the environmental processes adequately, total amounts are useful for the estimation of the quantitative influence of the various fractions determined.

This report characterises the regional distribution and temporal variability of five different forms (`exchangeable', `carbonate bound', `iron-manganese oxide bound', `organic/ sulphide bound' and `residual') of iron, manganese, zinc, copper, cadmium, lead, cobalt, nickel and chromium in surficial sediments within a broad salinity transition zone where the Periyar and Muvattupuzha rivers enter into the Cochin estuary. The aim of this case study is, thus, to demonstrate the usefulness of the sequential chemical extraction technique in assessing the behaviour of trace metal in estuarine sediments. The study successfully tests whether metal contamination and partitioning differ between an urbanized/industrialized estuarine area and an estuarine region subjected mainly to agricultural runoff. The influence of salinity changes on metal concentration and metal partitioning patterns have also been assessed within the estuary. Specific temporal and chronic influences of local metal inputs on monthly/seasonal variations in metal concentrations are delinated.

The data collected provide a measure of the persistence and bioavailability of trace metals in surficial sediments of Cochin estuary. This information in turn gives a general indication of potential for food chain bio-magnification of these contaminants and an early warning of changes in environmental pollution which will ultimately affect organisms at higher trophic levels.

### CHAPTER 2

### MATERIALS AND METHODS

- 2.1 DESCRIPTION OF THE STUDY AREA
- 2.2 SAMPLING PROCEDURE
- 2.3 LABORATORY PROCEDURES
- 2.4 STATISTICAL ANALYSIS

### 2.1 DESCRIPTION OF THE STUDY AREA

The area of the study, the Cochin estuary and the stations sampled are shown in Fig. 2.1. The Cochin estuary is the largest of the estuaries on the Kerala coast. The hydrography of the estuary is controlled mainly by discharges from Periyar and Muvattupuzha rivers and also by tidal action through the Cochin barmouth. The saline water intrusion to the southern parts of the estuary is regulated by the Thanneermukkam bund (marked as BUND in Fig. 2.1), a salt water barrier commissioned in 1975. Four more rivers, namely, Achancoil, Meenachil, Manimala and Pamba discharge their waters at the southern parts



Fig. 2.1 Map of the Cochin estuary showing location of sampling sites (Segment I- Stations 1 to 4; Segment II- Stations 5 to 7 and Segment III- Stations 8 to 10).

## as the burnd'is opened with the 18

onset of monsoon rains. Hydrographical conditions of this estuary have been altered by the inter basin transfer of water from Periyar river to Muvattupuzha river to facilitate a hydroelectric project.

A large number of heavy industrial units are situated on either banks of Periyar river and hence the northern part of the estuary receives large quantities of treated and untreated industrial effluents. This part receives also the effluents from the sewage treatment plant of Cochin metropolitan city. On the other hand, the southern part of the estuary receives water that drains through vast expanses of agricultural land and point sources of trace metal pollution can hardly be detected.

Samples were collected from ten stations(Stations 1 to 10) sited in the Cochin estuary (Fig. 2.1) at monthly intervals between November 1988 and December 1989. The selection of the sites was based on specific geographical features, water flow regimes and anthropogenic activities. The stations may be subgrouped, under three segments for critical appraisal, as :

Segment I - Stations 1 to 4 - Southern arm of the estuary Segment II - Stations 5 to 7 - Lower and middle estuary Segment III- Stations 8 to 10- Northern arm of the estuary

Segment I, the southern arm of the estuary extends up to the southern limit of salt water intrusion and to which point the Muvattupuzha river debounches into the estuary.

Segment II extends from the barmouth up to 10 km on the northern arm of the estuary and contains that part of the estuary that is saline throughout the year.

Segment III extends beyond segment II on the northern arm of the estuary. For most of the time these stations contain only freshwater; saline water intrusion could be recorded only during the periods of very low river discharge. Station 9 located in this segment is the recipient of industrial effluents, and Station 10 located above the industrial zone is free from the influence of industrial pollutants.

### 2.2 SAMPLING PROCEDURE

The surveys were conducted by means of an exclusively fibre glass dinghy "Flying Fish" of the School of Marine Sciences, Cochin University of Science and Technology. A stainless steel, plastic lined van Veen grab was used to collect the sediments. At each location five to six grab-full of sediments were sampled and the top 5 cm layer was carefully skimmed from all the grabs using a polyethylene spoon, homogenized, and stored at  $-5^{\circ}$  C, in polyethylene containers.

### 2.3 LABORATORY PROCEDURES

All glassware and plasticware used in the experiments were previously washed, soaked in nitric acid 20% (v/v) and then rinsed with double distilled deionized water. All reagents used were of analytical grade and were checked and excluded for possible trace metal contamination. Reagent solutions and standard solutions were prepared with deionized double distilled water.

i) Chemical partitioning

Trace metals in the sediments were partitioned according to the scheme depicted in Fig 2.2.

The metal species determined were: Fraction 1. Exchangeable fraction Fraction 2. Carbonate bound Fraction 3. Easily reducible fraction (metals combined with Fe-Mn oxyhydrates)



Fraction 4. Organic fraction including sulfides Fraction 5. Residual fraction

8 to 10 g aliquots of the field samples were used for the sequential chemical extraction. The extractions were done in 250 ml Erlenmeyer flasks with continuous agitation. The phases were separated by centrifugation at 6000 rpm for 10 minutes. The supernatent liquid was removed with a pipette and analysed for trace metals (by AAS) whereas the residue was carefully transferred back to the container with the next extractant to be used. Washings in between the extractions were dispensed with to avoid excessive solubilisation of solid phases. The samples were analysed in duplicate and the analysis was repeated when the values differed by over 10 percent. The mean of the two determination has been reported.

### (ii) Moisture percentage

Concurrent to partitioning studies, aliquots of sediment samples were carefully weighed out for moisture determination also. The samples were dried at 105°C in an electric oven to constant weight (Hakanson and Jansson, 1983). The moisture , was expressed as the percentage weight of water in the total wet weight of the sample. From the percentage moisture values, the sediment metal values were recalculated and the same was expressed in terms of oven dry weight basis.

### (iii) Total metals

The oven dry samples were ground to a fine powder in an agate mortar and 1 g aliqot of these samples were used for the determination of total metals. Each sample was carefully digested with 10 ml of a 5:1 mixture of nitric acid and perchloric acid (Lithnor, 1975). The digest was made up to 50

ml and was centrifuged. Trace metals were determined in the supernatent liquid.

### (iv) Trace metal analysis

Trace metal concentrations were determined by atomic absorption spectrophotometry (Perkin Elmer Model 2380) involving direct aspiration of the aqueous solution into an air-acetylene flame. A standard addition technique was used for the determination of cadmium, copper, chromium, cobalt. nickel, zinc and lead in the first four fractions extracted. For iron and manganese, the extractants were suitably diluted with deionised double distilled water. Blank corrections were applied wherever necessary.

### (v) Precision

In order to determine the precision of the analytical techniques and the reproducibility of data, five sediment samples were selected at random. From each of the five sediment samples, six aliquots each were separately weighed out and subjected to sequential chemical extractions and determination of trace metals. Then the mean, standard deviation and coefficient of variations were worked out. The maximum percentage coefficient variation recorded for the various analytes determined are presented in Table 2.1. The precision was generally low when the concentration approached the detection limits, but improved for higher concentrations. Tt. was observed that the reproducibility of the data was quite good within the operational limits and comparable with values reported in literature for similar studies (Adams et al. 1980; Chester et al. 1988). The precision was much better as the amount of analyte determined increased within limits.

The precision was also checked by summation of the values of all fractions and comparing the same with total metal concentration determined. It was found that the two sets of

	T	determine	fractions	five	notes the	M1 to M5 der
2.2	3.2	5.4	8.3	6.5	7.3	Zinc
3.8	5.1	6.3	9.2	7.3	8.8	Nickel
2.6	3.3	4.7	4.8	5.4	6.2	Manganese
3 <b>.</b> 5	4.7	6.7	8.8	9.7	10.2	Lead
3.0	4.5	6.8	12.1	8.4	9.7	Iron
2.0	2.8	6.4	2.5	7.3	8.5	Copper
3.3	5.5	7.5	13.4	11.8	12.3	Cobalt
3.8	6.0	8.5	12.8	15.0	14.8	Chromium
4.0	7.0	8.5	15.3	12.3	10.5	Cadmium
Total	M5	M4	M3	M2	M1	Metal
recorded	ariation	efficient v ined.	entage coe es determ	perce analyte	Maximum for the a	Table 2.1

values agreed within 10 percent for ninety percent of the samples. In this thesis, the summation values are presented as `total values' because of the convenience in computing the percentage values.

### (vi) Complementary analysis

Concurrent recording of water quality parameters such as salinity, dissolved oxygen, pH and suspended solids was performed by standard procedures as detailed below:

pH : Elico pH meter Salinity : Argentometry (Strickland and Parsons, 1977) Dissolved oxygen : Iodometry (Strickland and Parsons, 1977) Suspended solids : Gravimetry (Butler and McManus, 1979)

The sediments were analysed to determine the organic carbon content by the dichromate method (Walkley and Black, 1934, as modified by El Wakeel and Riley, 1957). The texture analysis was performed as per the technique described by Krumbein and Pettijohn (1938). Exchangeable calcium and magnesium were determined in the ammonium acetate extract by EDTA titration, using Patton and Reeder's indicator for calcium and Solochrome Black T as indicator for calcium plus magnesium (Basset et al. 1978). Magnesium was determined by difference. Total phosphorus was determined in the nitric - perchloric acid extract using phosphomolybdic acid method (Strickland and Parsons, 1977).

### 2.4 STATISTICAL ANALYSIS

The data management was performed using an HCL PC/XT. Initially the average, standard deviation and percentage coefficient variation for all the parameters recorded were computed to get an idea of the spread of the data. Both spatial and monthly variations were evident. The data were further pooled and season-wise as well as station-wise averages were worked out. The data were subjected to analysis of variance to bring out the significant variations in space and time. Not only the variations but also the interactions were found to be significant. As such the average values had to be interpreted cautiously. In this thesis the spatial variations were discussed mainly on the basis of annual mean concentrations recorded at each station. For discussing seasonal variations, four seasons namely, Winter (December, January, February), Pre-monsoon (March, April, May), Monsoon (June, July, August) and Post-monsoon (September, October, November) have been recognised. Each monthly data was considered as a replicate and the seasonal averages were worked The seasonal mean values are presented out. graphically, station-wise, to bring out the seasonal and spatial variation in metal partitioning. The original data has been utilised to work out the correlations between the parameters investigated.
# CHAPTER 3 PHYSICO-CHEMICAL CHARACTERISTICS OF SEDIMENTS

- 3.1 INTRODUCTION
- 3.2 MATERIALS AND METHODS
- 3.3 RESULTS AND DISCUSSION
  - 3.3.1 HYDROGRAPHICAL FEATURES
  - 3.3.2 SEDIMENT CHARACTERISTICS
  - 3.3.3 TRACE METALS
  - 3.3.4 CORRELATION BETWEEN VARIABLES
  - 3.3.5 THE GEOCHEMICAL BACKGROUND

## 3.1 INTRODUCTION.

Monitoring of trace metal concentration in sediments is an integral part of any environmental management programme such as mineral exploration (Hawkes and Webb, 1962), pollution assess-

ment (WHO, 1982; Forstner and Wittmann, 1979; Hakanson and Jansson, 1983; Sinex and Wright, 1988; Hoshika et al. 1991). geochemical flux calculations (Burton, 1978; Martin and Maybeck, 1979; Salomons and Forstner, 1984; Seidemann, 1991) determination of effectiveness of and pollution control programmes (Salomons and Eysink, 1981) . Several investigations have been carried out to assess the trace metal content of Veerayya, sediments of the Cochin estuary (Murty and 1981; Venugopal et al. 1982; Paul and Pillai, 1983a; Ouseph, 1987; Nair et al. 1990). These studies differed in approach and were spatially non-comprehensive. Few studies have taken cognizance of the influence of varying hydrographical and sedimentological parameters. The large variations in metal concentrations which are a consequence of the differences in grain size, organic matter content, water content and sources of anthropogenic inputs, make the interpretation of metal concentration in sediments extremely difficult without adequate prior knowledge on sediment and environmental characteristics. River flow, tidal incursions, geomorphology, bathymetry etc. are important factors in an estuary which interact with each other to produce varying patterns of sediment accumulation (Feijtel et al. defining the hydrographical 1990). Therefore, features and sediment characteristics constitute an essential and useful basis for the study of trace metals. In this chapter, the data recorded during the survey (December 1988 to November 1989) on some important hydrographical parameters (salinity, dissolved oxygen, pН and suspended particulate matter) as well as sediment parameters (water content, grain size distribution, organic carbon content, exchangeable calcium, exchangeable total phosphorus, total and total magnesium, iron, trace metals) are presented and discussed.

3.2 MATERIALS AND METHODS

Details are furnished in Chapter 2

3.3 RESULTS AND DISCUSSION

3.3.1 HYDROGRAPHICAL FEATURES

The annual mean values of salinity, dissolved oxygen, pH. and suspended solids recorded at the surface and bottom are presented in Tables 3.1 and 3.2, along with the summary statistics. The seasonal mean values are depicted in Fig. 3.1. The hydrographical features differed greatly between the stations and between seasons. Stations 5, 6 and 7 of Segment II had the highest salinity throughout the year whereas salinity intrusion seldom reached the Stations 8,9 and 10 of Segment III, except during the winter and the premonsoon months, when the river inflow was minimum. On the other hand, all the Stations in the southern arm of the estuary evidenced salinity intrusion except during the months of heavy rains and a characteristic horizontal salinity gradient was obvious. Both surface and bottom salinities decreased with the onset of the monsoon and a distinct stratification was recorded at the barmouth (Station 5) during the monsoon and the postmonsoon periods. Lakshmanan et al. (1982), Sankaranayanan et al. (1986) and several others have recorded similar observations. Thug. the available data on salinity unambiguously established the influence of the influx of fresh water from rivers and intrusion of sea water into the estuary via barmouth, on salinity distribution. The stratification recorded might play an important role in the settling of detritus and hence, in the metal fluxes.

Dissolved oxygen values of both surface and bottom waters exhibited significant spatial and seasonal variation in

Table 3.	1 Sumn	nary stati	istics on	hydrogra	phical par	cameters (Sali	nity and	dissolved	oxygen)	
Station	Mean	SD	C.V.\$	Min.	Max.	Mean	SD	C.V.\$	Min.	Max.
		                 	Surface	* * * *	 			Bottom		
					Ś	alinity				
1	2.0	1.7	87.8	0.0	5.3	6.6	5.8	87.7	0.0	15.0
2	5.4	4.7	87.8	0.0	12.8	7.6	6.9	90.6	0.0	17.3
3	8.6	7.4	86.3	0.2	18.9	8.8	7.5	85.8	0.2	19.2
4	7.2	6.6	92.6	0.0	17.6	8.3	7.4	88.3	0.0	20.0
5	16.0	10.0	62.7	1.3	31.5	27.6	5.1	18.6	15.1	33.6
6	14.3	9.3	64.6	2.7	29.5	19.7	10.5	53.3	5.5	32.2
7	5.4	4.7	87.2	0.7	16.5	8.1	7.2	88.9	0.6	23.1
8	1.2	1.8	155.9	0.0	6.1	2.1	3.6	175.0	0.0	12.6
6	0.3	1.0	331.7	0.0	3.5	0.7	2.0	268.5	0.0	7.2
10	0.3	1.0	331.7	0.0	3.5	0.4	1.3	331.7	0.0	4.8
				•	Dissolved	oxygen (ml/l	-			
-	5.2	1.2	23.9	4.0	8.4	4.9	1.0	20.6	3.7	7.3
2	5.6	1.3	24.1	3.8	8.5	4.9	1.6	32.3	3.2	8.7
3	3.1	1.0	32.5	1.5	5.3	3.8	1.3	32.8	1.3	6.8
4	3.0	1.3	42.2	1.3	6.6	3.2	0.5	15.6	2.4	4.0
5	6.1	1.5	24.1	4.1	9.1	5.7	2.2	39.0	2.3	8.8
6	7.1	1.7	23.6	3.6	9.1	6.6	1.6	24.6	4.5	9.5
7	7.4	1.8	24.7	4.3	10.2	7.1	1.6	23.1	4.6	9.8
8	7.4	1.7	23.0	4.3	9.5	7.9	1.8	23.0	5.0	11.1
6	7.8	2.2	28.0	5.2	11.4	7.2	2.5	34.9	2.5	11.4
10	7.4	2.2	30.0	4.1	11.4	7.5	2.5	33.1	3.0	11.4

Table 3	1.2 Sı	ımmary	statis	tics on h	ı yd rograpi	nical param	əters (pH a	and suspe	nded solic	1s)	
Station	Mea	, , , , , , , , , , , , , , , , , , ,	SD	C.V.\$	Min.	Max.	Mean	SD	с.v. <del>%</del>	Min.	Max.
		           		Surface	 				Bottom		
						lq	H				
1	7.(	0 0	.2	2.9	6.6	7.3	7.0	0.2	2.3	6.7	7.3
2	7.	0 0	.2	2.9	6.7	7.5	7.1	0.2	2.5	6.8	7.4
Э	7.	0	1.2	2.6	6.7	7.3	6.9	0.3	4.5	6.0	7.2
4	.7	0 0	1.2	3.4	6.7	7.5	7.0	0.1	2.0	6.8	7.3
5	1.	4 0	.3	3.7	7.0	8.0	7.4	0.2	3.3	7.2	8.0
6	7.	3 0	.3	4.0	6.9	7.9	7.3	0.3	4.1	6.9	8.0
7	7	2 0	.2	3.0	6.9	7.7	6.6	1.9	28.9	4.3	7.8
8	7.	0 0	.2	2.5	6.6	7.2	7.1	0.2	2.9	6.7	7.5
6	.0	8 0	7.7	11.0	4.5	7.7	6.8	0.8	11.3	4.4	7.5
10	.0	9 6	.8	11.5	4.3	7.3	6.8	0.8	11.6	4.3	7.4
					SI	ispended so	lids (mg/l)				
						-	, ,				
1	ں م	9	.6	17.7	1.4	15.2	14.5	8.9	61.6	5.0	36.4
2	0	2 4	6.1	53.2	4.4	20.3	16.1	7.0	43.8	4.2	27.4
3	21.	8 13	1.4	61.5	1.2	54.4	24.0	14.2	59.1	5.2	56.4
4	14.	2 7	.4	51.8	4.7	26.1	23.7	0.0	37.8	10.4	36.1
5	20.	6 23	1.4	113.8	7.7	97.2	68.8	49.7	72.2	9.3	201.1
6	17.	3 12	8.	74.0	3.5	48.3	72.9	75.7	103.9	14.3	289.1
7	11.	4 12	7	111.1	1.5	41.0	11.2	5.5	49.0	2.8	21.3
8	6.	7 6	.5	97.1	1.5	25.4	7.7	3.7	47.7	2.2	15.2
6	8.	0 8	6.1	110.2	1.3	32.9	8.9	10.4	116.5	2.2	41.8
10	7.	3 8	1.6	117.7	1.3	29.4	6.9	8.7	88.6	2.1	32.1



Fig. 3.1 Seasonal mean values of hydrographical parameters (SS = suspended solids (mg/l); DO = dissolved oxygen (ml/l))

conformity with observations recorded by earlier workers (Vijayan et al. 1976; Sarala Devi et al. 1979; Anirudhan, 1988). Except for the Stations 5 and 6 in the lower reaches of the northern arm of the estuary, the surface waters recorded the lowest values of dissolved oxygen during the premonsoon months. This may be due to the high temperature prevailing during this period. Dissolved oxygen values were relatively lower in the southern part of the estuary, especially during the postmonsoon and the winter periods. Nair et al. (1990)observed comparatively high content of dissolved organic matter in this part of the estuary during postmonsoon periods. Gopalan et al. (1987) reported rapid decaying and deposition of salvinia (a freshwater macrophyte) under the influence of intruding saline waters. Put together, these observations lead to the conclusion that the depletion in dissolved oxygen was due to the rapid decomposition of organic matter. The lowest value for dissolved oxygen was recorded at Stations 3 and 4 on the southern arm, during the winter season. The variations between the surface and bottom waters were insignificant. In the riverine reaches of the estuary the dissolved oxygen content was comparatively higher during the monsoon season and it could be attributed to the high turbulence in river flow during this rainy period.

With regard to the pH of surface and bottom waters, significant seasonal or spatial variations were not recorded except for the high acidic waters observed at Stations 9 and 10 during December 1988; this was presumably due to some surreptitious discharge of acid containing industrial effluents. There was a steady increase of pH from 6.8 to 7.4 as the river water mixed with saline waters in the northern arm, but in the southern arm such a variation was not noted.

The highest concentration of suspended matter in the water column corresponded with periods of maximum river discharge in

the upstream stations (Fig. 3.1) and with the period of maximum primary productivity in the lower reaches of the estuary. Suspended matter concentrations in the surface waters of the estuary were, thus, controlled by seasonal fluctuations of river runoff and biological productivity, with river runoff being predominant during the monsoon months and biological productivity being predominant during the premonsoon months. In near bottom waters, suspended matter concentrations were 3-4 times higher than surface water values and showed no obvious seasonal variability. The near bottom concentration increases were related to local resuspension events which were effected by tides and the input of fresh water from rivers.

### 3.3.2 SEDIMENT CHARACTERISTICS

The sediment-associated parameters considered in the present study were moisture content, grain size distribution, organic carbon content, exchangeable calcium, exchangeable mangnesium, total phosphorus and total iron. The twelve monthly mean and standard deviation determined for these parameters at the ten stations surveyed are presented in Table 3.3 to 3.6 . Fig. 3.2 depicts the seasonal mean values. Lower values were observed for the sediments of Segment III, whereas higher values were recorded for the sediments of Segment II. Seasonally (Fig. 3.2), the monsoonal values were lower except for Stations 1 and 2 in Segment I. The premonsoonal values were generally higher than those of other seasons. The variation in water content could be attributed to the variations in organic matter and clay content.

Natural sediments consist of particles of varying sizes and for deciding the size groups, several class intervals based on average diameter have been suggested. One such system includes the grading of particles into sand (>63 $\mu$ m) silt (4-63 $\mu$ m) and clay (<4 $\mu$ m) sizes (Krumbein and Pettijohn, 1938).

Table	з• з	Stationwise (moisture,	summary sand and	statistics silt)	on sediment	parameters
Station		Mean	SD	C.V.\$	Min.	Max.
				Moisture f		
1		26.76	10.21	38.17	18.50	49.30
2		25.16	2.58	10.25	20.60	31.20
Э		31.70	10.31	32.51	23.20	58.90
4		31.07	10.14	32.65	20.20	55.50
5		44.50	8.31	18.68	28.60	56.40
6		51.15	3.17	6.20	47.00	57.80
7		37.80	8.15	21.57	27.80	52.30
8		20.31	2.27	11.17	16.80	24.30
6		21.13	3.75	17.73	16.80	32.60
10		19.56	2.68	13.71	14.30	25.00
				Sand §		
		82.37	12.08	14.66	62.31	97.12
2		83.88	7.37	8.78	67.26	90.37
e		75.93	20.36	26.81	25.41	91.34
4		78.05	18.04	23.11	33.88	91.69
5		39.97	23.72	59.34	10.79	83.99
6		15.65	10.07	64.34	4.00	35.88
7		63.53	21.72	34.19	26.83	89.39
8		97.04	1.93	1.99	93.03	99.26
6		96.14	2.80	2.91	90.07	99.50
10		93.76	4.52	4.82	82.46	98.17
				Silt 将		
<del>, - 4</del>		7.94	5.90	74.21	1.05	18.37
2		9.20	5.67	61.70	2.88	20.64
e		11.81	10.18	86.26	3.31	36.26
4		13.67	13.04	95.35	2.58	44.93
5		36.25	16.74	46.18	5.89	52.90
6		46.69	8.77	18.78	32.55	57.45
7		17.82	10.64	59.73	4.13	39.62
8		1.01	0.79	78.18	0.25	2.72
6		1.99	1.31	66.07	0.12	4.25
10		3.10	2.84	91.64	0.49	9.24

fable 3.4	Stationwise (clay, <16	summary pm fracti	statistics or on and orgar	n sediment nic carbon)	parameters
ation	Mean	SD	C.V.\$	Min.	Max.
			Clay %		
	9.68	6.26	64.64	1.83	19.32
	6.92	2.22	32.01	4.10	12.10
	12.26	10.35	84.37	3.85	38.33
	11.60	9.01	77.65	4.78	33.90
	22.13	10.00	45.17	10.12	46.20
	37.66	3.18	8.44	31.55	43.14
	18.65	11.37	60.96	6.48	36.80
	1.95	1.31	67.29	0.23	4.25
	1.88	1.58	84.35	0.25	5.68
0	3.14	1.84	58.62	0.75	8.30
		<16 μm	size fracti	su %	
	12.51	7.98	63.78	2.15	27.34
	9.79	3.09	31.59	6.00	16.35
	18.24	17.98	98.57	5.79	61.95
	16.01	12.89	80.52	6.25	48.50
	35.41	18.10	51.13	13.12	67.02
	62.21	7.65	12.30	49.85	70.90
	28.48	16.95	59.49	9.75	54.55
	2.55	1.65	64.72	0.38	5.20
	2.88	2.04	70.66	0.44	7.35
0	4.17	2.69	64.53	0.98	11.33
		Organ	nic carbon (1	ng/g)	
	7.72	5.58	72.24	1.70	19.35
	5.12	2.20	43.05	3.00	10.35
	9.04	6.31	69.87	3.72	23.32
	8.68	6.40	73.70	1.62	26.46
	21.42	7.16	33.43	6.90	28.74
	29.76	3.54	11.89	22.32	33.54
	21.89	9.26	42.32	7.77	34.98
	1.59	1.20	75.86	0.36	4.68
	2.60	3.88	149.22	0.18	14.08
0	1.66	1.28	76.73	0.30	4.53

Table	3 <b>.</b> 5	Stationwise (exchangeab	summary le Ca, Mg	statistics on and Ca+Mg).	sediment	parameters
Station		Mean	SD	C.V.\$	Min.	Max.
         	     		Exchang	eable Ca (me	q/100g)	
1		1.88	0.75	39.76	0.96	3.64
7		2.33	1.16	49.88	1.37	5.41
e		5.60	2.61	46.58	0.41	11.19
4		4.99	1.69	33.87	2.53	8.34
5		16.52	3.17	19.22	11.90	22.36
6		14.73	3.71	25.17	7.79	21.41
7		3.53	2.12	59.90	0.90	8.28
8		0.74	0.83	113.02	0.06	3.20
6		0.43	0.50	115.12	0.03	1.56
10		0.37	0.30	80.53	0.04	0.90
			Exchang	eable Mg (me	q/100g)	
		2.85	1.10	38.67	1.72	5.46
2		2.50	1.11	44.54	0.77	4.95
3		4.45	3.82	85.80	0.49	15.36
4		3.10	2.61	84.24	0.45	7.50
5		11.33	2.85	25.13	5.40	14.42
9		16.59	5.36	32.31	7.63	30.92
7		4.22	4.00	94.87	0.92	14.40
8		0.87	1.29	148.77	0.06	4.60
6		1.76	4.20	238.52	0.03	15.23
10		0.56	0.79	141.24	0.03	2.80
		E	Exchangeab	ole Ca + Mg (	meq/100g)	
-		4.73	1.34	28.40	3.02	7.34
2		4.84	1.33	27.47	2.51	6.73
e		10.05	5.92	58.96	3.21	26.55
4		8.09	3.02	37.28	3.83	12.60
5		27.85	4.70	16.86	17.62	35.33
6		31.32	5.72	18.25	21.19	43.03
7		7.75	4.69	60.49	2.00	17.05
8		1.60	2.09	130.72	0.12	7.80
6 6		2.19	4.62	210.50 106 07	0.06	16.79
Π		0.43	0°.40	10.0UL	0.01	0

Table 3.	6 Stationwise and iron.	summary stati	istics on total	phosphorus
Station	Mean (mg/kg)	SD	Min. (mg/kg)	Max. (mg/kg)
		Phospho	rus	
1	472.2	280.7	218.9	1336.7
2	431.0	101.5	313.7	651.0
Э	1100.0	993.6	456.8	3526.2
4	750.8	125.3	549.5	940.2
5	1620.4	458.1	1067.2	2597.5
6	2185.0	340.8	1675.6	3003.5
7	1995.5	660.5	1255.9	3241.2
8	406.0	180.9	144.1	677.4
6	325.5	157.8	96.2	606.6
10	318.4	165.4	122.8	669.6
		Iron		
1	7898.6	1665.8	5515.7	10636.8
2	8302.1	851.1	7126.6	9972.2
Э	6585.2	931.1	4998.2	8500.6
4	6928.6	953.8	5551.8	8780.4
5	14543.1	2173.6	11655.1	18886.2
6	15982.3	2437.6	12554.4	20182.8
7	7700.7	1083.7	5593.3	9502.3
8	5007.0	817.4	4021.0	6941.7
6	4655.8	1460.8	2810.9	7918.0
10	5642.2	1164.3	3863.3	7326.9



Fig. 3.2 Seasonal mean values of sediment characteristics

The Shepard (1954) diagram is utilised to describe the composition of grain sizes. The relative abundance of the various grain-size groups in the surficial sediments are presented in Tables 3.3 and 3.4. The seasonal variations are presented in Fig. 3.2. The sediments of Segment III were mainly composed of sand particles. Cobbles and gravel also constituted a major fraction of the sediments, especially in the samples collected during the monsoon period. In the sediments of Segment I also (Stations 1 to 4) sand size particles constituted the bulk of the sediment, but the size of the particles varied from fine sand to medium sized (63 to 1000 um) grains. The sediments of Segment II, however, had the lowest contribution from this size class. A slightly higher sand percentage, neverthless, was recorded during the monsoon. Recently, Sundaresan (1991) also reported similar features in the seasonal distribution of sediment particles in the Cochin harbour area. Sediments of Station 7 in Segment II showed a clear cut seasonal variation in sand content, indicating significant bedload movements during the monsoon and the postmonsoon periods. The distribution of clay-size particles showed the opposite trend. An exceptional feature noted in the grain size distribution of sediment particles at Station 1 was the presence of a higher percentage of clay content during the monsoon and the postmonsoon seasons at Station 1, whereas the general trend was a depletion in clay content for the above periods. In addition to the above three size classes, the size group consisting of <16 µm fractions, was determined, since this fraction was considered to be of importance in metal transport in estuaries. Distribution pattern of <16 µm size fraction followed the same trends as shown by clay size particles.

The sediment distribution pattern in an estuary depends on several factors such as sediment sources, the texture of the sedimentary material supplied, the bottom topography of the

basin and general hydrographic features (Seralathan, 1986). Tn the Cochin estuary, the bulk of the sedimentary material is being supplied by the two rivers, the Periyar on the north and the Muvattupuzha on the south, mainly during the rainy months (i.e. from June to September). The sediments are transported as bedload as well as suspended load (Veerayya and Murty, 1974). The percentage distributions of different size fractions of sediments broadly indicated the existence of a partition finer components of the sediments. between the coarser and Coarse sand dominated in the upper reaches (Segment III) of the northern arm, whereas medium to fine sand and silt dominated the southern parts (Segment I). Major part of Segment II was underlain with sediments of fine size. Further, it was indicated that in the estuarine region, the settling of finer fractions of the sediments took place preferentially during the winter and pre-monsoon season. The decreased percentage of the silt and other finer particles during the monsoon periods could be attributed to flushing of finer materials from the estuarine region to the sea. Likewise, the increase noted in the case of finer fractions during the winter and premonsoonal period could be attributed to the low fluvial discharge and a better mixing of saline and freshwater that facilitated flocculation and settling of suspended particles.

The distribution pattern of organic carbon content also exhibited distinct spatial and seasonal trends (Fig. 3.2). Relatively higher values were recorded for the sediments of Segment II. The coarse sandy sediments of Segment III were associated with the minimum content of organic carbon. Seasonal maximum was recorded during the pre-monsoon period and minimum during the monsoon period. A comparison of data on grain size distribution and organic carbon distribution revealed that the organic carbon distribution in estuarine sediments followed broadly the sediment distribution pattern, in that, finer the sediment, higher was the organic carbon

content. Both were high in sediments of mid-estuarine reaches in the north. From this relation it could be inferred that the adsorption of organic compounds onto clay minerals was promoted under saline conditions; organic coating on clay minerals affected the size distribution and settling rates of inorganic particles and thus contributed to the increase in both claysize particles and organic matter.

The present study was confined to the 0-5cm layer of surface sediments, which was usually influenced by the varying inputs into the estuary. Qasim and Sankaranarayanan (1972) studied the seasonal variations in the quantity and composition of detritus which settled to the bottom and reported that the bottom sediment was chemically similar to the detritus. These authors had also noted that the quantity of settled detritus was minimum during the monsoon and postmonsoon periods when the turbidity in the estuary was maximum; they attributed it to the stratification which developed in the estuary from June to October as a result of discontinuity in the vertical profiles of temperature and salinity, which probably reduced substantially the settling rates of detritus. The rate of detritus settlement showed a sharp increase as the estuary returned to a well-mixed condition. The seasonal variations observed in fine particle composition and organic carbon content, during the present study, also indicated that surficial sediments were quite sensitive to the periodic addition of detritus, mainly composed of inorganic matter (fine silt and sand) to which organic matter adhered and formed aggregates.

Calcium and magnesium ions determined in neutral, normal `exchangeable' ammonium acetate was designated as and expressed as milliequivalents per 100g of sediment. The content of these in the surficial sediments seemed to be regulated bv both salinity of ambient water and particle size distribution in sediments. It is interesting to note that magnesium and calcium shared the available sorption sites almost equally in both riverine and estuarine sediments.

Total phosphorus determined in the sediments after digestion with nitric-perchloric acid mixture exhibited significant seasonal and spatial variations. The highest value was recorded in the sediments of Segment II and the spatial distribution pattern followed the texture in the sense that higher the percentage of finer particles, higher was the phosphorus content. This dependency of phosphorus content on particle size is in agreement with an earlier report by Murty and Veerayya (1972) from the southern parts of the estuary. The post-monsoonal values were high except at Stations 1, 2, 6 and 7 where the maximum was recorded during the monsoon. Seasonal fluctuations and range of values recorded during the current investigations were in unison with the observations of Sankaranarayanan and Punampunnayil (1979).

Total iron values in the estuarine sediments varied from 2811 to 20183 mg/kg (Table 3.6). The highest value was observed at Station 6 (Segment II) during the postmonsoon season (Fig. 3.2). Stations 2, 3 and 4 of Segment I, 5 of Segment II and 8, 9 and 10 of Segment III, recorded the minimum iron concentrations during the monsoon season. The winteral/premonsoonal values were generally higher at these Stations. For the remaining Stations (Stations 1 of Segment I, 6 and 7 of Segment II) the minimum value for total iron was recorded during winter with values increasing slightly through monsoon and post monsoon. The variation in seasonal distribution followed the trends observed in the case of total phosphorus and a close association of these two elements was indicated. It was concluded from the above relation that the iron supply to the estuary mainly depended on land drainage and erosion of agricultural soils of this tropical region, where iron

phosphate formation was the rule rather than a mere probability as judged from chemistry of tropical soils (Juo, 1981).

#### 3.3.3 TRACE METALS

The summary statistics giving the minimum and maximum values, mean, standard deviation and percentage coefficient variation of the total content of iron, manganese, zinc, copper, cadmium, lead, cobalt, nickel and chromium in sediments of the Cochin estuary are furnished in Table 3.7. The concentration of these metals varied over a wide interval: the minimum variation (%CV= 47.41) was recorded in the case of iron and maximum (%CV = 113.32) for cadmium. The percentage coefficient variation had a bearing on the abundance of metals: iron was the most abundant (mean = 8325 mg/kg) and cadmium, the least abundant (mean = 1.66 mg/kg) metal in this system. The observed order of abundance of other metals zinc> was, manganese > chromium > nickel > lead > copper > cobalt.

The ranges, mean and the 95% confidence intervals for the mean of the total concentrations of the nine metals studied are presented stationwise in Fig. 3.3. The data revealed the following salient features with regard to the spatial distribution of trace metals in the sediments of the Cochin estuary.

(i) The highest values, for all the nine metals studied were recorded in the sediments of Segment II (Stations 5 to 7)

(ii) The contents of iron, manganese, cobalt, nickel, chromium and copper were higher in the sediments of Segment I (Stations 1 to 4) in comparison to that of Segment III (Stations 8 to 10)

(iii) The content of zinc was slightly higher in sediments of Segment III than that of Segment I.

Table 3.7	Summary statis	tics on total m	etal distribut	ion (n = 120)	
Metal	Mean (mg/kg)	SD	СV%	Min. (mg/kg)	Max. (mg/kg)
Iron	8325	3947	47	2811	20183
Manganese	66.03	48.32	73.18	7.00	169.60
Zinc	70.40	67.22	95.48	13.53	258.89
Copper	12.20	11.43	93.70	1.33	50.19
Cadmium	1.66	1.88	113.22	0.34	8.44
Lead	14.86	10.99	73.95	3.49	49.85
Cobalt	9.23	6.89	74.57	1.29	24.92
Nickel	21.15	17.74	83.88	1.86	70.63
Chromium	29.99	24.80	82.70	4.13	99.73





(iv) Total lead and cadmium were distributed more or less in equal levels in Segments I and III.

(v) Within Segment I, the concentration of manganese, iron, chromium, nickel and cobalt exibited a slight decreasing trend towards the marine side.

The above observations demonstrated that the total concentration of metals in sediments was biased towards particle size distribution in sediments which in turn was controlled by the fluvial processes that operated at each Station. Thus in Segment II, the zone of accumulation where the fine suspended particles got deposited at a faster rate, the total metal concentrations were relatively high. The stations of Segment III located on the northern arm of the estuary, which was erosional in character, as revealed by the coarse texture of the sediments, contained low levels of metals. Sediments of Segment I, in the southern arm of the estuary which was transportational in character, had intermediate metal values. However, an exemption to the above generalisation was noted in the case of zinc as the values were higher in Segment III than in Segment I, and this exceptional behaviour could be attributed to anthropogenic inputs from the chemical industries operating on the banks of Periyar river as have been well documented (Venugopal et al. 1982; Paul and Pillai, 1983; Nair et al. 1990)

In order to reduce the bias due to the textural class differences, metal data were normalised to the iron values and the seasonal mean values were computed. Iron was selected primarily because of its abundance and also because of its well established estuarine behaviour. The data presented in Fig. 3.4 brought out some additional information on the distribution pattern of trace metals. One striking observation was that the spatial distribution pattern of the trace metals cadmium,





copper, lead and zinc were quite similar. In the northern arm of the estuary these metals exhibited a gradual increase from upstream upto the middle estuary (Station 7) and then a sharp decrease towards the marine regions. In the southern arm of the estuary also these metals exhibited a slight increasing trend towards the marine side. The uniform distribution pattern suggested similar transportation and mixing behaviour for these metals in this estuary. A decrease in the content of copper, lead and cadmium in the estuarine sediments under the marine influence was recently reported by Forstner et al. (1990) for the Elbe estuary and they observed that the same corresponded with an increase in marine borne sediments which diluted the contaminated sediments. Similar conclusions were made bv Nolting et al. (1990) from studies in the Rhine and Meuse estuaries. Several other workers attributed this reduction in trace metal content of sediments to desorption reactions as the particulates interacted with saline waters of high ionic strengths (Aston, 1978; Keeny-Kennicutt and Presley, 1986). For the Cochin estuary also, it was reported that the sediments of the harbour area was comprised of both riverine and marine inputs (Gopinathan and Qasim, 1971; Pillai, 1989) and that the sediments from inshore region were transported into the estuary (Gole and Tarapore, 1966) under the influence of tidal action. Hence the reduction in metal values could be ascribed to both the events: dilution by marine-borne particles and desorption under the influence of saline waters.

The metals nickel, chromium and cobalt also exhibited a decreasing trend towards the marine side in Segment II, especially during the periods of increased salinities, but the decrease was not as sharp as was noted for zinc, lead, copper and cadmium. However, the behaviour of manganese was exceptional and the values did not exhibit any variation between the Stations in Segment II (Fig. 3.4). This exceptional behaviour of manganese was reported from other estuaries as well (Wollast et al. 1979; Trefry and Presley, 1982; Wilke and Dayal, 1982). This was attributed to the fact that the manganese was not readily susceptible to impoverishment in sediments. nearshore marine However within Segment I, manganese values decreased sharply from riverside to marine side. Higher manganese content in low-saline and freshwater sediments was reported also from the Hudson estuary (Williams et al. 1978). A decreasing trend from head to mouth was observed in the case of cobalt and nickel also, in Segment I.

The increasing trend in the total content of cadmium, copper, lead and zinc within Segment I and III towards the marine side was attributed to the increasing influence of salinity during flocculation, precipitation and settling of fine colloidal and suspended particles. Manganese requires a higher salinity (Fukai and Huynh-Ngoc, 1968) for precipitation and therefore may move unchanged within this estuary. However, the sharp downstream decrease noted in the case of manganese, cobalt and nickel in Segment I could not be explained on the basis of these chemical factors, and this aspect would be considered in greater detail while discussing seasonal variations.

The seasonal variations in total trace metal content are also depicted in Fig. 3.4. The salient results that could be summarised from the data are as follows.

i) All metals exhibited significant seasonal variations with lower values during the monsoon and postmonsoon periods (corresponding to high discharges through the river), and higher values during the winter and premonsoon. For all the metals, the highest seasonal value was recorded during the premonsoon period almost at all stations.

ii) Seasonal variations were not very conspicuous for Stations located in the southern part of the estuary (Segment I), as it was for Stations on the northern arm of the estuary (Segment II and III)

iii) Sediments from Stations 1 and 2 located on the southern most part in Segment I, exhibited an exceptional behaviour in the seasonal distribution of trace metals. A well defined seasonal pattern, as observed at the other Stations was absent here, and most of the metal values were slightly higher during the monsoon in comparison to the premonsoon values.

When the data on seasonal distribution of total metals were further examined together with the data recorded by earlier workers (Nair et al. 1990) a sinusoidal type of variation in metal levels was indicated . This cyclical seasonal pattern with a minimum during monsoon and maximum during premonsoon was most pronounced for the stations located on the northern arm of the estuary. The premonsoonal maxima in trace metal content could be related to the biogeochemical changes occurring in the river itself and also to the hydrological and sedimentological character which vary along with reduced discharge rate conditions during these months. The premonsoon maxima coincided with a salinity maximum of the ambient water and also with an increase in sediment organic carbon. It is postulated that, during periods of low riverine flow, the longer residence times of water and suspended matter would result in an enhancement of metal scavenging, coagulation and settling, resulting in higher content of metals in sediments.

The seasonal enrichment factors on total trace metals, organic carbon and <16  $\mu$ m size fractions are presented in Table 3.8. The values were obtained by dividing the mean seasonal value with the mean monsoonal value at each station. At a given Station, the seasonal enrichment factors for the different

Table	3.8	Seasona	l enrich	nment of	trace	netals, org	ganic cart	on (0.C)	and <16	µm size	fraction	
Station		fe	ПМ	5	Си	Cd	Ъb	Co	Nİ	Сr	•°.0	<16 μm
	a)	Winter/	Monsoon				8 1 1 1 1 1 1	5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1				
Ļ		0.9	0.7	1.4	1.5	1.3	1.1	0.8	1.0	1.3	0.8	0.7
2		1.0	0.6	1.1	1.0	1.0	1.2	0.8	1.2	1.0	1.0	0.6
с С		1.2	1.2	1.0	1.1	1.3	1.6	1.2	1.1	1.1	1.4	1.2
4		1.3	2.1	1.3	1.3	1.3	1.4	1.0	0.9	1,3	1.3	1.3
5		1.2	1.5	1.0	1.9	1.9	1.7	1.9	2.2	1.8	1.6	1.9
6		0.8	1.3	1.1	1.1	1.6	1.0	1.3	1.2	1.6	1.2	1.2
7		0.8	1.4	1.3	2.5	1.0	1.4	1.5	1.6	1.1	0.7	2.6
8		1.4	1.1	1.9	1.9	1.4	1.9	1.6	1.4	1.6	7.0	4.5
6		1.3	1.9	3.7	3.9	2.9	1.4	2.6	3.0	2.2	20.3	1.4
10		1.5	1.2	0.8	3.0	1.6	1.8	1.5	0.8	1.4	5.3	2.6
	( q	Premon	soon/Mo	usoon								
1		0.8	0.4	1.4	1.1	1.1	0.9	0.7	0.8	1.2	0.4	0.2
2		1.2	0.7	1.2	1.2	1.7	1.5	1.2	1.1	1.1	0.5	0.5
e		1.3	3.4	2.1	3.8	2.7	2.9	3 <b>.</b> 3	2.2	1.8	3.2	4.5
4		1.2	3.0	2.1	2.2	1.4	2.1	1.3	1.2	1.8	2.6	2.7
5		1.3	1.8	1.7	2.3	5.3	2.1	2.2	2.3	2.1	1.5	1.7
9		1.0	1.3	1.2	1.2	1.9	1.1	1.2	1.3	<b>1.</b> 9		1.2
7		0.9	2.0	1.6	3.4	2.1	2.4	2.0	2.5	1.8	1°3	<b>4.1</b>
8		1.1	1.9	2.1	2.1	1.7	2.8	1.6	1.3	2.4	3.2	<b>4</b> .9
6		2.2	3.5	5.1	10.9	2.1	3.6	3.1	4.7	1.6	3.4	2.9
10		1.4	1.7	1.0	2.9	0.9	1.8	1.6	1.0	2.0	1.2	1.2
	с)	Postmon	1soon/Mc	noosn								
1		1.2	1.0	1.2	1.2	1.1	1.0	1.4	1.7	1.3	2.3	1.6
2		1.1	0.8	1.0	1.2	1.3	1.1	1.0	1.0	1.0	0.8	0.6
с С		1.1	1.0	1.1	0.8	1.1	1.0	0.9	0.6	0.7	1.4	1.3
4		1.3	2.1	2.1	1.9	1.9	1.1	1.3	1.1	1.3	2.5	2.9
5		1.1	1.1	1.2	1.1	1.0	1.1	1.1	1.1	1.1	1.2	1.0 ,
6		1.2	0.7	1.2	1.1	1.3	1.2	<b>1.</b> 0 î.0	0 <b>.</b> 9	1.4	7. C	<b>1.</b> 1
L		1.1	1.3	1.2	1.2	1.0	0.9 0	6.0	1.4	ۍ د ۲	0.0 1	D.1
8		1.0	0.9	0.8			0.8	1.1	1.0	<b>-</b>	3.L	н Ч
5 5		1.5	1.0	6.0	1.0	0.8	1.2	1.5 A f	۰ ۲۰	 	0 u 7 n	0.0
TO		1.3	1.0	L.J	1°0	<b>T • T</b>	0°T	۲.0	<b>T</b> •U	<b>+ • +</b>	) • )	

metals and other parameters varied within very narrow ranges and this suggested that the factors and processes responsible for the seasonal variation were more or less similar for all the metals. In other words, it is the physical and biological processes rather than the chemistry of individual metals which was more important in controlling the seasonal enrichment of total content of trace metals. However, the seasonal enrichment factors varied greatly between the stations, indicating that the physical processes acted with different intensities at different stations. Seasonal enrichment was much higher in the sediments sampled from the northern arm of the estuary than in those from the southern part. Sediments of Station 9, the industrial effluent receiving site, recorded the highest seasonal enrichment factor, for all the metals during the premonsoon period.

An exceptional feature noted in the seasonal enrichment pattern was the depletion of manganese and cobalt during winter and premonsoon periods at Station 1 and 2. Iron and nickel also recorded slight depletion at Station 1 during winter and premonsoon periods. With respect to the spatial distribution it was noted that these metals (iron, manganese, cobalt and nickel) exhibited a decreasing trend downstream the in southern part (Segment I). This indicated additional sources of metals at these two Stations on the southern end of the study area, during the monsoon and postmonsoon periods. This anomalous seasonal behaviour was noted also in the sediment parameters like water content, organic carbon content, and particle size distribution.

A possible reason for this unique behaviour of sediment based metal content at Station 1 and 2 could be the impounding of water beyond the Thanneermukkam bund for the winter and premonsoon periods (December- May) and the releasing of the same after the onset of monsoon (May- June). During the periods

of impoundment the suspended matter with high content of organic matter and trace metals (Mallik and Suchindan, 1984) settled just inwards of the bund as unconsolidated mud. On opening the bund just after the onset of monsoon rains, the loose mud got eroded fast alongwith considerable amount of muddy turbid water discharging into the southern Cochin estuary; the southern end of the study area (Fig. 2.1) received these deposits in the shallower but broad estuarine regions around Station 1 and 2. Only after prolonged periods of high discharge, later during the monsoon months when all the fine sediment was transported, the system returned to normalcy.

### 3.3.4 CORRELATION BETWEEN VARIABLES

The relationship between the distribution pattern of various metals studied has been determined by finding out the Pearson coefficient `r'. The values for the various metal pairs are presented in Table 3.9. The table also provides the correlation coefficient of trace metals with some of the sediment and hydrographic parameters studied. Significant and positive correlation between sediment moisture content and metal contents indicated that both were closely related to the fine particle and organic matter content in sediments. The relation between salinity and metal content was also found to be positive and significant; the relationship observed for iron, chromium, manganese, nickel and cobalt was more strong than that observed for zinc, copper, cadmium and lead. These again could not be attributed features to anv direct relationship between metals and salinity, but had to be traced to the hydrological conditions that controlled the transport of silt and suspended matter. The erosional and transportational zones in the upper estuary were conducive neither for fine particle settling nor for salt water intrusion; so also the high discharge rates during the monsoon periods encouraged transportation by resuspension as well as bedload

	positive).								
Parameter	Fе	Сr	uM	Ni	Zn	Pb	Co	Cd	Cu
Moisture <b>%</b>	0.68	0.85	0.75	0.86	0.68	0.82	0.89	0.75	0.86
SS	0.55	0.58	0.50	0.58	0.32	0.40	0.44	0.50.	0.33
Salinity	0.66	0.71	0.64	0.69	0.38	0.50	0.60	0.48	0.46
0.C. %	0.75	0.86	0.71	0.84	0.77	0.84	0.87	0.79	0.79
Phosphorus	0.65	0.76	0.57	0.73	0.75	0.81	0.82	0.74	0.79
<16 μm %	0.76	0.84	0.76	0.86	0.72	0.83	0.88	0.77	0.82
Cu	0.61	0.82	0.60	0.79	0.88	0.93	0.89	0.87	
Cd	0.66	0.87	0.62	0.82	0.88	0.89	0.84		
Co	0.75	0.91	0.83	0.92	0.76	0.86			
Pb	0.73	0.87	0.61	0.83	0.88				
Zn	0.57	0.74	0.46	0.66					
Ni	0.84	0.96	0.87						
Mn	0.76	0.81							
Cr	0.83								
SS = Suspend <16 µm = 1	ded solids; percentage o	0.C. f <16 p	= Organic m size fr	carbon .					
•		•							

movement. However a depositional environment existed at the lower reaches of the estuary under the influence of tidal action. Lack of a highly significant relation between salinity and total content of zinc, copper, cadmium and lead suggests that there areother important factors, other than hydrualic processes described above which influenced the distribution of these trace metals in this estuary. As in the case of salinity, the suspended solid particles in bottom waters also exhibited strong and positive relationship with total iron, chromium, manganese and nickel, but weaker relationship with zinc, lead, cobalt and copper. Several workers (e.g., Jouanneau et al. 1983; Satyanarayana et al.1985b) have pointed out that the metals iron, chromium, nickel and manganese are mostly associated with fast settling denser suspended particles whereas zinc, copper, cobalt etc., are more often associated with lighter particles of biogenic origin. The significant relationship between the trace metals and organic matter in sediments was due to the simultaneous coagulation and settling of metals and organic matter derived from urban sewage or industrial effluents.

A high degree of correlation and the significant regression relation among the nine metals studied, thus, established identity in the sources, transport and deposition of these metals within the estuary. Among the metals compared, the weakest relationship with other metals was recorded for zinc, followed by copper, cadmium and lead. This pointed out additional sources for the group of metals zinc, copper, cadmium and lead, different from that for the group consisting manganese, nickel, chromium and cobalt. of iron, Spatial distribution studies had revealed that this second group of metals were distributed in comparatively higher levels in Segment I than in Segment III whereas the metals of the first group were found to be in higher (e.g., zinc and cadmium) or in equal (e.g., lead and copper) levels in the above segments. The

results confirmed anthropogenic inputs of zinc and cadmium into the northern part of the estuary reported by earlier workers (Paul and Pillai, 1983a; Ouseph, 1987; Nair et al. 1990).

3.3.5 THE GEOCHEMICAL BACKGROUND

The ranges of metal values reported from the Cochin estuary by previous workers are shown in Table 3.10. A direct comparison of the data presented in the table was not possible as the earlier studies were confined to isolated parts of the estuary. Murty and Veerayya (1981) surveyed the southern parts of the estuary in 1969 and their survey extended beyond the present study area in the south and hence, part of the data from related to the present study area only, was reproduced their studies in the above Table. The study area of Venugopal et al. (1982) was confined to the northern part of the estuary. Paul and Pillai (1983a) conducted their studies in Periyar river covering a distance of about 50 kms upstream. Only the data related to the estuarine region was reproduced from their studies. The more recent studies (Ouseph, 1987; Nair et al. 1990) covered major parts of southern and northern parts of the estuary, but differed in the total number of samples and In spite of these also in the location of sampling sites. limitations, there is a striking similarity in the data reported by various workers on trace metal content of surficial sediments.

Major differences observed was in the case of iron and Earlier workers reported manganese. comparatively higher values. The major sources of iron and manganese to this estuary were river inputs. The Cochin estuary has undergone significant physical and biological changes during the last three decades (Gopalan et al. 1983). Water flow was diverted from Periyar for agricultural purposes. Marshes have been reclaimed to provide residential areas to the fast growing population. A lot of

ight)	6	erayya, 1981	ichindan, 1984	al. 1982	ai, 1983a		066	Y	uth.
(mg∕kg dry we	Reference	Murty and Ve	· Mallik and Su	Venugopal et	Paul and Pill	Ouseph, 1987	Nair et al. 1	PRESENT STUI	f Cochin barmo
Cochin estuary	Сr		5- 96			7- 130	3- 118	<b>4-</b> 100	part is north o
s of the	Nİ	10- 60	14- 85	22- 82	3 <b>-</b> 45		11- 123	2- 71	northern
n sediment	C	30- 82	2- 42	10- 43			3- 22	<b>1-</b> 25	armouth;
ported ir	Pb		3- 46			10- 90	8- 40	3- 50	Cochin b
ions re	Cd		0.1- 6.5		0.8- 18.0	0.5- 8.4	0.3- 4.0	0.3- 8.4	uth of
entrat:	Сп	3- 100	8 <b>-</b> 63	17- 71	5 <b>-</b> 61		1- 42	1- 50	to So
l conc	Zn		10- 417	103- 664	116- 1385	35- 780	23 <b>-</b> 245	1 <b>4-</b> 259	refers
e meta	чМ	120- 600	<b>44-</b> 1439	73- 318	15- 640			7- 170	part 1
e of trace	Чe	2200- 73700	3100- 90600		4400- 29100			2800- 20200	Southern
3.10 Rang	Location	Southern part	Southern part	Northern part	Northern part	Entire estuary (1985/86)	Entire estuary (1987/88)	Entire estuary (1988/89)	Note:
Table		i)	ii)	iii)	iv)	(	vi)	vii)	

changes have been brought about in the hydrology and hydrography of the estuary by engineering works (e.g., commissioning of the Thanneermukkam bund; inter-basin transfer of Periyar water to Muvattupuzha to facilitate the Idukky hydro-electric project; impoundment of water in dam reservoirs etc.). All these changes were of significance to trace metal behaviour within the estuary, although the links between the specific changes and specific effects were only poorly understood. The construction of dams and impoundment of water reduced considerably the suspended load in river water and this might have had serious consequences in the transport of trace metals to the estuary.

The present data compared well with previous data not only in the ranges of metal values but also in certain spatial and seasonal trends reported. The accumulation of metals in the mid-estuarine region and their depletion towards the marine side were observed by Nair et al. (1990). Venugopal et al. (1982), Ouseph (1987) and Nair et al. (1990) observed monsoonal low values and postmonsoonal enrichment for all the metals they studied, as also recorded during the present study. The downstream depletion of manganese and cobalt in the southern arm of the estuary was previously reported by Murty and Nair et al. (1990) observed comparatively Veerayya (1981). higher levels of manganese, chromium, nickel, cobalt and iron in the southern part in comparison to the northern upper riverine parts. The results from the present study confirmed the above observations.

Table 3.11 compares the metal data from the Cochin estuary (present study ) with that of the more recent data reported in sediments of various estuaries and rivers in India. Average value for riverine sediments and global average value for surface soils also have been furnished. In comparison to other major estuaries and harbours of India, the Cochin estuary was

Table 3.11 Mean trac	ce metal v	values r	eported	in sed	liments of	Indian	estuarie	es and	rivers	(mg/kg dry weight).
Location	не	Mn	ΥZ	Си	Cd	Pb	ទ	Nİ	Cr	References
						           	; ; ; ; ; ; ; ; ; ; ; ; ; ; ; ; ; ; ;			
Ashtamudi, Kerala	1800		60	60		38		12		Nair et al. 1987
Bombay harbour	86000	1140	155	105	10	48	50	111	124	Patel et al. 1985
Cauvery estuary	34000	1310	75	33	1.8	38		379	229	. Subramanian et al. 1989
Ganges estuary	42000	732	151	44		32		49	98	Subramanian et al. 1988
Godavary estuary	76000	1294		119		ß		91	128	Biksham and Subramanfan, 1988
Godavary river	57000	1070	54	82		11	47	51	126	Subramanian, 1987
Krishna estuary	10000	6978	1482	69		4		149	174	Subramanian et al. 1989
Krishna river	25000	906	26	35			32	32	82	Ramesh et al. 1990
Mindhola estuary		1109	157	113		19		92		Zingde et al. 1988
Narmada estuary	89000	1097	140	136				81		Borole et al. 1982
Tapti estuary	76000	1125	125	128				70		- do -
Thana creek	70000	867	146	115	2.3	52	32	117	51	Mahajan et al. 1987
Ulhas estuary	63600	230	149	97	2.3	37	44	93	86	- do -
Visakhapatnam - inner harbour	62600	454	7421	120	43	296	30	50		Satyanarayana et al. 1985b
- outer harbour	58000	490	4240	41	11	104	11	37		- do -
COCHIN ESTUARY	8300	66	70	12	1.7	15	6	21	30	PRESENT STUDY
Indian average (river sediment)	29000	605	16	28			31	37	87	Subramanian et al. 1985
World average (top soils)	40000	1000	06	30	0.4	35	œ	50	70	Martin and Whitfield, 1983

less contaminated with trace metals in surficial sediments. However, the levels of zinc was much above the Indian average for riverine sediments. Cadmium levels were higher than that of surface soils. Nair et al. (1990) compared the trace metal content of sediments of the Cochin estuary with that reported in sediments from other geographical units world-wide, and ranges of concluded that the mean and values recorded in sediments of Cochin estuary was comparable with that reported from other regions. The comparatively lower content of trace metals in the surficial sediments of the Cochin estuary might be due to the fact that the rivers that drain into the estuary traverse highly weathered tropical lateritic surface caps.

The large variations in trace metal distribution observed were primarily due to the variations in particle size, organic matter content, net deposition rates and sediment mixing. The ability of fine particles and organic materials to scavenge trace metals effectively in estuaries has been discussed by several authors (de Groot et al. 1976; Williams et al.1978; Thorne and Nickless, 1981; Olsen et al. 1982; Brook and Moore, 1988). In environments where particle size distribution and mineralogy were heterogenous, substantial variability in trace metal concentration were recorded even in the absence of metal contamination. This variability was further modified in estuaries by the fact that fine particles were more easily resuspended than coarser particles, retained in suspension for longer periods and were swept across wide areas before coming to rest in an environment that was topographically or dynamically suited for fine particles and associated trace metals to be accumulated. In the Cochin estuary, the mid-estuarine reaches in the northern arm provided such a dynamically suited housing excessive trace metal burdens relative to niche, coarser grained areas of the upper and lower estuary.

# CHAPTER 4 TRACE METAL PARTITIONING IN SEDIMENTS

# 4.1 INTRODUCTION

- 4.2 MATERIALS AND METHODS
- 4.3 RESULTS AND DISCUSSION
  - 4.3.1 IRON
  - 4.3.2 MANGANESE
  - 4.3.3 ZINC
  - 4.3.4 COPPER
  - 4.3.5 CADMIUM
  - 4.3.6 LEAD
  - 4.3.7 COBALT
  - 4.3.8 NICKEL
  - 4.3.9 CHROMIUM
### 4.1 INTRODUCTION

Sediments are heterogenous mixtures of dissimilar particles varying in size, composition and origin. Each particle is a complex assemblage of inorganic and organic components such as amorphous oxides, carbonates, organic matter etc. silicates, A quantitative knowledge on the selective distribution of trace metals among the specific components of the sediments is of great practical significance as it serves to evaluate their However, direct methods are fate in the environment. not presently available to quantify these metal distributions. Techniques available for the characterisation of metal binding in well defined media to specific well defined surfaces such as, oxides and organic materials (James and Healy, 1972; Hohl and Stumm, 1976; Davis and Leckie, 1978a), have only limited applications for the aggregated, multiligand, multicomponent surfaces that occur in natural sediments (Luoma and Davis, 1983). Empirical methods suggested for quantifying the various metal associations in sediments include sequential chemical extractions (Engler et al. 1977; Tessier et al. 1979) andstatistical correlations (Luoma and Bryan, 1981 & 1982; Luoma and Davis, 1983). Models describing partitioning in sediments have also been developed (Oakley et al. 1981; Davies- Colley et al. 1984; Nyffeler et al.1984: Tessier et al. 1985; Luoma, 1986; Reuber et al. 1987; Hirose, 1990). Historically, efforts describe quantitatively metal partitioning natural to in sediments have involved a search for extractants which selectively remove specific forms of metals from sediments. Their usefulness have been proved by actual usage rather than by theoretical discussion (Forstner, 1982; de Mora and 1984; Tessier and Campbell, 1991). Harrison, The study reported herein was aimed at demonstrating the usefulness of accumulative the chemical partitioning in delineating the phases of trace metals in surficial sediments of the Cochin estuary, distributed in space and in time. The trace metals studied were iron, manganese, zinc, copper, cadmium, lead, cobalt, nickel and chromium. Although iron and manganese are

not generally referred to as trace metals in the literature, they have been so designated and included in the studies to simplify the presentation of results and to amplify the scope of discussion.

### 4.2 MATERIALS AND METHODS

The details on site selection, sampling and chemical analysis of the samples are furnished in Chapter 2. For partitioning of trace metals into the various accumulative phases, the sequential extraction scheme proposed by Tessier et al. (1979) was used with slight modifications (Nair et al. 1991). There were five steps in the extraction procedure and the metal phase extracted at each stage was defined by the extractant used. Because of this operational definition, the fractions extracted have been referred to (in the text and figures) as fractions 1 to 5 along with the chemical symbol of the element under discussion. The numbers representing the fractions are 2. Carbonate bound 3. Fe-Mn 1. Exchangeable oxide bound 4. Organic/ sulphide bound and 5. Residual. The total metal concentration has been indicated by adding the subscript `T' to the chemical symbol of the metal. Thus, for iron, the various fractions have been described as Fe1, Fe2, Fe3, Fe4 and Fe5 and Several research workers have utilised the total iron as FeT. sodium dithionite as the extractant to extract the moderately reducible phases, and this fraction has been assigned the number 6 in the Tables.

#### 4.3 RESULTS AND DISCUSSION

The summary statistics, indicating the mean, standard deviation, percentage coefficient of variation and range of values for each of the five fractions of the nine metals studied are furnished in Table 4.1. The Table depicts information also on the ratio of the relative contribution of each fraction to the total concentration of the metal concerned. The data showed wide scatter as indicated by the percentage

	Max.	0.23 0.33 3.23 3.30 90 22	23.22 27.06 22.46 25.71	37.96 82.15	13.35 19.70 19.32 43.38 88.77	6.67 9.52 4.44 49.19 89.80
	Min.	0.01 0.00 0.30 0.21	0.90 0.90 1.82 1.41	10.10 27.84	0.24 0.38 2.43 6.71 28.01	0.37 0.32 0.00 7.56 46.57
ace metals	Percentage CV%	75.3 107.7 52.6 64.2	1.2 67.2 52.4	22.9 19.7	124.1 83.3 46.4 41.2 29.5	63.3 79.7 124.1 44.4 13.0
ons of tr	SD 1	0.04 0.07 0.67 0.75	1.21 5.05 4.59 4.67	5.98 9.10	2.185.354.3510.0117.19	1.32 2.17 0.91 9.03 9.66
hemical fracti	Mean	0.06 0.07 1.27 1.16	9.44 7.52 11.16 8.91	26.11 46.30	1.75 6.43 9.37 24.28 58.22	2.08 2.72 0.74 20.34 74.21
of different c	Max.	18.70 54.00 262 389	19301 33.50 34.10 21.10	53.60 81.70	13.16 37.42 46.19 81.40 91.01	$\begin{array}{c} 1.12\\ 0.96\\ 0.19\\ 15.89\\ 34.46\end{array}$
it ribut ion	kg Min.	0.40 0.30 16 12	2/08 0.30 0.30 0.10	1.50 2.60	0.07 0.20 0.77 1.15 5.85	0.02 0.06 0.00 0.16 0.94
cs on dis	ation mg/ CV%	74.1 160.6 92.3	47.2 132.4 93.7 91.1	79.8	157.6 $146.2$ $131.9$ $115.3$ $71.9$	76.7 70.3 89.9 119.5 96.3
y statisti	Concentr SD	3.46 10.60 63 99	3020 6.34 8.16 5.33	14.29 20.19	1.88 9.97 10.71 23.87 24.14	$\begin{array}{c} 0.13\\ 0.13\\ 0.04\\ 2.99\\ 8.96\end{array}$
Summar	Mean	4.66 6.60 103 108	6103 4.79 8.71 5.86	17.91 28.77	1.196.828.128.1220.7133.57	$\begin{array}{c} 0.17\\ 0.18\\ 0.04\\ 2.50\\ 9.31 \end{array}$
Table 4.1	Fraction	Fe1 Fe2 Fe3 Fe4	reo Mn1 Mn3 Mn3	Mn4 Mn5	Zn1 Zn2 Zn3 Zn4 Zn5	Cu1 Cu2 Cu3 Cu4 Cu5

Contd...

		Concentr	ation mø/k				Pe	rcentage		
Fraction	Mean	SD	CV\$	<sup>6</sup> Min.	Max.	Mean	SD	ငပနိ	Min.	Max.
Cd1	0.07	0.05	72.8	0.01	0.24	5.76	3.07	53.3	1.40	18.89
Cd2	0.08	0.06	71.1	0.01	0.26	7.48	4.84	64.8	0.68	22.58
Cd3	0.11	0.12	112.3	0.02	0.86	8.13	4.45	54.7	1.72	26.09
Cd4	0.68	0.96	140.6	0.07	3.83	34.90	8.94	25.6	15.22	59.53
Cd5	0.73	0.85	116.1	0.14	4.55	43.73	9.13	20.9	23.33	69.35
Pb1	0.37	0.26	69.4	0.05	1.13	2.86	1.59	55.6	0.60	7.55
Pb2	0.55	0.42	75.6	0.04	2.13	4.32	2.57	59.5	0.94	15.47
Pb3	0.75	0.72	96.5	0.05	4.93	5.68	4.08	71.8	1.04	20.94
Pb4	1.16	0.65	56.2	0.26	3.20	10.37	6.58	63.5	1.39	28.63
Pb5	12.03	10.05	83.5	2.25	45.45	76.78	10.24	13.3	52.47	92.70
Co1	0.18	0.12	65.3	0.02	0.72	2.40	1.20	49.9	0.57	5.69
Co2	0.16	0.11	68.4	0.02	0.57	2.35	1.86	79.1	0.37	9.95
Co3	0.29	0.17	59.2	0.03	0.86	4.09	2.39	58.3	0.49	13.59
Co4	1.81	1.39	76.9	0.27	6.32	19.81	4.42	22.3	10.37	30.47
Co5	6.80	5.34	78.6	0.79	18.98	71.35	6.17	8.6	55.77	83.51
Nil	0.23	0.15	65.0	0.04	0.89	1.45	0.68	46.8	0.34	3.57
Ni2	0.22	0.14	62.6	0.02	0.66	1.40	0.96	68.1	0.19	5.08
Ni3	0.35	0.24	69.5	0.02	1.12	1.98	0.97	49.0	0.42	4.99
Ni4	2.58	2.79	108.5	0.20	11.56	10.94	3.79	34.6	3.05	21.85
Ni5	17.77	14.70	82.7	1.52	58.91	84.23	4.25	5.0	71.39	92.04
Cr1	0.20	0.11	57.2	0.02	0.49	0.95	0.73	76.7	0.08	5.19
Cr2	0.12	0.06	50.3	0.02	0.27	0.64	0.53	83.5	0.08	3.28
Cr3	0.04	0.03	83.2	0.01	0.16	0.16	0.09	58.4	0.03	0.46
Cr4	4.30	5.48	127.5	0.12	22.57	11.01	6.64	60.3	2.24	27.63
Cr5	25.34	19.99	78.9	3.73	81.14	87.25	6.22	1.1	10.17	00.06

Table 4.1 contd....

coefficient variation, as for example, a variation of 160.6 % from mean was indicated for Fe2 and 157.6 % for Znl. The variability could be attributed to the total metal concentration, the site of sampling and time of sampling. Τo test this hypothesis the data was subjected to the analysis of The technique used was the one usually adopted variance. for analysis of data from experiments of asymmetric factorial designs. In this case three factors were considered, the fractions, the stations and seasons. The monthly data were considered as replicates, and were pooled into four seasons, namely, winter, premonsoon, monsoon and postmonsoon as detailed previously. For each metal, 600 cases were processed. The two way interactions between the factors considered were worked out. The results revealed highly significant variation due to fractions (total concentration), seasons (temporal) and stations (spatial) for all the nine metals studied. Interactions between the factors tested were also highly significant. Therefore a direct interpretation \_based on mean values was only of limited value (Snede) far and Kochran, 1968). Hence, in the following pages the results obtained for each metal have been discussed season-wise and station-wise in the order of iron, manganese, zinc, copper, cadmium, lead, cobalt, nickel and chromium.

# 4.3.1 IRON

In estuarine studies iron has received more attention than any other transition metal (Aston, 1978) owing to the possibility of formation of new solid phases during estuarine mixing.Oxidation of iron(II) to oxides and hydroxides is one of the most likely reactions of this type to occur when the pH of the river water is raised by mixing with sea water (Boyle et al. 1977; Mayer, 1982). The formation of these oxides are considered to be very important in controlling the uptake of other metals from solution (Bowers and Huang, 1987). Gibbs (1973) observed that iron coatings which covered the suspended grains of Yukon and Amazon estuaries were important in the transport of manganese, nickel, cobalt, chromium and copper. This incorporation was attributed to co-precipitation during iron oxide formation. Published data (Coonley et al. 1971; Windom, 1975; Holliday and Liss, 1976; Boyle et al. 1977; Murray and Gill, 1978; Eaton, 1979a; Hart and Davies, 1981; Morgan, 1982; Sharp et al. 1984) indicate that estuaries are efficient traps for iron, implying that for iron the seaward losses are unlikely (Helz and Sinex, 1986). Iron gets flocculated in the early stages of estuarine mixing, but the eventual removal from the estuary may be by currents and accompanying turbulent processes.

The levels of total iron recorded in the sediments of the Cochin estuary during the present study ranged from 2811 to 20183 mg/kg (Table 3.7). 93 to 99 percent of the total iron was found to be in the Fe5 phase (Table 4.1) which was generally considered as environmentally inert. Though the relative contributions from other fractions of iron were quite small, their distributions in absolute terms were quite recognisable. The various forms of iron ranked in abundance in the order Fe5>>> Fe4 = Fe3>> Fe2 = Fe1. Figure 4.1 depicts the spatial and seasonal mean distribution pattern of the five fractions of iron in absolute concentrations as well as in relative terms (as pecentage of total).

The Fel values ranged from 0.40 to 18.7 mg/kg dry weight contributing only about 0.01 to 0.23 percent of the total iron present in sediments (Table 4.1). Maximum value was recorded in the silty clay sediments of Segment II and minimum in the coarse sandy sediments of Segment III (Fig. 4.1). The values decreased towards the marine side in Segment II and towards riverine side in Segment III. There was more or less equitable distribution of Fel among the four Stations in Segment I, but a significant seasonal pattern was evident with very high values being recorded during the postmonsoon season except at Station 1, where the highest value was recorded during the monsoon In Segment II, higher values were noted during the season.





premonsoon season. In the extraction procedure, Fel phase includes the iron dissolved in interstitial water as well as the iron loosely bound to adsorbing surfaces which could be replaced by ammonium ions. It has already been recognised that the interstitial water contains more amounts of dissolved iron than the overlying waters (Brannon et al. 1977) and in sandy sediments the exchange of metals between the interstitial water and overlying water will be more rapid than in clayey environments. This, together with the presence of high organic matter which mobilise iron under mildly reducing conditions explain the higher amount of Fel in the silty clay sediments of the mid-estuary. A diminishing trend towards the marine side within Stations of Segment II was noted. This might be due to the increasing effects of salinity which suppressed iron solubility (Sholkovitz, 1976). Enhanced biological activity under the influence of increasing temperature and abundant organic detritus was reported from Segment II of the estuary (Anirudhan, 1988) during the premonsoon season. Hence enhanced values for the Fel during the premonsoon season, when the salinity of water was maximum, were regarded as indicative of the profound influence of biological factors over the physical factors in controlling iron solubility.

The Fe2 values ranged from 0.30 to 54 mg/kg contributing a maximum of 0.33 percent of the total iron (Table 4.1). The highest value was recorded at Station 6 followed by Station 7, with values declining sharply on either sides of Station 6. The relative percentage abundance of Fe2 also exhibited sharp declining trend on either sides of Station 6. In Segment I the percentage values decreased towards the riverine side (Stations 4 to 1). The higher values recorded during the monsoon and postmonsoon seasons, at Stations 6 and 7 were a feature noted only in the case of Fe2 distribution. No distinguishable seasonal trends in Fe2 distribution could be observed at other Stations.

Very high amounts of Fe2, both in absolute and relative terms recorded during the monsoon and postmonsoon periods suggested that this fraction was associated with the suspended load derived from terrestrial sources under the influence of storm water and erosion. Further, Station 6 in the northern arm of the Cochin estuary was reported to be а sheltered sedimentary environment even during the monsoon periods (Pillai, 1989). In the highly acidic soils of the tropics, the fixation of phosphate as iron phosphates became a well established phenomenon (Le Mare, 1981). On the basis of all these observations, it was concluded that the abnormally high values of Fe2 recorded at Station 6 during the monsoon and postmonsoon might be due to the trapping of eroded soils rich in iron phosphates. However it was noted that the Fe2 fraction was subjected to rapid diagenetic changes after deposition as was evidenced by the decreasing percentage values after the monsoon season, most probably under the influence of saline water which could change the solubility of phosphates [as demonstrated by Bale and Morris (1981) in laboratory simulation studies] or unquanti + fied biological activity [as shown by Hines et al. (1984) in the New Hampshire sediments] resulting in mineralisation of phosphates.

The Fe3 values ranged from 16 to 262 mg/kg contributing 0.30 to 3.23 percent of total iron (Table 4.1). Higher values were again recorded at Stations 6 and 7, with values declining on either sides of Station 6. Percentage wise, the maximum contribution was found to be in the sediments of Station 7 which received the sewage from the metropolitan city of Cochin. One characteristic feature noted in the distribution of Fe3 was the downstream declining trend both in absolute and relative values, in Segment I, but with significant seasonal trends. The Station 5 at barmouth recorded lower values than Station 6. Fletcher et al. (1983) also observed lower concentration of hydroxylamine hydrochloride extractable iron in the salt water intrusion zone in the Fraser estuary and they attributed this to iron desorption at intermediate salinities. However, a downstream increase was noted in Segment III, which was quite perceptible during the winter and premonsoon seasons when the river flow was low.

The extractant used for the extraction of Fe3 was acidified hydroxylamine hydrochloride (pH = 2.0) which could dissolve the oxides of iron and manganese that were relatively stable under oxidising conditions (Engler et al. 1977). Brannon et al. (1977) noted that the extractant dissolved approximately 85 percent of manganese oxides while dissolving only 5 percent of the total iron present in Mobile Bay sediments. Tessier et al. (1979) also observed that manganese oxides were more easily leached down than iron oxides. The iron oxides occur in estuarine sediments in a continuum of states ranging from a highly amorphous (or microcrystalline) oxyhydroxide which form upon precipitation to increasingly crystalline solids which form as the initial precipitate ages (Forstner, 1979c). The highly oxidised conditions and the turbulence restricted the amount of soluble iron reaching the system, and this explains the comparatively low values recorded for Fe3 in surficial sediments of the Cochin estuary. The enhancement of Fe3 downstream in Segment III, especially during the winter and premonsoon seasons could be attributed to the rapid precipitation under the influence of low salinities experienced at these stations under conditions of tidal incursions. The chemistry of Fe3 in Segments I and II appeared to be controlled by the presence of manganese oxides and organic matter. The influence of organic matter buried into the sediments might account for the high Fe3 values recorded at Stations 6 and 7 which received sewage effluents. In Segment I, a downstream decrease was noted for Mn3 also, as observed for Fe3 and both might be related to processes which resulted in the mobilisation of iron and manganese from sediments, as demonstrated by Fletcher et al. (1983) for the Fraser estuarine sediments.

The mean Fe4 values ranged from 12 to 389 mg/kg contributing 0.2 to 3.3 percent of total iron (Table 4.1). The range of

values recorded was comparable with that of Fe3, but the spatial and seasonal distribution pattern was different for these two fractions of iron. Higher values were, again, recorded at Station 6 and the content of Fe4 decreased on either sides of this Station in the northern arm of the estuary. In Segment I all the four Stations recorded more or less similar content of Fe4. In percentage-wise distribution, the maximum contribution from this fraction of iron was recorded at Station 7, which was a noticeable sewage effluent receiving site. There was a decrease in relative values on either sides. Seasonwise, higher values were recorded during the postmonsoon season in most of Stations in Segments I and III, both in absolute and relative values. A distinct seasonal trend was not observed at Stations in Segment II.

The two important features noted in the distribution of Fe4 were the enhanced values in Stations of Segment II and the relative enrichment in sediments during the postmonsoon The sewage effluent being received in Station 7 might periods. be an important factor in controlling the iron fractionation in this phase. However, in other two Segments, seasonal importance of biological productivity reported by Joseph and Pillai (1975) and Joseph and Nair (1975)could be the controlling factor of iron fractionation as evidenced by enhanced values during the postmonsoon season.

The Fe5 values ranged from 2768 mg/kg to 19561 mg/kg, 93 to 97 percent of the total contributing iron. Its distribution pattern mimicked that of total iron and was greatly influenced by the sedimentary environment. Thus higher values were recorded in the accumulative environments of Segment II and lower values in the erosional environments of Segment III. As noted for total iron, the seasonal trends were subdued or indistinct.

The Fe5 fraction represented the metal associated with primary minerals, secondary minerals and refractory organic

materials (Engler et al. 1977; Tessier et al. 1979). The major part of total iron was found in the Fe5 fraction which was regarded as chemically inert and biologically inactive and this explained the minimum seasonal variation observed in the case of total iron distribution in estuarine sediments.

Table 4.2 furnishes data on chemical partitioning of iron (reported in percentage) from other parts of the globe. It can be seen from the table that the range of values recorded for Fel, Fe2, Fe3 and Fe4 during the present study in the Cochin estuary compared well with values reported from other parts of the world. The difference noted in the case of Fe5 fraction could be traced to the differentiation into moderately reducible and residual fractions, as done by some workers (Brannon et al. 1977; Patchineelam and Forstner, 1983).

### 4.3.2 MANGANESE

Manganese is an essential element in biological systems. Availability of manganese from soils to plants depends upon a number of factors, like, the concentration in soil solution, concentration on exchange sites, pH, Eh etc. Manganese contamination is relatively insignificant except for the common biological toxicity effects found in plants exposed to excess of natural manganese in the soil under some peculiar conditions, particularly flooding. In oxic environments, it is present mainly as manganese (IV) oxides and oxyhydroxides. Under anoxic conditions, as in flooded paddy-soils, manganese (IV) compounds are readily reduced to manganese (II) and the process is considered to be both chemical and biochemical (Ponnamperuma, 1972). The manganese (II) when diffused or advected into oxic environments gets oxidised to manganese (IV). Manganese (II) was found to react with carbonates and bicarbonates and the chemistry of manganese carbonate has an important role in controlling the dissolved manganese concentration in soil solution and interstitial waters of sediments (Smith, 1990).

Table 4.2 Percentage of	iron fre	actions r	eported	in sedime	ants from	differe	nt geogra	phical locations
Location	Fe1	Fe2	Fe3	Fe4	Fe5	Fe6	FeT (mg/kg)	Reference
Yukon-SPM	0.01		40.60	11.00	48.20		63000	Gibbs, 1973
Mobile bay, Site 1	1.35		4.10	12.67	9.11	72.75	28000	Brannon et al. 1977
Site 2	0.97		2.28	19.41	8.79	68.54	30700	- do -
Yamaska river - SPM	0.01	2.10	32.00	4.80	61.00		43800	Tessier et al. 1980
St. Francois river - SPM	0.01	1.30	22.00	3.80	73.00		42700	- do -
Rhine river	0.10	2.00	0.10	47.10	50.80		23300	Forstner, 1982
Subae river-unpolluted			3.00		83.00	14.00	26161	Patchineelam and Forstner, 1983
-polluted			3.60	0.20	26.80	69.40	48540	- do -
Yellow river	0.10	1.00	1.00	7.00	65.00	26.00	14198	Hong and Forstner, 1983
Axios river	0.14	1.60	18.81	37.06	42.39		17370	Samanidou and Fytianos, 1987
Axios estuary.	0.19	0.96	18.31	37.28	43.26		15640	- do -
Aliakmon river	0.14	0.65	26.82	46.39	25.90		70300	- do -
Aliakmon estuary	0.23	3.65	31.61	40.05	24.46		65900	- do -
COCHIN ESTUARY-MEAN	0.06	0.07	1.27	1.16	97.44		8323	PRESENT STUDY
-RANGE	0.01 - 0.23	0.01- 0.33	0.30- 3.23	0.21- 3.30	93.25- 99.22		2800- 20200	- do -

Reactivity of manganese in estuarine systems has been widely studied. Manganese has been reported to exhibit a complex behaviour with removal, addition as well as conservative nature for different estuaries (Burton and Liss, 1976; Evans et al. 1977; Yeats et al. 1979; Morris and Bale, 1979; Wollast et al. 1979; Wilke and Dayal, 1982). Its mobilisation from sediments to overlying water has also been investigated (Graham 1976; Lu and Chen, 1977; Lacerda et al. et al. 1987). Manganese can occur in sediments as discrete oxide particles, as manganese (II) coatings on particles and adsorbed on organic materials (Graham et al. 1976; Feely et al. 1982). In estuaries organic matter accumulation may lead to the dissolution of manganese from sediments (Luoma, 1990). Since manganese can apparently be complexed by dissolved organic materials and can be retained in solution by the control of redox potential through oxidation of the more labile dissolved organics, the amount and kind of organic material present would determine the proportion of manganese exported from the estuary (Wangersky, 1986). Delfino and Otto (1986) observed that in the Chesapeake bay, the phase distribution patterns for manganese contrasted sharply with those of Fe. While iron was preferentially bound (89-99 percent) to the particulate phase, 75 percent of manganese was present in the soluble phase in Susquehanna river during fall, winter and the snowmelt runoff periods and about 50 percent was soluble in summer. It is difficult to understand the behaviour of manganese at the estuarine sediment/water interface because factors such as physical mixing, reactions between suspended loads and dissolved materials and input of pollutants complicate the interpretation of the data.

In the Cochin estuary the abundance of manganese on an average was only next to zinc. The values for total manganese in sediments varied from 7 to 170 mg/kg with a mean value of 66 mg/kg (Table 3.7). In contrast to the behaviour of iron, a considerable portion of manganese recorded in this estuary was found to be in the environmentally labile fractions (Table 4.1)

leaving an average of only 46 per cent as the residual This was indicative of the high chemical reactivity fraction. of this transition element in this estuary, which was noted in other estuaries also (Evans et al. 1977; Wollast et al. 1979; Rosental et al. 1986; Prohic and Kniewald, 1987; Samanidou and The order of abundance of the Fytianos, 1987). various fractions of manganese was Mn1< Mn3< Mn2<< Mn4<<< Mn5. It was also noted that the percentage coefficient variations decrease from Mn1 to Mn5 indicating that the less firmly held fractions of Mn exhibited wider variability in distribution.

The spatial and seasonal mean distribution of the various fractions of manganese determined in the surficial sediments of the Cochin estuary are presented in Fig. 4.2. The values for Mn1 fraction ranged from 0.3 to 33.5 mg/kg, contributing 0.9 to 27.1 percent of total manganese (Table 4.1). Higher values were recorded (Fig. 4.2) at Station 1 during monsoon and postmonsoon periods with values decreasing sharply downstream. Very low values of Mn1 were recorded at the three Stations in Segment III, except for the premonsoon period. During winter and premonsoon periods more or less similar values were recorded for all the Stations in Segments I and II . An unique feature observed in the case of Mn1 distribution was the absence of a mid-estuarine maxima. Percentage-wise distribution of Mn1 showed gradually decreasing contributions in the seaward direction, throughout the year, on both the arms of the estuary, the Stations 5 and 6 nearer to barmouth recording the minimum contribution.

The more or less equitable distribution of Mn1 fraction between the sediments which differed greatly in their grain size could be explained on the basis of the estuarine reactivity of manganese. Though the hydrogenous precipitation of the oxides of iron and manganese and the co-precipitation of trace metals were regarded as means of removal of trace metals from river water in the early stages of estuarine mixing, many workers (Holliday and Liss, 1976; Aston, 1978; Duinker, 1980;





Duinker et al. 1982; Gibbs, 1986; Nolting et al. 1990; Windom et al. 1989) observed variations in the rate of removal of various metals. Although the removal of iron was quite rapid and complete, the removal of manganese was not so. Fukai and Huynh-Ngoc (1968) demonstrated that the manganese precipitation required conditions of higher рH than those normally encountered in estuaries. The preponderance of soluble species of manganese in riverine sections, especially during period of high runoff, also has been reported (Delfino and Otto, 1986). The increased amounts of soluble manganese present in the riverine parts of the estuary compensated for the decreased number of sites involved in cation exchange in the sandy On the other hand the limited sediments. solubility of manganese under saline conditions restricted the availability of manganese ions which might be involved in exchange reactions in the silty clay sediments of Segment II. The anomalous values of Mn1 recorded at Stations 1 and 2 during monsoon and postmonsoon could be attributed to the impounding of water beyond the Thanneermukkam bund (see, Fig. 2.1) and its release to the estuary with the commencement of monsoons. The effect of impoundment in enhancing the manganese solubility is well documented (Delfino and Lee, 1971; Eaten, 1979b).

The Mn2 fraction in the sediments varied from 0.30 to 34.1 mg/kg contributing 1.8 to 22.5 percent of total manganese Higher values were recorded at Station 6 (Table 4.1). of Segment II, with values decreasing sharply on either side, in absolute as well as relative amounts (Fig. 4.2). Mn2 exhibited a downstream decrease in Seqments I and III at all seasons, both in absolute and relative terms. Season-wise, higher values were recorded during the premonsoon period and lower during monsoon/ postmonsoon at Stations of Segment II and III, but significant interactions between Stations and Seasons were noted in the seasonal distribution of Mn2 in Segment I.

Ponnamperumai (1984) reported that there was a decrease in the water soluble manganese after an initial rise following

flooding and this was attributed to the precipitation of manganese as manganese carbonates. Therefore the sediments of the impounded water which was released to the southern parts of the estuary with the commencement of monsoon might be a contributing factor for the high amounts of Mn2 detected in the sediments of southern parts. Within the estuary, this fraction of manganese was subjected to rapid conversion either under the influence of biological agencies (as evidenced by the increasing percentage of Mn4 fraction) or geochemical processes (as evidenced by increased percentages of Mn5 fraction). In Segment II, the enhanced values during the monsoon season might be, as explained for Fe2 fraction, due to the influence of terrestrial soil particles that contained manganese bound to phosphates and settled preferentially in these sheltered sedimentary environments. Once settled, these phosphates released manganese under diagenetic influences, thus showing the monsoonal peak and decreasing trend afterwards.

The content of Mn3 fraction in surficial sediments varied from 0.1 to 21.1 mg/kg contributing 1.41 to 25.71 percent of total manganese (Table 4.1). Sediments of Segment II recorded the maximum value and the values were significantly higher during the winter and premonsoon months (Fig. 4.2). As observed for Mn1 and Mn2 fractions, Station I in Segment Ι recorded relatively higher values during the monsoon and postmonsoon In general, the values of Mn3 exhibited a periods. down-river declining trend in Segments I and III. Percentage-wise, the maximum contribution from this fraction of manganese was recorded at Station 7 during the postmonsoon and winter Sporadic high percentages were observed at Station 9 periods. which received industrial effluents. Sediments of Segment Ι had the lowest contribution from this fraction of manganese. Α downstream decrease in the relative contribution of Mn3 was observed during the premonsoon and monsoon in Segment I and during the winter, monsoon and postmonsoon in Segment II.

The distribution pattern of Mn3 in Segment I was similar to that of total manganese indicating that a definite portion of manganese delivered to this part of the estuary was in the Mn3 The sporadic high values recorded for Mn3 at Station 9 form. had to be attributed to industrial effluents which might cause rapid precipitation of manganese oxides or to the addition of sand sized particles coated with manganese oxides. The higher content of Mn3 recorded at Stations in Segment II during the winter and premonsoon periods was due to the prevailing saline conditions which might cause the dissolved manganese to precipitate and flocculate.

The values of Mn4 fraction were found to be in the range of 1.5 to 53.6 mg/kg contributing 10 to 38 percent of total manganese (Table 4.1). Higher values were recorded at Stations 5 and 6 of Segment II during the winter and post-monsoon periods (Fig. 4.2). The premonsoonal values were higher at all stations except at Stations 1 and 2 in Segment I. The sediments of Segment III had the lowest content of Mn4 at all A downstream decrease was noted in the content seasons. of Mn4, in Segment I during postmonsoon and winter periods. The highest relative contribution from this fraction was observed in Segment I during the premonsoon periods. For the Stations in Segment II also, the higher relative contribution was recorded during winter and premonsoon periods. One interesting feature noted was the minimum variation observed in percentage values between Stations in Segments I and II. However, wide seasonal variation was indicated in Segment III, the higher percentage values being recorded during monsoon and post-monsoon periods.

The enhanced values of Mn4 during the winter and premonsoon periods could be attributed to the enhanced biological activity which resulted in the seasonal accumulation of autochthonous organic residues. The minimum variation in relative values of Mn4 indicated that sediments in an equilibrium was attained between the precipitation and

remobilization of this fraction of manganese. The influence of terrestrial inputs of organic bound manganese was indicated bv the high relative values recorded at the riverine Stations of Segment III during the monsoon and postmonsoon periods. The organic matter status of sediments had a profound influence on the distribution of Mn4 fraction, as was indicated by the strong relation between sediment organic carbon and Mn4 fraction.

The content of Mn5 fraction varied from 2.6 to 81.7 mg/kg, contributing 29 to 82 percent of total manganese in surficial sediments. In absolute terms, higher values were recorded at Stations in Segment II; the values exhibited high seasonal variation with monsoonal low and premonsoonal high values (Fig. 4.2). During the premonsoonal period, the values increased down-stream in Segment I, but during the other seasons, а down-stream decrease was indicated. In percentage-wise distribution, a gradual increase was noted for this fraction in Segment I and no trends could be detected in Segment II. Tn Segment III, a declining trend down-river was indicated during the winter and premonsoon periods and an increasing trend was indicated during the other periods.

In Segment I, a down-stream increase was observed in the percentage distribution of Mn5, whereas, the total manganese content showed a decreasing trend. A logical conclusion from this observation was that the more labile and active forms of manganese underwent changes in the sediments yielding the more resistant forms. A simultaneous decrease observed in the relative distribution of Mn1, Mn2 and Mn3 fractions also substantiated this view. It had been pointed out by several workers that microorganisms in the sediment/water interface would play a pivotal role in the transformation of these fractions (Ehrlich, 1980). The dissolution and removal of labile phases of manganese might also cause a net reduction in the total manganese, at the same time enhancing the relative contribution of Mn5 phase. Manganese substitution in the two

layered silicate minerals could be another factor in the enrichment of residual fractions in an estuarine environment. The decrease in the percentage contribution of Mn5 fraction in the sediments of Segment II was attributed to the precipitation of authigenic fractions (as indicated bv the enhanced percentages for Mn2 and Mn3 fractions at these Stations) which enriched the total manganese load in sediments bringing down the percentage contribution of Mn5. The behaviour of manganese at Station 9 might be due to industrial effluents that influenced the fate of manganese.

Table 4.3 furnishes data on the chemical partitioning of manganese reported from other parts of the globe. All the values are given in percentages. The mean and range values of the data recorded during the present study are also furnished for comparison. It can be seen that the data recorded for the Cochin estuary, for the fractions Mn 1, Mn2, Mn3 and Mn4, are within the range reported from other centres. The difference in the values for Mn5 fraction was mainly due to the extraction if of moderately reducible phase by some workers, and the values for Mn5 and Mn6 were summed up, then the difference would be marginal.

A comparison of the selective chemical extraction data for manganese and iron showed that there were greater differences in fractional distribution of manganese than for iron. Appreciable and roughly equivalent quantities of manganese were found in Mn1, Mn2 and Mn3 fractions, in the sediments of the Cochin estuary, and this was in unison with the observations recorded by Tessier et al. (1980) in the sediments from Yamaska and St. Francois rivers. An appreciable quantity of manganese was found to be associated with the organic phase (Mn4). Similar features in Mn4 distribution were observed in the sediments of Mobile Bay (Brannon et al. 1977) and Axios and Aliakmon estuaries (Samanidou and Fytianos, 1987), as well. An equitable distribution of manganese in various accumulative phases was observed, also in sediments from Puget Sound,

Table 4.3 Percentage c	of mangan	ese fract	ions rep	orted in	sediments	from	different	geographical locations.
Location	Mn1	Mn2	Mn3	Mn4	Mn5	Mn6	MnT (mg/kg)	Reference
Yukon-SPM	0.50		45.70	6.60	37.10		1270	Gibbs, 1973
Mobile bay, Site 1	14.93		35.50	35.53	8.07	5.97	1069	Brannon et al. 1977
Site 2	7.88		14.92	44.36	22.12	10.74	638	- do -
Yamaska river-SPM	29	23	24	1.70	22		1250	Tessier et al. 1980
St. Francois river-SPM	26	20	28	1.4	25		1400	- do -
Rhine river	8.5		32.4	3.8	17.2	38.1	583	Forstner, 1982
Subae river-unpolluted			45.8		47.1	7.1		Patchineelam and Forstner, 1983
-polluted			58.8	0.38	c,	34.40		- do -
Yellow river	6.5	31.0	8.0	17.0	27.0	11.0	283	Hong and Forstner, 1983
Keum estuary	3.8		72.4	5.3	18.5			Lee, 1985
Krka estuary – range	12- 35	45- 78	bdl- 15	bdl- 11	1- 6		700- 2400	Prohic and Kniewald, 1987
Axios river	7.24	19.64	27.36	22.44	23.32		984	Samanidou and Fytianos, 1987
Axios estuary	5.57	21.94	27.55	23.99	20.95		815	- do -
Aliakmon river	3.26	23.15	32.91	30.69	9.99		534	- do -
Aliakmon estuary	3.95	23.33	32.67	31.23	8.82		504	- do -
COCHIN ESTUARY-MEAN	7.52	11.16	8.91	26.11	46.30		99	PRESENT STUDY
– RANGE	0.90- 27.06	<b>1.82-</b> 22.46	1.41 - 25.71	10.10 - 37.96	27.84- 32.15		7- 170	- do -

bdl = Below detection limit

Washington (Feely et al. 1986) and Amazon river, Brazil (Lacerda et al. 1990). The appreciable contribution of Mn1 and Mn2 fractions revealed that, contrary to iron, an important proportion of manganese was present in reduced forms. Thig corroborated the observation that in most natural waters manganese(II) oxidation was a much slower process than iron(II) (Stumm and Morgan, 1970). Because of the oxidation large amount of exchangeable manganese, a significant percentage of sediment manganese would be solubilised following an increase in salinity. The spatial distribution trend in the percentage values of Mn1 recorded during the present study emphasised the importance of salinity in regulating the exchangeable fractions of manganese. A decrease in relative abundance with increase in salinity was observed also for Mn3 and Fe3 fractions. The relative high abundance of less firmly held fractions of manganese indicated that this metal was predominantly associated with the labile fractions.

## 4.3.3 ZINC

Zinc plays an important role in the metabolic cycles of humans, animals and plants as a catalytic and structural element in numerous enzymes involved in energy metabolism. However, in higher quantities zinc is potentially hazardous to biota. The main pollutant sources of zinc are metalliferous mining activities, ore-dressing and processing, agricultural use of sewage sludge and composted materials, and the use of agro-chemicals such as fertilizers and pesticides. Kiekens (1990) furnished an excellent summary of the environmental chemistry of this element.

A number of studies have been reported on the behaviour of zinc in estuaries; the behaviour of the metal has been regarded as conservative in large unpolluted estuaries (Holliday and Liss, 1976; Elderfield et al. 1979; Danielsson et al. 1983; Paulson et al. 1989) and nonconservative in polluted estuaries (Duinker and Nolting, 1982; Duinker et al. 1985; Campbell et

al. 1988). Considerable anthropogenic input of zinc into the reported Cochin estuary was by earlier workers (Sankaranarayanan et al. 1978; Paul and Pillai, 1983a; Ouseph, Speciation studies 1987; Nair et al. 1990; Shibu et al. 1990). on zinc in estuarine waters (Shibu et al. 1990) have shown that the river Periyar received considerable quantities of labile zinc with high zinc-organic matter associations. In freshwater most of the zinc was found in the dissolved phase where as in estuarine waters, the relative amount was found reduced (Shibu et al. 1990). Nair et al. (1990) observed significant seasonal and spatial variation in the distribution of zinc in surficial sediments. Chemical partitioning of zinc in surficial sediments (Nair et al. 1991) revealed that the various fractions of zinc were widely distributed in the Cochin estuary.

In estuarine sediments of the Cochin estuary, zinc ranks in abundance next only to iron. Total zinc (ZnT) varied from 14 to 259 mg/kg (Table 3.7). The high reactivity of zinc in the estuary was indicated by the high variability recorded for all zinc determined (Table 4.1). fractions of As noted for manganese the percentage variability recorded was in good agreement with the relative bonding energy of the different the percentage variability decreasing fractions; with the increasing bonding energy. The order of abundance of five fractions determined was, Zn1= Zn2< Zn3< Zn4<< Zn5. High seasonal and spatial variability was recorded for all forms of spatial and seasonal variations zinc determined. The are depicted in Fig. 4.3.

Large percentages of zinc were found in Zn4 and Zn5 fractions: other fractions also contributed significantly to the total pool although to a less extent. The Zn1 fractions varied from 0.07 to 13.2 mg/kg, contributing 0.2 to 13.4 percent of ZnT (Table 4.1). The content of Zn1 was very low in the sediments of Segment I in the southern arm of the estuary but abnormally high values were recorded in sediments from the northern arm of the estuary, especially at Station 9 which





received the industrial effluents (Fig. 4.3). Even the sediments from Station 10, situated upstream of the industrial belt contained high content of Znl occasionally. On an average, the Station 7 which received the sewage effluents recorded the highest values of Znl. There was a sharp decrease in Zn1 content towards the marine side in Segment II. Seasonal variations were quite prominent with the maximum values recorded during the periods of low discharge (Winter and pre-The relative amount of Zn1 was found to monsoon). decrease sharply from Station 9 (effluent receiving station) indicating that the major part of zinc received in the estuary was in the labile form, and the same underwent rapid transformation down-The relatively low values recorded for Znl stream. for the sediments in Stations 8 and 9 might be primarily due to the coarse sandy nature of sediments. The seasonal variation in percentage distribution was inconsistent indicating that the anthropogenic inputs played a decisive role in determining the partitioning of zinc metal in sediments, over and above to the The decrease in Zn1 seasonal influences. in sediments of Segment II might be due to the desorption of metal under the increasing influence of calcium and magnesium with seawater mixing. Competition of major cations (Na<sup>+</sup>, Ca<sup>++</sup>, and Mg<sup>++</sup>) for surface sites on fine particles was probably ineffective under freshwater conditions, as the trace metals were bound very strongly (Dempsey and Singer, 1980); however the competition for surface sites could be more pronounced in seawater (Luoma and Davis, 1983). The premonsoonal increase in the absolute amounts of Zn1 at all stations could be attributed to the enhanced availability of exchange sites during the periods of low discharge because of the settling of finer suspended particles. The very low values on the southern part of the anthropogenic estuary established the absence of any zinc source in this part of the estuary.

Concentrations of Zn2 fractions ranged from 0.2 to 37.4 mg/kg contributing 0.4 to 19.7 percent of ZnT (Table 4.1). Higher concentrations were recorded at Stations 6 and 7 in

Segment II. Segment III had higher content of Zn2 than Segment I (Fig. 4.3), although the former had coarser sandy sediments. A unique characteristic feature observed in the seasonal distribution of Zn2 was the premonsoonal low values at majority of the Stations and abnormally high values in the winter. Verv high amounts of Zn2 were found to be associated with sediments of Stations 6 and 7 during the monsoon and postmonsoon periods. The relative abundance of this fraction of Zn was significantly higher in all the Stations of Segment II and III (exception Station 5 in Segment II). The seasonal variation pattern in the relative abundance was not distinguishable, again indicating the inconsistency caused by anthropogenic inputs.

As pointed out in the case of iron and manganese, phosphates might be playing an important role in controlling Zn2 distribution. The affinity of phosphates for zinc was well established (Burrow, 1987): an excess of phosphates in soil could cause zinc deficiency in crops (Kiekens, 1990). Thus the higher concentration of Zn2 recorded at Stations 6 and 7 during monsoon and postmonsoon periods could be attributed to eroded soils from uplands which might get trapped in these sheltered sedimentary environments (Pillai, 1989). Also, the contents of industrial effluents also might be influencing Zn2 form as was indicated by an exceptionally high percentage of this form for the winter and premonsoon seasons at Station 9.

The Zn3 fraction varied from 0.8 to 46.2 mq/kq in concentrations, and from 2.4 to 19.3 in percentagewise distribution (Table 4.1). Maximum values were recorded at Station 7, the sewage effluent receiving site in Segment III (Fiq. 4.3), with values decreasing sharply towards the seaward direction. Seasonal higher values were recorded during the winter and postmonsoon periods and lower, during monsoons. Comparatively higher values were recorded in Segment III than in Segment I. Percentage wise also, the sediments in Segments II and III (with the exception of barmouth, Station 5) had significantly higher content of Zn3 than Stations in Segment I. No seasonal

variation was observed at Station 7 in relative contribution but other Stations recorded slightly higher values during the premonsoon and winter periods.

The distribution pattern of Zn3 in sediments of northern arm of the estuary clearly demonstrated the influence of ironmanganese oxide precipitation in scavenging the metal under the influence of estuarine mixing. Thus at the low salinity regions of the estuary (Station 7) much of zinc was fluxed into the sediments in the easily reducible phases. Increase in hydroxylamine leachable zinc was also observed in suspended particles of the Fraser estuary as the freshwater changed into brackish waters (Grieve and Fletcher, 1977). The depletion of this form of zinc towards the more marine regions of the estuary might be either due to the dissolution [as pointed out by Hart and Davis, (1981) in Yarra river estuary] or dilution of sedimentary zinc with particles containing very low values of zinc as evidenced by the decreasing values for zinc downestuary. Otherwise it might be due to diagenetic changes influenced by decomposing organic matter leading to further reduction of easily reducible fractions (as also indicated by a decrease in the abundance of Mn3 and Fe3 fractions). The enhancement of Zn3 at the upper riverine Stations during the periods of low river discharge indicated a change in the location of mixing zone as also indicated by the enhanced values for Fe3 during the corresponding periods.

The Zn4 values ranged from 1.2 to 81.4 mg/kg contributing 6.7 to 43.4 percent of ZnT (Table 4.1). Higher values were recorded at Stations 6 and 7 (Fig. 4.3). Sediments of Segment III had higher levels of Zn4 than that of Segment I. The minimum was recorded during the monsoons and maximum during the winter and premonsoon periods. Percentagewise, all the Stations in Segment II and III had higher content of Zn4 than the Stations of Segment I. Seasonal trends percentagewise in distribution was not quite distinct, although the monsoonal and postmonsoonal values were higher in most of the Stations.

One distinguishing feature noted in the distribution of Zn4 fraction was the minimum variability observed among the Stations within a Segment. The percentage values also showed less variation among the Stations in the northern arm, although the values were always higher than that of the southern parts. It indicated the influence of organic sources of both detrital and autochthonus nature in influencing the Zn4 fraction which was considered to be associated mostly with organic phases, in this estuary. In terrestrial soils it had been recorded that nearly 60-75 percent of soluble zinc was present in chelated zinc form (Hodgson et al., 1966, Geering and Hodgson, 1969) and rain water draining the soil might contain appreciable amounts of zinc in this form. The high content of organic zinc in river waters would get flocculated and sedimented at the early stages of mixing, thus contributing an appreciable percentage to ZnT as evidenced by the higher percentage of this fraction during the monsoon and postmonsoon periods. However, the higher content of Zn4 recorded during winter and premonsoon had to be attributed to the autochthonus production of organic materials under the influence of enhanced productivity during these periods, and also to the coagulation and sedimentation of these products in a comparatively quiescent aquatic environment. Obviously, organic matter was an important factor affecting the behaviour of zinc in sediments. A high content of organically bound zinc was reported in the waters of this estuary (Shibu et al. 1990). Due to their colloidal nature, zinc humates could be considered as an organic storage pool of sedimented zinc.

The content of Zn5 fractions varied from 6 to 91 mg/kg contributing 28 to 89 percent of ZnT. In absolute values, the sediments of Segment II had the highest content of Zn5 (Fig. 4.3). Stations in Segment I had slightly higher content of Zn5 than Stations in Segment III. A monsoonal minimum and premonsoonal maximum were also observed at most of the Stations. Seasonal variations were not distinguishable in the percentagewise distribution, but the Stations in the northern arm of the

estuary had relatively low contribution (less than 50 percent) from this fraction than the Stations in the southern arm of the estuary, where the percentage wise distribution was above 70 percent.

The Zn5 fraction was generally considered as an inert detrital and lattice origin fraction being (Salomons and Forstner, 1980) and was usually taken as a measure of the geochemical contamination of the sediment sample. The slightly higher values of this fraction recorded for the sediments in southern part of the estuary were in agreement with the distribution of other residual metals, and therefore could be attributed to the particle size distribution of sediment particles. However, the consistent lower percentage contribution recorded for Zn5 fractions at all Stations of the northern estuary, at all seasons clearly demonstrated the effect of anthropogenic activities on metal partitioning. The distribution pattern of the various forms of zinc in Muvattupuzha (unaffected by anthropogenic inputs) and Periyar (severely contaminated by man's activities) rivers clearly demonstrated that the anthropogenic inputs were usually present in relatively loosely bound fractions as also recorded in various other locations (Tessier et al. 1980; Forstner, 1982; Calmano and Forstner, 1983; Samanidou and Fytianos, 1987; Pardo et al. 1990). This form might undergo rapid biogeochemical changes in the dynamic environments of estuaries giving rise to other forms of the metal.

percentage of zinc fractions In Table 4.4 reported in surficial sediments of different geographical locations are shown, along with the data from the present study. The trends in partitioning of zinc recorded during the present study agree well with the trend recorded from other centres. One major difference is the comparatively low values recorded for Zn3 fractions, although the ability of iron-manganese oxides to scavenge zinc from solutions is well established (Morgan and Stumm, 1964; Jenne, 1968) and this could be attributed to the

low manganese values reported from this estuary, as it has been shown (Liang et al. 1990) that the manganese oxide fraction is more important than iron oxide fractions in the chemistry of zinc.

### 4.3.4 COPPER

(1989)Flemming and Trevors reviewed the available scientific literature on copper toxicity and chemistry in the environment and concluded that the chemical form of copper was important in controlling its geochemical and biological behaviour. Baker (1990) summarised the chemistry of copper in soils and pointed out that the plant availability depended on a number of soil factors which controlled its chemical speciation. Under natural conditions, copper forms sulphides, sulphates, carbonates and other compounds. Also, this metal is associated with organic matter, oxides of iron and manganese, silicate clays and other minerals. Copper in aqueous systems received attention mostly because of its toxic effects on biota.

Interestingly, the estuarine behaviour of copper differs markedly from that of iron - iron is removed almost totally from solutions in estuaries but copper behaves the other way (Wangersky, 1986). The free ion certainly interacts with the organic matter present but the result is usually an increase in solubility, rather than sedimentation (Nelson and Lund, 1982: Windom et al. 1983). The behaviour of copper in rivers was predicted to be similar to zinc based on the multi-river associations summarized by Martin and Meybeck (1979). They suggested that 10-50 percent of the total copper would be in the dissolved phase. The strong tendency of copper to remain in solution suggests that complexing agents, for example, organic matter from the decay of vegetative matter may stabilise copper in the soluble form. Very recently Paulson et al. (1991) demonstrated that the behaviour of copper in estuaries was linked to that of organic matter mineralisation.

Table 4.4 Percentage o	f zinc f	ractions	reported	in sedim	ents from	differe	nt geogr	aphical locations.
Location	Zn1	Zn2	Zn3	2n4	Zn5	Zn6	ZnT (mg/kg)	Reference
Mobile bay, Site 1			4.15	52.04	43.79		257	Brannon et al. 1977
Site 2			2.25	58.97	38.76		177	- do -
Yamaska river-SPM	0.9	21	41	Ŋ	33		301	Tessier et al. 1980
St. Francois river-SPM	2.2	24	39	6	29		335	- do -
Rhine river	10.3		35.4	8.3	12.7	33.7	921	Forstner, 1982
Weser estuary	4		45	2	12	38	350	Calmano and Forstner, 1983
Rotterdam harbour	1		59	1	<b>3</b>	37	1130	- do -
Neckar river	e		48	2	13	34	610	- do -
Subae river-unpolluted			9.6	17.3	73.1		31	Patchineelam and Forstner, 1983
-polluted			34	6.2	17	42.4	1117	- do -
Yellow river	0.3	1	3	24	52	19	40	Hong and Forstner, 1983
Keum estuary			24.5	1.5	74		897	Lee, 1985
Krka estuary-range	9- 33	27- 43	1- 4	9 <b>-</b> 15	20- 37		32- 72	Prohic and Kniewald, 1987
Axois river	0.76	8.36	28.41	32.03	30.44		249	Samanidou and Fytianos, 1987
Axios estuary	0.99	9.08	27.08	29.12	33.71		220	- do -
Aliakmon river	1.75	3.80	13.37	52.06	29.02		78	- do -
Aliakmon estuary	1.76	4.21	13.65	50.36	30.02		81	- do -
Pennsylvania (sludge)	0.5	17		9	42	34	527	Elliot et al. 1990
Pisuerga river	1.06	13.55	33.75	41.31	10.32		246	Pardo et al. 1990
COCHIN ESTUARY-MEAN	1.75	6.43	9.37	24.28	58.22		70	PRESENT STUDY
-RANGE	0.24-13.35	0.38- 19.70	2.43- 19.32	6.71- 43.38	28.01- 88.77		14- 259	- do -

Shibu et al. (1990) observed that the copper in the fresh and estuarine waters of the Cochin estuary was extensively associated with organic colloidal matter. Fresh water contained 69 percent copper in dissolved fraction, but in estuarine water it was reduced to 21 percent. These authors observed an increase in exchangeable and total particulate copper from freshwater region to estuarine region, in the Distribution of copper in the surficial Cochin estuary. sediments exhibited wide spatial and seasonal variability (Nair et al. 1990).

Adediran and Kramer (1987) observed variations in copper adsorption on clay, iron and manganese oxides and organic fractions along a salinity gradient. Luoma (1986) from both theoretical modelling and laboratory studies showed that copper was not partitioned exclusively to a single component of estuarine sediment and that copper distribution among components might change from sediment to sediment. Luoma and (1981) also concluded similarly from statistical Bryan Thus all assessment of copper distributions. available approaches suggest that copper competitively partitions among a variety of components in oxidised estuarine sediments and that partitioning is driven by the highly variable chemical characteristics of those sediments. Preliminary studies on chemical partitioning of copper in sediments of the Cochin estuary (Nair et al. 1991) showed that partitioning pattern varied widely with the sites sampled.

Total copper values in the sediments of the Cochin estuary determined during the present study varied from 1.3 to 50.0 mg/kg with a mean value of 212.2 mg/kg (Table 3.7). A very high percentage of copper (47 to 90 percent) was found to be in the Cu5 fraction (Table 4.1); the fraction next in abundance was Cu4 (8 to 49 percent). The least abundant fraction was Cu3 (<1 percent on an average). Cu1 and Cu2 fractions were distributed more or less in equal amounts constituting about 2 percent of total copper.

The stationwise seasonal mean values of the five fractions of copper determined by chemical extractions are depicted in Fig. 4.4. The values are shown in absolute as well as percentage values. The Cul values ranged from 0.02 to 1.12 mq/kq with an average of 0.17 mq/kq (Table 4.1). Relatively higher values were recorded in sediments of Segment II, followed by Segment I (Fig. 4.4). The values at Segment III were, generally lower, but for an anomalously higher value recorded at Station 9, (industrial effluent outfall) and this might be due to some surreptitious local discharge of copper containing effluents. Seasonal variations in the distribution of Cul were not quite evident, probably due to the low amounts detected. In relative amounts, the riverine Stations on either arms of the estuary had higher percentages of Cu1 than the saline Stations of Segment II. Higher percentages were observed during the monsoon period.

The percentage distribution pattern of Cul indicated that copper in sediments was released as the sediment particles entered the saline water region from freshwater regions. Salomons and Bril (1990) demonstrated from laboratory studies, simulating changing estuarine conditions, that copper contaminated sediments from Fly river were able to release Salomons and Eagle (1990) confirmed release of copper copper. in Fly river sediments, from field studies. These reports are A simple in agreement with the results of this study. explanation that can be given for the copper release from sediments is linked to ion-exchange process. In fresh water, the trace metals like copper were more competitive in getting adsorbed to the exchange sites available on particulates, as the competition from alkaline earth metal ions was minimum. However, in the low salinity regions of upper estuary, calcium and more effectively magnesium exchanged with copper (and other divalent trace metals) thus impoverishing the particles in trace metals. Apart from desorption processes which occurred when sediments were subjected to an increase in chlorinity, the reverse process, i.e. adsorption might also occur. Adsorption



Fig. 4.4 Seasonal mean concentrations and percentages of copper fractions (concentrations in mg/kg dryweight)

processes were possible when the total number of surface sites in sediments increased as in silty clay sediments. However, even with an increase in number of surface sites, the increase in chlorinity counteracted the effect and caused desorption.

As observed for Cu1, Cu2 also was distributed in comparatively low amounts, the values ranging from 0.06 to 0.96 mg/kg with an average of 0.18 mg/kg. The highest value was recorded at Station 5 and in general the premonsoon values were higher and monsoonal values lower. As observed for Cul, anomalous high value was observed at Station 9 for Cu2 during the premonsoon period. In relative terms the three stations higher values, especially during the in Segment III had monsoon and postmonsoon periods. The percentage value of this fraction was minimum at Stations 6 and 7 and the highest was industrial recorded at the Station 10, located upstream of the belt especially during the monsoon period. This indicated the terrestrial input of this form of copper probably associated with phosphates of agricultural soils eroded during monsoon A low percentage contribution of this fraction of rains. was reported from the Axios and Aliakmon copper river (Samanidou and Fytianos, 1987) and also from the Pisuerga river (Pardo et al.1990), as observed here in the Cochin estuary.

Of all forms of copper determined, Cu3 recorded the minimum values, the values ranged from below detection limits to 0.2 mg/kg. The low values detected in most of the samples placed restrains on any discussion on its spatial and seasonal distribution and hence it was not attempted to. However the consistent low values recorded for this fraction was a feature different from what had been recorded in other estuaries. Many workers detected significant portion of copper in this phase (e.g., Tessier et al. 1980). On the other hand, Brannon et al. (1977) detected only very little copper extracted by hydroxylamine hydrochloride or dithionite reagents, in Mobile Bay sediments. Also they found no correlation between copper and manganese or iron extracted in these fractions. Moreover
they observed that dithionite could extract more of copper than the hydroxylamine hydrochloride and this they attributed to the greater association of copper with iron oxides than with manganese oxides. Data from the Rhine (Forstner 1982), the Neckar (Calmano and Forstner, 1983), the Axios and Aliakmon (Samanidou and Fytianos, 1987) and the Pisuerga (Pardo et al. 1990) rivers all recorded very low percentage of copper in Cu3 fraction. In the Cochin estuary, it is surmised that the iron reaching the estuary readily amorphous undergoes transformation to crystalline forms with poor ability for co-precipitation, under the influence of the tropical climate.

The Cu4 fraction varied from 0.16 to 15.9 mg/kg and the average detected was 2.5 mg/kg (Table 4.1). The maximum value detected (Fig. 4.4) was at Station 7, the sewage effluent receiving station, with very high premonsoonal enrichment. Very high value was detected also at Station 9, during the premonsoon period.  $d_{i}$  All the four Stations in Segment I had more or less similar values of Cu4, with significant premonsoonal enrichment. The relative value of Cu4 exhibited a sharp decrease down the estuary in Segment I and II. In Segment III the maximum percentage was recorded at Station 9, the Seasonal industrial effluent receiving station. trends were indistinct in percentage wise distributions.

(1983) effluent Feely et al. showed that the sewage receiving stations had very high amounts of trace metals in organic bound form and therefore, the high values of organic bound copper in Station 7 could directly be attributed to the city sewage discharged into this region. Tessier et al. (1980) also observed higher percentage of total copper bound to organic matter for the particulate matter of Yamaska and St. Francois rivers. This result was partly explained by the fact that for most ligands copper exhibited the highest stability constants of all the metals considered (Smith and Martell, 1976; Stumm and Morgan, 1970). Copper-organic matter association in sediments was reported by other researchers also

72

(Samanidou and Fytianos, 1987; Pardo et al. 1990). The reduction in percentage values indicated partial reduction of organic rich sediments which served to free copper into water column, thus reducing the CuT concentration in sediments. The enhancement of Cu4 values during the premonsoon periods indicated coagulation of organic matter rich in trace metals and their deposition during the more quiescent periods.

The values for Cu5 fractions ranged from 0.9 -34.5 mg/kg with an average of 9.3 mg/kg (Table 4.1). The highest seasonal mean value (Fig. 4.4) was recorded at Station 7 with values decreasing sharply on either sides. The sediments of Segment III recorded relatively lower values for this fraction of copper. Seasonal higher values were recorded during the premonsoon and winter periods. The percentage-wise distribution showed the highest value in sediments of Stations 5 and 6 in Segment II and the seasonal variation was minimum. In segment I, the values increased downstream. No spatial or seasonal pattern in percentage wise distribution could be observed in riverine stations in the north, although the percentage values were lower than that recorded in lower saline Stations.

Forstner (1982), Calmano and Forstner (1983) and Hong and Forstner (1983) observed that a significant portion of CuT is associated with the moderately reducible phases of iron oxides in sediments. In the present study, this fraction was not separately estimated, and therefore the Cu5 fraction determined included this phase along with the residuals in the lattice This explains the high percentages recorded for this minerals. fraction of copper. Gibbs (1977) also reported high percentages (74-84%) of residual copper from the particulate matter of Amazon and Yukon rivers. The decreasing trend in CuT values (Chapter 3) together with an increasing trend in percentage Cu5 values suggested that the decrease was taking place at the expense of less firmly bound and organic bound fractions, and this hypothesis was further supported by the significant reduction observed for Cu1, Cu2, Cu3 and Cu4 fractions in Segment II. The spatial distribution pattern of Cu5 followed the same pattern followed by other metals and was thus controlled by the geochemical and hydrological processes. It is known that the smectite type of clays in sediments could preferentially take up copper ions in their lattice structure, and in the marine environments these types of clays would be found in abundance. Therefore, the transformation of labile copper into less mobile Cu5 fraction by lattice substitution also could be expected; thus enhancing the percentage Cu5 values.

In Table 4.5 data reported on chemical partitioning of copper from other parts of the globe are presented, along with the data resulted from the present study. The range of values reported from the Cochin estuary is quite comparable with the global values for all the fractions of copper studied.

# 4.3.5 CADMIUM

Among the trace metals considered in this study, cadmium is of particular interest because of its high mobility in the environment and its long biological half-life in man (WHO. 1982). Pollution of the environment has been rapidly increasing in recent years as a result of rising consumption of cadmium by industry. Estimates of global emissions (Nriagu and Pacyna, 1988) relative to primary production of the metals showed that cadmium had the highest rate of emissions amongst all metals. Even before cadmium was used commercially, contamination with cadmium was reported as occurring from a wide range of materials containing cadmium as an impurity (Alloway, 1990); zinc minerals, lead minerals and phosphatic fertilizers contain cadmium as impurity.

Cadmium has no known essential biological function and as such is highly toxic to plants and animals (Craig, 1986). The concentrations normally encountered in the environment do not

Table 4.5 Percentage of	ť copper	fractions	reporte	d in sed	iments fr	om diff	erent geo	graphical locations.
Location	Cu1	Cu2	Cu3	Cu4	Cu5	Cu6	CuT (mg/kg)	Reference
Yukon-SPM	2.3		3.8	3.3	87.3		416	Gibbs, 1973
Mobile bay, Site 1			0.24	32.63	61.02	6.11	38	Brannon et al. 1977
Site 2			0.05	41.71	51.41	13.83	43	- do -
Yamaska river-SPM	1	8	20	31	41		64	Tessier et al. 1980
St. Francois river-SPM	1.2	14	12	52	22		165	- do -
Rhine river	2.7		0.2	14.5	34.2	48.4	202	Forstner, 1982
Weser estuary	1		<1	1	41	57	27	Calmano and Forstner, 1983
Rotterdam harbour	6		47	4	c,	37	1062	- do -
Neckar river	3		<b>&lt;</b> 1	6	16	72	242	- do -
Yellow river	1.3	4	9	27	24	37	6	Hong and Forstner, 1983
Keum estuary	1.2		47.7	11.9	39.2		283	Lee, 1985
Krka estuary (range)	8- 78	9- 48	2 <b>-</b> 35	<b>4-</b> 29	6- 16		10- 95	Prohic and Kniewald, 1987
Axios river	0.80	1.47	3.08	44.65	49.69		47	Samanidou and Fytianos, 1987
Axios estuary	0.98	2.19	6.01	42.98	47.84		46	- do -
Aliakmon river	1.15	2.25	4.97	70.85	20.78		43	- do -
Aliakmon estuary	0.96	2.11	4.67	73.98	18.28		37	- do -
Pennsylvania (sludge)	1	5.8		<b>b.</b> 3	55	32	50	Elliot et al. 1990
Pisuerga river	0.31	3.31	0.05	64.2	32.12		67	Pardo et al. 1990
COCHIN ESTUARY-MEAN	2.08	2.72	0.74	20.34	74.21		12	PRESENT STUDY
- RANGE	0.37- 6.67	0.32- 9.52	<0.01- 4.44	7.56- 49.19	46.57- 89.80		1.0- 50	- do -

cause acute toxicity. However, in view of the chronic accumulation of cadmium in the human body, the factors influencing its bio-magnification through the foodweb are of great importance. Therefore, strict control of the agricultural use of dredged materials, restrictions on the non-essential uses of cadmium, stringent regulation of its emissions etc. are recommended (WHO, 1982).

Studies on the behaviour of cadmium in estuaries received attention along with the studies on copper, zinc and lead. Duinker (1985) reported extensive removal of cadmium, copper and zinc in the Rhine estuary. Elbaz-Poulichet et al. (1987)reported high concentration of cadmium in the Rhone river due to intensive use of phosphatic fertilizers in the catchment. Huynh-Ngoc et al. (1988) observed that hydrographical conditions greatly influenced the partitioning of cadmium between dissolved and particulate phases. Zwolsman and van Eck (1990) who studied the Scheldt estuary observed that the content of cadmium, copper and zinc were high in fresh water, that there was a rapid decrease in the low salinity region and that the distribution of these metals was greatly controlled by biological processes. Forstner et al. (1990) reviewed metal pollution in the Elbe river and observed that cadmium enrichment was relatively small in the dissolved phase even in strongly polluted section on the river. Jouanneau et al. (1990), demonstrated with the data from the Gironde estuary, the existence of an estuarine particulate stock with a long residence time during which the contaminated particulates originating upstream released a large proportion of their Evaluation of available data cadmium in the dissolved form. from the Cochin estuary indicated that the estuary received anthropogenic inputs of cadmium (Paul and Pillai, 1983a; Nair et al. 1990; Shibu et al. 1990; Nair et al. 1991). Bioaccumulation of cadmium in the native flora and fauna also have been reported (Paul and Pillai, 1983b) although the levels recorded were quite below the permissible levels. Analysis of surficial sediments (Nair et al. 1990) revealed accumulation arising out of anthropogenic inputs especially in the northern The chemical partitioning of this element in parts. surficial sediments revealed high spatial variability (Nair et al. 1991). Shibu et al. (1990) observed gradual decrease of dissolved cadmium as the river water entered the estuarine regions. About 15-25 percent of cadmium in water was found to be associated with organic compounds. It was also observed that the increased suspended solids concentration at the barmouth region did not contribute to the progressive increase of. cadmium associated with particulates and it was explained that the increasing salinity inhibited the association of cadmium into the particulate phase.

least abundant of Cadmium was the the trace metals determined in the sediments during the present study. The CdT values ranged from 0.34 to 8.44 mg/kg (Table 3.7) with а mean value of 1.66 mg/kg. The order of abundance in the distribution of various fractions of cadmium was Cd5> Cd4>> Cd3> Cd2> Cd1. The stationwise seasonal mean values for the various fractions of cadmium are depicted in Fig. 4.5.

The values of Cd1 ranged from 0.01 to 0.24 mg/kg contributing 1.4 to 18.9 percent of CdT (Table 4.1). A11 the three Stations in Segment II had comparatively higher content of Cd1 values showing pronounced enhancement during the winter and premonsoon periods. Sediments of Station 9, which received the industrial effluents, also recorded higher Cd1 values during the winter and premonsoon periods. All the four Stations in Segment I (southern parts) had more or less similar content of Cd1. The percentagewise distribution pattern exhibited, an entirely different trend. The three Stations in Segment II had the lowest percentage contribution from this fraction of cadmium, and the percentage values showed an increasing trend towards the riverine end members. Moreover, relatively higher percentages of this fraction were noted during the monsoon in most of the Stations. Station 9 was an exceptional case in the sense that Cdl was much higher in this Station both in

76



Fig. 4.5 Seasonal mean concentrations and percentages of cadmium fractions(concentrations in mg/kg dryweight)

absolute and relative terms and this was due to the reactive nature of cadmium being received by the estuary through industrial effluents. Elsokkary and Muller (1990) observed that in polluted sediments of the Nile river, a significant portion of cadmium existed in the exchangeable form whereas in unpolluted sediments, the values were much lower. The higher values recorded at the Stations in Segment II might be due to the sewage discharges and the retention of cadmium by the increased number of surface sites that might take part in exchange reactions. However, the marked decrease in the percentage values towards the seaward direction clearly demonstrated that a significant portion of Cd1 had undergone transformation to other forms under the influence of increasing salinity. Cadmium removal from exchange sites could be either due to exchange with calcium and magnesium or due to complexation of cadmium by the increasing chloride concentrations as had been observed in laboratory studies (Bourg, 1983). High concentration of Cd present in this phase will have adverse impact on water quality during dredging and disposal operations.

The Cd2 values ranged from 0.01 to 0.26 mg/kg contributing 0.7 to 22.6 percent of CdT (Table 4.1). One unique feature noted in its spatial distribution (Fig. 4.5) was the relatively The values higher values in Segment III than in Segment I. showed gradual increase downstream in Segment III and I. A11 Stations in Segment II had more or less similar concentrations. Seasonally, the highest values were observed during the premonsoon and winter as the saline waters invaded the farther regions of the estuary. In percentagewise distribution a distinct increase towards the downward direction was noted in Segments I and III, but a sharp decline was noted in Segment II. In the southern part, highest percentages were noted during the monsoon. Significant interactions in seasonal distribution between the Stations also were noted.

Both riverine inputs and estuarine processes had control over the distribution of Cd2 fractions. This fraction of cadmium was expected to be associated with carbonates and phosphates in the sediments. The higher percentages recorded during the monsoon could be attributed to terrestrial sources associated with the phosphates. probably Most of the phosphatic fertilizers contain cadmium as an impurity and thus, this association could be explained. The significantly higher content of Cd observed in the northern part of the estuary was due to industrial identified as discharges and sewage The increasing trends recorded downstream spatially effluents. and towards premonsoon season seasonally indicated that the gradually increasing salinities influenced precipitation of this form of cadmium. However, under higher salinity conditions, this fraction appeared to be unstable and underwent rapid remobilization as indicated by the sharp decline in The distribution pattern of Cd1 percentage values. and Cd2 fractions revealed that an appreciable percentage of sediment cadmium would be solubilized and become readily mobile following an increase in salinity and/or a slight lowering of That, substantial percentages of particulate cadmium would ρH. be readily desorbed from suspended sediments in contact with seawater was shown by Van der Weijden et al. (1977), for the Rhine river sediments.

The Cd3 fractions exhibited wider variability (C.V.= 112.3 percent) than the other two fractions (C.V.= 72.8 percent and71.1 percent) described above. Values ranged from 0.02 to 0.86 mg/kg, contributing 1.7 to 26.1 percent of CdT (Table 4.1). Higher values were recorded at Segment II (Fig. 4.5). Sediments Segment I and III had more or less similar values. of Pronounced premonsoonal hike was recorded at Station 7. Premonsoonal hike was recorded at Stations 2, 3 and 4 of Segment I also. Significant seasonal variation was not observed at the barmouth (Station 5). In percentagewise distribution a gradual decreasing trend was observed from the riverine end members towards the marine regions, the minimum percentage contribution being recorded at Station 6 of Segment II. Percentage values were higher during the monsoon season in many of the stations.

In comparison to the values reported by Forstner (1982),(1983), Calmano and Forstner Hong and Forstner (1983),Samanidou and Fytianos (1987) and Pardo et al. (1990), very low percentage values had been recorded for Cd3 fraction in the Cochin estuary. However, the trends in spatial distribution paralleled that of Mn3 and Fe3 fractions determined in the estuarine sediments. Therefore, the lower percentages recorded for Cd3 fractions could be due to the lower abundance of Mn 3 and Fe3 fractions in the sediments.

An unique feature in the partitioning of cadmium fractions was the preponderance of Cd4 fraction over the other three fractions discussed. Of all the metals studied in the estuary, cadmium showed the highest relative abundance in this phase. The Cd4 values ranged from 0.07 to 3.83 mg/kg contributing 15 to 60 percent of CdT. The highest value was recorded in Segment II. Very high enrichment was observed during the premonsoon season (Fig. 4.5). Contents of Cd4 in other Stations were more or less equal, with slightly higher values during the premonsoon period. Percentagewise also, the sediments of Segment II had higher values and the values exhibited premonsoonal enrichment. However, at the four Stations in Segment I, the percentage values were higher during the postmonsoonal period, indicating strong interaction between sites and seasons in the distribution of Cd4 forms.

The distribution pattern of Cd4 fraction in the Cochin estuary was quite different from what had been reported from other estuaries by Tessier et al. (1980) and Calmano and Forstner (1983), that, a higher portion of CdT was found as Cd4 fraction in the Cochin estuary. The percentage distribution pattern suggested that this form of cadmium had undergone minimum transformation during the estuarine mixing processes (as was evidenced by minimum variation between the Stations during a particular season). However, seasonal variability in the biological processes had an effective control on the distribution of this fraction of cadmium as indicated bv enhanced values for the premonsoon period. Corroborative observations to the influence of biological activity in controlling the estuarine behaviour of cadmium were reported from other estuaries, as well (Noriki et al. 1985; Kuwabara et. al. 1989; Zwolsman and van Eck, 1990; Slauenwhite and Wangersky, 1991). During the winter and premonsoon periods, cadmium desorbed from the Cd1, Cd2 and Cd3 fractions appeared to be counteracted by their subsequent removal in the lower estuary which was most likely caused by phytoplankton bloom occurring in this region. Direct uptake of cadmium might be involved similar to the findings in the Delaware estuary (Sharp et al. 1984). The phytoplankton bloom might also affect the behaviour of cadmium by causing a sudden pH increase in the lower estuary, which strongly favoured adsorption of these metals by suspended matter (Bourg, 1983). Increased content of Cd2, especially during the winter and pre-monsoon seasons was noted during this study. The behaviour of Cd4 is much similar to that of Zn4 in Segment II, but contrasted that of Cu4. This could be explained on the basis of stability of organic complexes of these metals. Organic complexation of dissolved cadmium and zinc was less significant because the stability constants of their respective humic acid- metal complexes are four orders of magnitude lower (Turner et al. 1981). On the other hand, the strong association between copper and humics suppressed not only prevented uptake by phytoplankton but also adsorption onto suspended matter at high pH values.

The Cd5 content varied from 0.1 to 4.6 mg/kg, contributing 23 to 69 percent of CdT (Table 4.1). As observed for this fraction of other metals, higher values were recorded at Station 6 and the values decreased sharply on either side (Fig. 4.5). Cd5 less, equitably distributed fraction was more or in the sediments of other Segments. Generally, the premonsoon values were higher and monsoonal values lower. The highest relative

contribution from this fraction was recorded at Stations 6 and 7 during the monsoon period. At other Stations, the seasonal trend was varying. The effluent receiving station had the lowest relative contribution from this fraction.

The lowest percentages of Cd5 in the sediments of Stations 8 and 9 pointed out anthropogenic input of this metal in more mobile forms. This was confirmed by the increasing percentages of Cd1 and Cd2 forms recorded in these Stations. The comparatively lower percentages of this fraction of cadmium (Cd5) indicated that cadmium was highly reactive in this estuary, and equally higher percentages in the Cd4 fraction suggested that the phytoplankton the in estuary were essentially involved in the biogeochemical cycling of this toxic trace metal.

Table 4.6 presents the percentages of cadmium fractions reported from surficial sediments of different geographical locations, along with data recorded during current investiga-Except for the values of Cd3 and Cd5 fractions, tion. the data recorded agree well with the range reported from other centres. The lower values noted for Cd3 been have already explained on the basis of relative low abundance of freshly precipitated manganese and iron oxides in this tropical environment. Higher percentage of Cd5 fractions indicated that the magnitude of anthropogenic inputs over the natural inputs was comparatively low. However, the sequential chemical extraction technique employed was efficient enough to bring out the reactive nature of anthropogenic inputs. The sedimentary environments in Segment II housed large amounts of Cd4 and Cd5 fractions which would get released into the estuarine water, should the environmental conditions change. The contribution of Cd4 fraction was highest during the winter and premonsoon periods signifying the high biological productivity during this season. These studies revealed that relatively warmer an important sink for cadmium in sediments of the Cochin estuary

Table 4.6 Percentage o	f cadmiun	n fractio	ns report	ed in se	diments	from dif	ferent ge	ographical locations.
Location	Cd1	Cd2	Cd3	Cd4	Cd5	Cd6	CdT (mg/kg)	Reference
Los Angels harbour	1	5	6	38	14	34	2.2	Gupta and Chen, 1975
Yamaska river-SPM	26	49	12	<1	12		59.8	Tessier et al. 1980
St. Francois river-SPM	28	31	13	<1	27		4	- op -
Rhine river	26.7		42.3	16.8	12.3	1.9	6	Forstner, 1982
Weser estuary	18		47	6	2	24	2.6	Calmano and Forstner, 1983
Rotterdam harbour	16		60	9	2	16	7.9	- do -
Neckar river	13	64	12	1	10		33.6	- do -
Subae river-unpolluted			82.6	10.8	6.6		0.1	Patchineelam and Forstner, 1983
-polluted			78	18.2	1.4	2.4	27.6	- do -
Yellow river	12.9	22	3	2	18	43	0.1	Hong and Forstner, 1983
Axios river	10.79	24.89	29.60	19.05	15.67	1	17.6	Samanidou and Fytianos, 1987
Axios estuary	9.59	35.53	22.16	10.77	22.13		13.5	- do -
Aliakmon river	1.22	13.54	42.82	24.10	18.32		2.0	- do -
Aliakmon estuary	1.03	15.25	41.43	24.80	17.49		1.7	- do -
Pennsylvania (sludge)	5.8	19		38	38		1.6	Elliot et al. 1990
Pisuerga river	3.85	13.56	46.76	23.77	13.14		1.1	Pardo et al. 1990
Nile river - range	1.8- 43.8	7.3- 50.4	9.1- 23.6	6.2- 22.7	1.4- 9.1	4.8- 36.4	0.1- 1.1	Elsokkary and Muller, 1990
COCHIN ESTUARY-MEAN	5.76	7.48	8.13	34.90	43.73		1.7	PRESENT STUDY
– RANGE	1.40- 18.89	0.68- 22.58	1.72 - 26.09	15.22- 59.53	23.33- 69.35		0.3- 8.4	- do -

was the non-residual fraction, especially the one associated with organic matter.

#### 4.3.6 LEAD

Lead is an environmental contaminant in all habitats, except those remote areas far away from populous regions or where human settlement is very recent. It enters the environment from several sources, the main inputs being the lead alkyls used in petroleum byproducts. Lead has a relatively long residence time in comparison with most other pollutants. As a result, lead and its compounds tend to get accumulated in soils and sediments. There are fears that body burdens below those at which clinical symptoms of lead toxicity would appear may cause mental impairment in children. Plants take up this metal from contaminated soils but there are no significant evidences as to the essentiality of this metal in biota (Prosi, 1989; Davies, 1990).

There are only few studies reported on lead from aquatic systems, mainly because of the associated analytical problems (Elbaz- Poulichet et al. 1984; Windom and Smith, 1985). The available reports deal mainly with the distribution of total lead in dissolved and suspended phases which have importance in identifying areas of contamination and also in elucidating the estuarine behaviour of the metal. Available evidences suggest that in estuaries, lead is readily removed from solutions andbound to particulates. remains

Considerable seasonal and spatial variation in the distribution of lead in sediments of the Cochin estuary was reported by Nair et al. (1990). Sewage discharges were considered to be an important anthropogenic source. Higher values were recorded during the premonsoon and postmonsoon periods. Lead content was reported to be lower during monsoon probably due to bedload movement downstream as a result of higher river discharge and the sedimentation of fresh riverine

particles. Nair et al. (1991) reported significant spatial variation in the distribution of different forms of lead determined chemically in the sediments of the estuary. The investigations reported herein, were aimed at improving the database.

lead content of the estuarine sediments varied The total from 3.5 to 49.9 mg/kg (Table 3.7). These values were quite low in comparison with values reported from other estuaries of India (Table 3.11), but the values closely agree with the values reported earlier from the Cochin estuary (Nair et al. 1990; Nair et al. 1991). The abundance of various forms of lead ranked in the order of Pb1 < Pb2< Pb3< Pb4 <<<Pb5. In the distribution behaviour, different forms of lead were found to behave similar to the corresponding form of copper except in details.

mean values of various fractions The seasonal of lead determined in sediments sampled from various sites, together with their percentage values are depicted in Fig. 4.6. Values of Pb1 ranged from 0.05 to 1.13 mg/kg contributing 0.6 to 7.6 percent of PbT (Table 4.1). Highest value in absolute terms was recorded at Station 6 during all the seasons, and the values decreased on either side. Stations in Segment I had slightly higher values than Stations in Segment III. The values recorded at Station 9 exhibited significant seasonal variations. Premonsoonal values were generally high and monsoonal values were low. Higher percentagewise distribution was noted in the Stations in Segment I, as also noted for the sediments at Station 9, in Segment III. The percentage contribution of Pb1 to PbT was low in the sediments of Segment II. Seasonal variations in percentagewise distribution did not show any clear cut trends.

The enhanced values in Segment II were ascribed to the silty clay nature of sediments which had high adsorption sites to



Fig. 4.6 Seasonal mean concentrations and percentages of lead fractions (concentrations in mg/kg dryweight)

hold lead ions amidst the competition from calcium and magnesium ions in high saline waters. However, the reduction in percentage values clearly demonstrated desorption of lead under these conditions. As noted for copper, lead also was more soluble in riverine waters than the estuarine waters, as indicated by a gradual decrease in the percentage contribution of Pb1 downstream. The seasonally higher values recorded at Station 9 was due to the industrial effluents. The enhancement in absolute values during pre-monsoon was resulted from the coagulation and settling of fine particles which had more exchange sites.

Pb2 values ranged from 0.04 to 2.13 The mg/kg and contributed 0.9 to 15.5 percent of PbT (Table 4.1). The highest value was recorded at Station 9, followed by the barmouth (Station 5) during the premonsoon period (Fig.4.6). Premonsoon values were higher at all Stations and the monsoonal values were minimum . In percentage-wise distribution, sediments of Station 9 had the highest contribution from this fraction of lead at all seasons. Percentage values decreased sharply towards the marine side on both arms of the estuary indicating that this form of lead was undergoing transformation under the changing salinity conditions, resulting in comparatively low percentage of Pb2 in sediments of Segment II. Seasonal influence in the percentage-wise distribution was not distinguishable.

The high Pb2 values both in absolute and relative terms recorded at Station 9 could be attributed to the influence of industrial effluents. The higher value recorded at Station 5 reflected the influence of boat and ship traffic at the barmouth Station. Precipitation of lead carbonates or occlusion of lead into hydrogenous carbonate minerals under saline influence was indicated by the pronounced enhancement in seasons. Pb2 fraction during winter and premonsoon However, reduction in the percentage values towards the marine side indicated remobilization of Pb2 under the influence of

increased salinities, either due to competition from calcium and magnesium ions for exchange sites or due to formation of chloride complexes of lead.

The content of Pb3 in sediments ranged from 0.05 to 4.93 mg/kg contributing 1.0 to 20.9 percentage of PbT. During the monsoon and postmonsoon seasons, Pb3 was found to be distributed more or less equally at all Stations, irrespective of the texture of the sediments. However, higher values were recorded during the premonsoon and winter periods, with peak values appearing at Station 5 and Station 9 as observed for Pb2 fractions. Percentage-wise distributions exhibited a declining trend towards the marine side on either arms of the estuary with very high percentages recorded at Station 9 during the premonsoon and at Station 8 during the monsoon.

This fraction of lead was considered to be associated with the freshly precipitated oxides and hydroxides of iron and manganese. These oxides would remain as coating on both fine and coarse particles and hence could get more or less equally distributed throughout the estuary during the monsoon and postmonsoon periods. However, as the season proceeded from postmonsoon to premonsoon through winter, the rate of formation of authigenic hydroxide precipitates changed under the influence of varying salinities (as observed for Fe3 and Mn 3 fractions), and this change was reflected in the distribution of Pb3 also during these seasons. The anomalous values recorded at Station 9 both in absolute and relative terms indicated contamination through industrial effluents.

Pb4 content in sediments ranged from 0.26 to 3.2 mg/kg contributing 1.4 to 28.6 percent of PbT. During the monsoon, postmonsoon and winter seasons, the values exhibited little variations in absolute amounts between sites, but seasonal variations were significant. The values increased from the monsoonal low values gradually to winteral high values. In Segments I and III, the premonsoonal deposition of this fraction was highest among the seasons, but in Segment II, the values showed a decrease during premonsoon. Higher contribution from this fraction of lead was indicated in the riverine sediments of southern parts especially during the winter and pre-monsoon periods. The sediments in Segment III also had elevated percentages than the saline Stations in Segment II.

The distribution pattern of the relative amounts of Pb4 followed closely that of Cu4, except that Cu4 values were much higher than that of Pb4 values. The high percentage values in the riverine end members suggested proportionally large river input of this fraction, which would undergo transformation to more stable forms towards the lower parts of the estuary. Large scale remobilization of Pb1, Pb2, Pb3 and Pb4 fractions was indicated by the sharp decrease noted in the percentage values of these fractions in the lower parts of the estuary.

Pb5 was the most abundant of the lead fractions determined in sediments, and the values ranged from 2.3 to 45.5 mg/kg, contributing 52.5 to 92.7 percent of PbT (Table 4.1). Stations II recorded higher values, 5, 6 and 7 in Segment both in absolute and relative terms (Fig. 4.6). Sediments of Segments I and III had comparatively low values, but the premonsoonal enrichment was quite significant at these Stations. No trends could be distinguished in seasonal variations of the percentage values.

An unique feature noted was the increased percentages of Pb5 at the Stations of Segment II, whereas, a reduction was noted in the PbT values (Fig. 3.4). This suggested that the other forms of lead recorded in the sediments were quite unstable and were susceptible to transformation at the high saline conditions of the lower estuary. It was demonstrated in nearshore environments 1981 (Murty and Veerayya, and Satyanarayana et al. 1985) that the metals like copper and lead were associated with the smallest size particles the of

These particles flocculated only sediment. particles. when subjected to rapid environmental changes as experienced in an estuarine situation. In Segment II more and more finer particles which were relatively rich in lead got flocculated and sedimented. The lower percentage contribution of Pb5 in Station 9 indicated anthropogenic input of lead in less tightly bound fractions which got converted rapidly to other forms when the sediments were transported downstream under conditions of bedload movement.

Table 4.7 compares the data resulted from the present study on chemical partitioning of lead with that reported from other parts of the globe. The range of values recorded from this study is in unison with values reported from other centres. The high percentages of Pb3 reported from other centres were different from the values reported here, and as explained for copper, might be due to the lower abundance of freshly precipitated iron and manganese oxides. Compared to zinc and copper a higher proportion of lead was found to be associated with residual fraction indicating low geochemical mobility and negligible biological availability of lead in this tropical estuary.

# 4.3.7 COBALT

Cobalt is an essential micronutrient for plants and animals. Its requirement especially by ruminants and N-fixing organisms has stimulated a great deal of research into the forms, quantities and chemical behaviour of this element in soils (Smith, 1990). Although the presence of cobalamine, a cobalt containing vitamin was detected in seawater, published work on the behaviour of cobalt in aquatic systems, especially under the estuarine systems, is scanty. In oceanographic considerations, behaviour of cobalt is usually related to that of manganese (Murray, 1975b; Amdurer et al. 1983; Li et al. 1984) in the sense that both these metals `recycle' within the system. Though they get removed from solution phase during estuarine mixing their removal mechanisms differ since

Table 4.7 Percentage	of lead l	ractions	reported	in sedim	ents from	differe	ant geogr	aphical locations.
Location	Pb1	Pb2	Pb3	Pb4	Pb 5	Pb6	PbT	Reference
Yamaska river-SPM	bdl	18	46	19	17		173	Tessier et al. 1980
St. Francois river-SPM	lbdl	18	49	20	13		111	- do -
Rhine river	1.5		0.6	5.2	54.1	38.6	118	Forstner, 1982
Weser estuary	2		2	2	53	41	69	Calmano and Forstner, 1983
Rotterdam harbour	く1		66	7	15	18	256	- do -
Neckar river	く1		2	28	48	22	150	- do -
Subae river-unpolluted					100		25	Patchineelam and Forstner, 1983
-polluted			17.8	30.1	8.1	44	1775	- do -
Yellow river	0.3	16	4	く1	30	51	7	Hong and Forstner, 1983
Keum estuary			31.4		68.6		58	Lee, 1985
Krka estuary-range	9 <b>-</b> 58	11- 66	10- 49	bd1- 4	2- 21		23- 67	Prohic and Kniewald, 1987
Axios river	0.96	15.54	25.50	29.80	28.20		217	Samanidou and Fytianos, 1987
Axios estuary	1.16	13.26	23.82	33.13	28.63		154	- do -
Aliakmon river	0.21	24.68	16.09	35.40	23.62		95	- do -
Aliakmon estuary	0.23	24.65	19.72	29.55	25.85		156	- do -
Pennsylvania (sludge)	4.20	2.60		13.0	72.0	8.4	204	Elliot et al. 1990
Pisuerga river	1.70	15.53	33.67	12.81	36.29		19	Pardo et al. 1990
Nile river – range	bdl- 2.1	bdl- 9.6	bdl- 0.6	26.7- 70	7.6- 36.7	12.8- 33.3	15- 184	Elsokkary and Muller, 1990
COCHIN ESTUARY-MEAN	2.86	4.32	5.68	10.37	76.78		15	PRESENT STUDY
-RANGE	0.60- 7.55	0.94-15.47	1.04 - 20.94	1.39- 28.63	52.47- 92.70		3.5- 49.9	- do -

manganese and cobalt are far less `particle reactive'. Only very small percentages of cobalt and manganese were associated with particles in the water column; equally smaller percentages occurred in colloidal fractions (Amdurer et al. 1983). Environmental pollution problems are relatively insignificant for this trace metal.

Murty and Veerayya (1981) observed that in the Cochin estuary there was a decrease in manganese and cobalt content in the estuarine sediments in comparison with river sediments of similar texture and attributed this behaviour to the desorption of these elements from the river-borne detritus on coming in contact with saline waters and/ or to the mixing with marine sediments containing low concentrations of manganese and cobalt. Venugopal et al. (1982) while working on sediments from the northern parts of the estuary observed that the cobalt content of sediments was relatively low in monsoon periods than in other seasons. Nair et al. (1990) also reported higher rate of incorporation of this metal into sediments during postmonsoon and premonssoon seasons.

from the present investigations confirmed Results the relatively low levels of cobalt reported by earlier workers in sediments of the Cochin estuary. The range of values for total cobalt was 1.3 to 24.99 mg/kg with a mean of 9.2 mg/kg (Table 3.7). The abundance of different forms determined was in the order  $Co5 \rightarrow Co4 \rightarrow Co3 \rightarrow Co2 = Co1$  (Table 4.1). The seasonal mean values of the different fractions of cobalt determined along with their relative abundant values are plotted in Fig. 4.7.

The Col values ranged from 0.02 to 0.72 mg/kg contributing 0.6 to 5.7 percent of CoT (Table 4.1). A unique feature noted in the distribution of Col fraction was the gradual increase downstream in Segments I and II both in absolute and relative values (Fig. 4.7). The downstream increase in percentage values was quite pronounced in these two Segments, but in



Fig. 4.7 Seasonal mean concentrations and percentages of cobalt fractions (concentrations in mg/kg dryweight)

Segment III, no spatial trend could be detected. In absolute terms the values were higher during the premonsoon at Stations 3, 4 and 8, but monsoonal peak values were recorded at Stations 1 and 2 thus indicating strong interaction between sites and seasons in the distribution of Col. For the Stations in the southern part, the highest relative contribution from this fraction of cobalt was recorded during the monsoon but in other stations, seasonal trends were indistinct.

The increase in Col values in the downward direction of the estuary, both in absolute and relative values implied that the distribution of Col fraction was controlled by processes which could enhance the solubility of cobalt minerals even under saline conditions. It was inferred that the oxyhydoxide and carbonate phases of cobalt in sediments might undergo solubilization by reduction in the interstitial water just below the sediment surface releasing manganese and cobalt. Manganese might undergo oxidation when coming in contact with oxidised surface water, but cobalt would remain in solution, enhancing the quantity of ions that might take part in exchange reactions. However, the reduction in the percentage values of Col in Segment II hinted that other forms of cobalt were being added to these sediments in an accumulating environment thus reducing the significance of Col fractions. A parallel to this distribution pattern was seen in Fel distribution. It is interesting to note that barmouth Station did not show any decrease in Col values in comparison with other Stations in Segment II, but, most other metals in this phase had recorded depletion.

The Co2 values ranged from 0.02 to 0.57 mg/kg contributing 0.4 to 10.0 percent of CoT. The values showed sharp decrease down the estuary in Segment I and increase in Segment III, with no clear cut spatial trends in Segment II. Comparatively higher values were recorded at Stations 1 and 2 in spite of the sandy nature of the sediments. In percentage-wise distribution, the spatial trends were similar, but, the Stations in Segment II recorded the lower percentages. Seasonally, the values were higher during the premonsoon period in all of the Stations of Segment I both in absolute and relative terms; but in Segment III relative values were higher during the postmonsoon period.

Downstream reduction in Segment I signified the dissolution of Co2 probably at subsurface layers in the sediments where organic matter decomposition would create reducing conditions. Premonsoonal enhancement in absolute values indicated the importance of increasing biological production and changing pH values that might influence cobalt precipitation in this phase, most probably, related to authigenic carbonates. Reduction in the percentage values down the estuary was paralleled by a gradual increase in percentage values of Co1 fractions, thus indicating that they were related events.

The Co3 values ranged from 0.03 to 0.86 mg/kg contributing 0.5 to 13.6 percent of CoT. Higher values were recorded at Stations of Segment II, and a noted feature was that the premonsoonal values were lower for this form of cobalt in Segment II. Stations of Segment I had comparatively higher values than those of Segment III. In Segment III, premonsoonal value was the highest; but in Segment I the seasonal pattern was indistinct. Percentage-wise distributions exhibited more or less equal values for all the Stations but seasonal differences were significant. Sediments had the highest contribution from this fraction of cobalt during the monsoon period.

The spatial distribution pattern of Co3 was similar to that of Fe3 and Mn3 fractions. Smith (1990) reported that there was a general tendency for total cobalt to increase with manganese, particularly with the easily reducible fraction, in the acidic grassland soils of England. In aquatic systems manganese and iron are probably removed from water column by oxyhydroxide precipitation and cobalt by co-precipitation. Their behaviour in the sediment is very sensitive to redox conditions (Amdurer et al. 1983). When mixed down into suboxic sediments (depending on the concentration of organic material in the sediments, temperature, bioturbation etc., this may be just below the sediment surface) manganese oxide is reduced releasing manganese and cobalt to the porewater. Since the kinetics of cobalt oxidation is relatively slow, and the pore water exchange in these sandy sediments is quite rapid, the cobalt released may get recycled rather than being `trapped' in the oxidised sediment layer as pointed out by McCaffrey et al. (1981) for Narragansett bay sediments.

The values of Co4 fractions ranged from 0.3 to 6.30 mg/kg contributing 10.4 to 30.5 percent of CoT. The highest value was recorded at Station 6 and the values decreased on either Sediments of Segment I had comparatively lower side. values than that of Segment III. Seasonal changes were characterized by monsoonal low and premonsoonal peak values. In percentagewise distribution, the sediments of Segment I had relatively higher contribution from this fraction of cobalt than other Segments. Seasonal variations were not significant, although premonsoonal values were slightly higher.

The spatial distribution pattern of Co4 resembled that of Mn4 again indicating a similarity in the chemistry of these two elements in this estuarine environment. The enhanced values in the sediments of Segment II might be due to the high organic matter content in these sediments. The enrichment observed during premonsoon could be ascribed to the incorporation of and cobalt into the authigenic organic materials their subsequent deposition. The minimum variability in the percentage values suggested that the sediments attained equilibrium between the precipitation certain and remobilization of this fraction of cobalt. Though Co and Mn were similar in their oceanographic behaviour, Santchi (1988)has shown from microcosm experiments that in enclosures which received added nutrients, cycling of cobalt was completely decoupled from manganese, i.e., cobalt was removed more slowly than manganese. The differences noted in partitioning

behaviour of cobalt and manganese could ,thus, be attributed to the difference in their chemical and biological behaviour.

The content of Co5 fractions varied from 0.8 to 19.0 mg/kg contributing 56 to 84 percentage of CoT. Its distribution pattern paralleled that of CoT (Fig. 4.1) suggesting that a definite portion of CoT was always found in this fractions. This observation was further supported by its relative distribution values which exhibited minimum variation within segment. The seasonal variations in percentagewise anv distribution values were also insignificant. The absolute values of Co5 exhibited the same seasonal distribution trend, as exhibited by CoT, thus indicating that the cobalt loading in the sediments was controlled by natural processes and influence of anthropogenic inputs was minimum. Gilkes and McKenzie (1988) have shown that cobalt could replace manganese from manganese minerals under oxidizing conditions and under ideal conditions cobalt adsorbed on manganese hydroxides might become strongly bound to crystal lattices. The data on Co5 indicated that lattice substitution on manganese minerals could be an important process controlling the phase transformations.

Reports on chemical partitioning of cobalt in sediments 4.8. are scanty. Available data is furnished in Table The ranges reported from the present study agrees well with the values reported by Gibbs (1977) and Tessier et al. (1980)except for the Co3 fractions and that of Pardo et al. (1990),except for that of Co4 fractions. The lower abundance of freshly precipitated iron and manganese in the sediments of the Cochin estuary have already been stressed to explain the above feature.

# 4.3.8 NICKEL

Nickel and nickel compounds are used in a variety of industries such as stainless steel, alloys, electronic compounds, petroleum products and electroplating. Nickel is

Table 4.8 Percentage of	cobalt fract	ions report	ed in sedir	ments from	different g	eographica	l locations.
Location	Co1	Co2	Co3	Co4	Co5	CoT (mg/kg)	Reference
Yukon-SPM	4.7		29.2	12.9	51.4	40	Gibbs, 1973
Yamaska river-SPM	bdl	lbdl	31	12	51		Tessier et al. 1980
St. Francois river-SPM	bdl	bdl	47	13	35		- do -
Keum estuary	3.7		18.6	6.6	71.1	44	Lee, 1985
Krka estuary - range	2- 18	14- 23	20- 32	21- 34	13- 26	3- 10	Prohic and Kniewald, 1987
Pisuerga river	4.67	6.12	8.62	66.87	13.76	11	Pardo et al. 1990
COCHIN ESTUARY-MEAN	2.40	2.35	4.09	19.81	71.35	6	PRESENT STUDY
-RANGE	0.6- 5.7	0.4- 10	0.5- 13.6	<b>10.4-</b> 30.5	55.8- 83.5	<b>1.3-</b> 24.0	- do -
		pq	ll = below No data ave	detection 1: ailable on (	imits Co6		

also used as a catalyst for hydrogenation of oils and also in domestic detergents. Thus the element reaches the environment from a number of point and non-point sources (Nriagu, 1980). Atmospheric inputs of nickel are also important (Nriagu and Pacyna, 1988) in many regions. Because of its widespread occurrence and known toxic effects at higher levels of contamination, several investigations on the environmental behaviour of this elemment have been taken up (McGrath and Smith, 1990). Anderson and Christensen (1983) observed that important factor determining the distribution of the most nickel between solid and solution phases was pH. The mobility of nickel in soils increased as the pH and cation exchange capacity decreased. Other factors which were reported to he influencing the mobility of nickel in soils were, the type and amount of clay present, amount of organic matter and presence of iron and manganese oxides. Lake et al. (1984) noted that there was an overall shift in the solid phase metal forms in sludge amended soils such that the residual forms decreased while that extracted with milder reagents increased. Waterbodies generally receive nickel from sewage discharges, oil spills and atmospheric outfalls (Nriagu and Pacyna, 1988). Though estuaries were likely to receive considerable supply of this metal much work has not been reported from estuaries (Campbell et al. 1988; Sadiq, 1989).

Murty and Veerayya (1981) reported total nickel values in the range of 10-60 mg/kg in sediments from the southern parts of the estuary. Venugopal et al. (1982) observed a downstream decrease in sediment nickel content on the northern parts of the estuary. Paul and Pillai (1983) reported monsoonal high values, in the sediments collected from Periyar river, nearer to the industrial belt. However, Nair et al. (1990) observed that the nickel distribution pattern was characterized by а preonsoonal mid-estuarine build-up following irregular postmonsoonal deposition. The total nickel (NiT) values determined in sediments during the present study varied from 1.9 to 70.6 mg/kg with an average value of 21 mg/kg (Table 3.7). Chemical partitioning (Table 4.1) revealed that 71 to 92 percent of the total nickel was present in the Ni5 fraction which was considered to be environmentally inert. The abundance of other forms of nickel was in the order Ni2< Ni1< Ni3 << Ni4. The seasonal and spatial distribution pattern of the various forms of nickel are depicted in Fig. 4.8. The data revealed significant spatial and seasonal variations. Significant interaction between the seasons and stations were also recorded.

The Nil values ranged from 0.04 to 0.89 mg/kg, contributing 0.3 to 3.6 percent of NiT. Highest value for Nil was recorded at Station 6 (except for the monsoon season) and the values decreased on either side. Comparatively higher values were recorded in the sediments in Segment I than that of Segment III. Seasonally higher values were recorded during premonsoon and winter. Percentagewise distribution indicated higher contribution of this fraction of nickel in the sediments of upper riverine regions of the estuary, whereas the percentages were very low in the saline regions. No seasonal trend was evident from the data available.

The very low percentages of Nil hinted that nickel was one of the least tenaciously sorbed of the transition metals so far discussed. On goethite surface, the sequence of adsorption was Zn > Cd > Ni (Breummer et al. 1988). The higher percentages of Nil recorded in riverine Stations indicated the higher solubility and mobility of nickel in the slightly acidic to neutral riverine waters in comparison to the more saline and alkaline waters of the lower estuary.

Ni2 values ranged from 0.02 to 0.66 mg/kg and contributed 0.2 to 5.1 percent of NiT. Comparatively higher values were recorded in Segment I and the values decreased sharply downstream. Seasonal variations were indistinct. The percentagewise distribution also exhibited down estuarine decrease in both the arms of the estuary at all seasons.





The distribution pattern suggested the influence of land drainage on the supply of this form of nickel to the estuary, most probably associated with soil phosphates. Transformation of this fraction into other forms was also suggested by the sharp decrease in down estuarine values both in absolute and relative terms.

The Ni3 values ranged from 0.02 to 1.12 mg/kg contributing 0.4 to 5.0 percent of NiT. Stations 5 and 6 exhibited higher values during all seasons. Values in Segment Ι were significantly higher than the values recorded at Segment III. Premonsoonal values were generally higher. The percentage values were slightly higher in the riverine end-members. The seasonal variations were irregular in percentage wise distribution. The very low percentages of nickel associated with this phase indicated the low mobility of nickel in these environments.

The Ni4 values ranged from 0.20 to 11.56 mg/kg contributing 3 to 22 percent of NiT (Table 4.1). The highest values were recorded at Station 6, and the values decreased sharply on either sides except for the premonsoon season (Fig 4.8). Slightly higher values were recorded at Segment I than at Segment III. In percentage values also, Station 6 recorded the highest value. Sediments of Segment III had the least contribution from this fraction of nickel. In Segment I, postmonsoonal values were higher for this fraction. The seasonal and spatial distribution trends of Ni4 resembled that of Mn4 signifying a similar behaviour for these two fractions.

Ni5 values ranged from 1.52 to 58.91 mg/kg contributing 71 to 92 percent of NiT. The distribution pattern followed exactly the distribution of NiT and depended mainly on the granulometry of sediments. Thus the silty clay sediments of the lower estuary (Segment II) had higher content of Ni5; and coarse sandy sediments of Segment III, lower values. Generally premonsoonal values were higher. Percentagewise distribution did not exhibit much variation except in Stations 3 and 9. The seasonal low percentages recorded for the postmonsoon period at these two Stations indicated occasional inputs of anthropogenic nickel, but there was no evidence of its spread to other sites.

The available data on chemical partitioning of nickel in sediments from other centres are furnished in Table 4.9 along with the data from the present study. Data from the Cochin estuary compared well with data from other centres. The tendency of nickel to remain in less mobile fractions was clearly evident from the data. The high percentage of nickel in Ni5 fraction, which was considered to be the geochemical contaminant suggested that input rates of anthropogenic nickel to this tropical estuary was much low.

#### 4.3.9 CHROMIUM

Nowadays, environmental importance of chromium, its sources and hazards are receiving increased attention (Thessan et al. 1988). Chromium is found in varying concentrations in air, water, soil and essentially in all biological tissues. Chromium enrichment results from a number of anthropogenic sources including cement production kilns, smelting furnaces, tanning and pigment production etc. (McGrath and Smith, 1990). The coal, oil and wood, and the burning of incineration of municipal refuse and sewage sludge also contribute to atmospheric chromium concentrations. The presence of industries like tanneries and plating works help to trace the elevated levels of chromium in sewage sludges and water. Its role as an essential element in plant and animal nutrition has been well documented (Anderson, 1981); its essentiality in humans also is established. However, at elevated concentrations chromium is toxic. Chromium (III) compounds have no established toxicity but in the hexavalent state toxicity through skin ulcerations and nasal perforations as well as lung cancer have been reported (Bennet, 1982).

Table 4.9 Percentage of	f nickel	fractions	reported	in sedi	ments fr	om diffe	rent geog	raphical locations.
Location	Ni1	Ni2	Ni3	Ni4	Ni5	Ni6	NiT (mg/kg)	Reference
Yukon-SPM	3.1		47.8	16	31.0		136	Gibbs, 1973
Mobile bay	7		2	7	06		127	Brannon et al. 1976
Ashtabula	1		2	10	65	23	185	- do -
Bridge port	1		2	26	69	£	182	- do -
Yamaska river-SPM	1.7	3.9	14	4.4	77			Tessier et al. 1980
St. Francois river-SPM	0.8	3.8	24	6.2	66			- do -
Keum estuary	0.9		18.9		80.2		40	Lee, 1985
Krka estuary - range	<b>4-</b> 31	11- 42	5 <b>-</b> 21	l3- 31	13- 30		40- 72	Prohic and Kniewald, 1987
Pennsylvania (sludge)	0.6	12		4.5	51	31.8	44	Elliot et al. 1990
Pisuerga river	4.34	16.34	14.06	53.66	12.31		47	Pardo et al. 1990
Nile river – range	0.2- 3.7	1.8- 7.8	1.0- 3.3	34.5- 41.7	43.2- 49.4	8.3- 14.9	74- 98	Elsokkary and Muller, 1990
COCHIN ESTUARY-MEAN	1.45	1.40	1.98	10.94	84.23		21	PRESENT STUDY
-RANGE	0.34- 3.57	0.19- 5.08	0.42- 4.99	3.05 - 21.85	71.39- 92.04		2- 73	- do -

Chromium exists in a number of oxidation states but the most stable and common forms are chromium(III) and chromium(VI). Chromium(VI) is an anion and is considered to be more toxic. Chromium(III) is much less mobile and adsorbs to particulates more strongly (McGrath and Smith 1990). The solubility of chromium(III) decreases above pH 4.0 and at about pH 5.5 complete precipitation occurs.

Most of the environmental studies reported on chromium were related to soil application of sewage sludges. In majority of soils the relatively insoluble and less mobile chromium forms predominated (Cary et al. 1977; Grove and Ellis, 1980). Cary et al. (1977) observed that addition of chromium either as chromium (III) or chromium (V1) to soils reverted to forms which were unavailable to plants. In sewage amended soils very stable organic complexes accounted for the extremely low availability of the element (Lake et al. 1989). However complexing chromium with soluble organic acids (for example, citric acid, fulvic acids and soil extracts of water-soluble organic matter) maintained chromium (III) in solution and was, therefore, a means of enhancing its mobility.

Reports on chromium content of sediments of the Cochin estuary are very few. Recently Nair et al. (1990) reported a chromium content of 3 to 118 mg/kg in the surficial sediments of the estuary. They observed that the southern part of the estuary was more contaminated with chromium than the northern part; seasonally, monsoonal values were low with gradual build up during postmonsoonal period, followed by a premonsoonal redistribution of sediment metals.

The total content of chromium determined in surficial sediments during the present study varied from 4 to 100 mg/kg, with an average value of 30 mg/kg (Table 3.7) and the values were in agreement with the earlier report (Nair et al. 1990). The summary statistics showing the maximum, minimum, mean and

standard deviation in the values recorded for different fractions of chromium are presented in Table 4.1 along with percentage values. The abundance of the various forms of chromium was in the order, Cr5>>> Cr4>> Cr1> Cr2> Cr3. The seasonal mean values (both in absolute and relative terms) and spatial variations are depicted in Fig. 4.9.

The values for Cr1 fraction ranged from 0.02 to 0.49 mg/kg, and this fraction contributed 0.08 to 5.19 percent of the CrT. Higher values were recorded at Station 6, and values declined on either sides. Seasonal variations were characterized by higher values during premonsoon at most of the Stations. The percentagewise distribution did not show significant variations either in space or in time except at Station 9, the industrial effluent receiving Station. The very low percentage values (usually <1 percent) recorded is in agreement with the nonmobility generally assigned to this element in neutral to alkaline environments.

The values for Cr2 fraction ranged from 0.02 to 0.27 ma/ka, contributing 0.08 to 3.28 percent of CrT. Values were significantly higher in the southern part of the estuary, especially during the premonsoon period. The northern part exhibited low values in absolute terms, but the percentage values were higher in northern parts, especially in Segment III. Generally , the contribution of Cr2 fraction to CrT was quite insignificant evidencing the lack of affinity of chromium to carbonates and phosphates.

The Cr3 form was the least abundant of the chromium forms determined; the values ranged from below detection levels to 0.16 mg/kg contributing less than 0.4 percent of CrT. Τn absolute terms, higher values were detected at Station 5, the barmouth, with values decreasing slowly towards the riverine side.In the lower and middle estuarine regions, higher percentages were recorded during the monsoon and postmonsoon




T 551.468.6 (213.5):669.018:541.123.4 - G1.5020-

seasons. The extremely low abundance of this form also signified the non-reactivity of chromium in the environment.

The Cr4 fraction varied from 0.12 to 22.57 mg/kg in absolute values, and from 2.24 to 27.63 percent in relative values. Maximum values were recorded at Station 6. Sediments of Segment I and III recorded very low values. Premonsoonal values were highest and monsoonal values lowest. Percentage wise also, Station 6 and 7 of Segment II recorded higher values. Significant seasonal variation was exhibited in Segment III, with higher premonsoonal and winteral values. Seasonal trends in percentage distribution were not detectable in Segments I and II.

The ability of chromium to form stable organic complexes has been reported from sewage amended sites (Lake et al. 1989). Α significant portion of chromium existed in organically bound fractions in the sediments at Stations 6 and 7 and this evidenced the influences of effluents sewage in the transformation of chromium fractions. A high chromium content was reported also from organic rich regions of Guanabara, tropical estuarine system of South America (De Luka Rebello et al. 1986). It is probable that under such situations chromium (III) is the predominant species and the same get adsorbed fast on colloidal particles which may eventually be deposited under estuarine influences. This fast removal of chromium to sediments in an estuarine system rich in organic matter is in disagreement with the observations of Sholkovitz and Copland (1981) which assumed that complexation of heavy metals would render them unavailable for sediment adsorption.

Cr5 was the most abundant form of chromium determined in the sediments of the Cochin estuary. The values of Cr5 ranged from 3.73 to 81.14 mg/kg; this form contributed 71 to 96 percent of CrT. The spatial distribution pattern was related to the granulometry of the sediments, the silty clay sediments in Segment II having the highest Cr5 value. The seasonal variation also was significant and was related to the depositional processes in the estuary; the highest values being recorded when the river flow was minimum during the premonsoon and winter periods. Lower values were recorded during monsoons. In percentagewise distribution a highly significant spatial pattern was exhibited. While the sediments in Segment Ι and III maintained very high proportions in the Cr5 fractions, the sewage effluent receiving Station and nearby Station in Segment II recorded relatively lower contribution from this fraction signifying the importance of other forms of chromium. The high values of chromium recorded at these stations together with the high percentages of organic bound fractions (Cr4) clearly demonstrated that the sewage effluents were an important source of chromium to this estuary.

Table 4.10 compares the data recorded during the present study with the data reported in estuarine sediments from other centres. The trends exhibited in distribution of chromium into various forms are similar for many cases. Most of the workers had observed the predominance of Cr5 (residual) fractions. Elsokkary and Muller (1990) attributed the higher percentages recorded in Cr4 fraction to anthropogenic reasons; in the Cochin estuary also, the higher percentages in Cr4 fraction could be traced to the sewage effluents.

Table 4.10 Percentage	of chrom	lium frac	tions rep	orted in	sediment:	s from	different	geographical locations.
Location	Cr1	Cr2	Cr3	Cr4	Cr5	Cr6	CrT (mg/kg)	Reference
Yukon-SPM	2.3		7.2	13.2	64.5		115	Gibbs, 1973
Rhine river	0.1		0.4	9	7.1	86.3	338	Forstner, 1982
Yellow river	0.1	1	1	19	57	22	32	Hong and Forstner, 1983
Krka estuary – range	11- 39	12- 36	3- 14	2- 20	29- 38		30- 105	Prohic and Kniewald, 1987
Axios river	0.52	0.52	8.52	32.51	57.93		193	Samanidou and Fytianos, 1987
Axios estuary	0.19	0.63	9.69	30.70	58.79		167	- do -
Aliakmon river	0.23	0.69	3.54	66.90	28.64		212	- do -
Aliakmon estuary	0.22	0.65	2.87	69.88	26.38		179	- do -
Pennsylvania (sludge)	1	2.4		10	49	38	171	Elliot et al. 1990
Nile river – range	<0.1- 0.1	0.2- 1.9	bdl- 0.8	29.9- 53.2	12.2- 62.5	8.3- 31.9	174- 940	Elsokkary and Muller, 1990
COCHIN ESTUARY-MEAN	0.95	0.64	0.16	11.01	87.25		30	PRESENT STUDY
-RANGE	0.08- 5.19	0.08- 3.28	0.03- 0.46	2.24- 27.63	71.01- 95.56		<b>4-</b> 105	- do -

# CHAPTER 5 CHEMICAL PARTITIONING - IMPLICATIONS IN A TROPICAL ESTUARY

7

- 5.1 INTRODUCTION
- 5.2 TRACE METAL PHASES IN SEDIMENTS
  - 5.2.1 EXCHANGEABLE PHASE
  - 5.2.2 CARBONATE PHASE
  - 5.2.3 EASILY REDUCIBLE PHASE
  - 5.2.4 ORGANIC/SULFIDE PHASE
  - 5.2.5 RESIDUAL PHASE
- 5.3 PHASE CONCENTRATION FACTOR
- 5.4 ESTUARINE ENRICHMENT
- 5.5 THE OUTLOOK

## 5.1 INTRODUCTION

The spatial and seasonal distribution pattern of the five chemical fractions of each of the trace metal was discussed in the previous Chapter, in relation to sedimentological and geochemical processes that are active in a typical tropical estuary. It was established that the estuarine processes had the potential to significantly alter the chemical partitioning of trace metals in the sediments. Since, in environmental studies, a better understanding of the processes andinteractions involved is expected, when a group of constituents are considered, an attempt is made herein to rationalise the behaviour of trace metals in the estuarine environment with respect to a few parameters that reflect the geochemical/ oceanographical/ biological behaviour of the metals studied. The various accumulative phases have been characterised separately and their interactive processes compared and contrasted with variations in salinity.

## 5.2 TRACE METAL PHASES IN SEDIMENTS

The study quantified the trace metal phases in surficial sediments so as to assess the short term transformations that might affect the fate and transport of trace metals in toto and also to throw light on baneful ecological consequences. The average concentration of the different fractions of the nine metals studied and their average percentage distribution are presented in Figs. 5.1a and 5.1b. The observed order of abundance of the trace metals, was

Fe >>> Zn > Mn > Cr > Ni > Pb > Cu > Co > Cd,

which was the same as that reported in surface soils (Martin and Whitfield, 1983) except for the position of Zn, which is identified as a major contaminant in the northern parts of the estuary. A high variability was observed, in both



Fig. 5.1 Chemical partitioning of trace metals in sediments of the Cochin estuary. (a) concentration (mg/kg) (b) percentage

the absolute and relative amounts of various fractions of the metals. This, however, was not unexpected, for, the chemistry of metals differed greatly; so too their extractability from sediments by various extractants. Their chemistry which determines the extractability of the metals with various reagents used and consequently decide their fate in this estuary have been discussed at length.

## 5.2.1 EXCHANGEABLE PHASE

Ammonium acetate used in the sequential extraction scheme has been the most extensively used extractant to liberate the exchangeable metals from soils and sediments (Tessier et al. 1979). Ammonium acetate is strongly buffered around the neutral point and is thus admirably suited for the simultaneous determination of all exchangeable cations, but has no effect on silicates, sulfides or organic matter and therefore the metals extracted could be regarded as the best estimate of exchangeable trace metals present in sediments.

A number of sediment forming materials with large surface area such as clay minerals, amorphous iron and manganese oxides as well as organic substances are capable of adsorbing trace metals either by cation exchange or as а result of intermolecular attraction (Forstner, 1979c). In sediments these substrates for ion adsorption occur in the form of associations and the properties of such aggregates are quite different from that of the constituent particles. Thus it is impossible to establish a general order of affinities, applicable to individual heavy metals, based solely on ionic charge and ionic radii. Under estuarine conditions, the changing salinities and pH variations further complicate the problem (Bourg 1983; Bilinski et al. 1991).

The stationwise annual mean data for the exchangeable fraction of metals, is presented in Table 5.1 and the mean values for the percentages are furnished in Table 5.2. The

വ	.1	Annual mean in sediments	concentr of the C	ation (mg Cochin est	g/kg) of tuary.	the exch	langeab le	fraction	of trace	metals
		Fe1	Mn1	Zn1	Cu1	Cd1	Pb1	Co1	Ni1	Cr1
ean SD	ł	6.69 5.56	17.46 12.14	$0.15 \\ 0.02$	$0.12 \\ 0.05$	0.04 0.01	0.30 0.08	0.15 0.07	0.20 · 0.04	$0.13 \\ 0.05$
ean SD		4.01 1.79	8.08 4.41	0.10 0.03	$0.10 \\ 0.02$	0.04 0.01	0.35 0.15	0.19 0.05	0.19 0.08	$0.12 \\ 0.07$
ean SD		5.57 2.26	3.34 1.71	$0.12 \\ 0.02$	$0.17 \\ 0.05$	0.05 0.02	0.39 0.17	0.26 0.19	0.27 0.08	$0.20 \\ 0.07$
ean SD		5.85 4.39	3.29 1.46	$0.17 \\ 0.03$	$0.17 \\ 0.05$	0.04 0.01	0.36 0.13	$0.21 \\ 0.04$	0.27 0.06	$0.14 \\ 0.04$
ean SD		5.33 1.63	3.85 1.82	$0.82 \\ 0.86$	$0.19 \\ 0.06$	$0.11 \\ 0.06$	0.47 0.25	0.30 0.07	$0.34 \\ 0.10$	0.35 0.06
ean SD		6.52 1.42	3.91 1.38	$1.94 \\ 0.95$	$0.21 \\ 0.04$	$0.10 \\ 0.02$	0.83 0.21	0.31 0.08	0.50 0.23	0.41 0.05
ean SD		7.85 1.70	3.78 3.01	2.59 0.55	$0.23 \\ 0.07$	$0.11 \\ 0.04$	$0.41 \\ 0.25$	$0.17 \\ 0.07$	0.24 0.11	$0.27 \\ 0.06$
ean SD		2.45 1.51	1.40 1.38	1.28 0.32	$0.11 \\ 0.05$	0.04 0.01	0.16 0.07	$0.10 \\ 0.02$	$0.13 \\ 0.02$	$0.14 \\ 0.07$
ean SD		0.98 0.55	$1.06 \\ 0.73$	3.82 4.34	$0.27 \\ 0.33$	0.10	$0.30 \\ 0.24$	$0.04 \\ 0.02$	$0.11 \\ 0.04$	$0.16 \\ 0.05$
ean SD		1.40 0.95	$1.72 \\ 0.36$	0.92 0.50	$0.10 \\ 0.02$	0.03 0.01	$0.11 \\ 0.05$	0.08 0.03	$0.10 \\ 0.03$	$0.10 \\ 0.03$
Mear SD	<b>i</b> .	1 4.66 3.46	$4.79 \\ 6.34$	1.19 1.88	$0.17 \\ 0.13$	$0.07 \\ 0.05$	$0.37 \\ 0.26$	$0.18 \\ 0.12$	0.23 0.15	$0.20 \\ 0.11$

Table	5.2	Annual mean sediments of	percenta the Coch	ages of in estuar	the .y.	əxchangeab l	e fracti	olo	trace me	tals in
Station		Fel	Mn1	Zn1	Cu1	Cd1	Pb1	Co1	Nil	Cr1
1	Mean SD	0.08 0.07	15.96 7.89	0.60 0.21	2.31 1.15	7.11 2.04	3.93 0.97	1.85 0.64	$1.24 \\ 0.43$	0.64 0.25
2	Mean SD	0.05 0.02	8.96 3.73	0.38 0.12	$1.48 \\ 0.43$	5.90 1.83	4.25 1.80	2.03 0.57	.1.02 0.37	0.48 0.28
£	Mean SD	0.09 0.03	8.03 1.53	0.59 0.18	2.75 1.21	7.30 1.96	4.35 1.32	3.69 1.04	$1.84 \\ 0.46$	$0.94 \\ 0.22$
4	Mean SD	0.08 0.05	5.70 1.84	$0.64 \\ 0.16$	$2.43 \\ 0.85$	6.40 1.99	4.06 1.40	3.68 0.85	1.84 0.37	0.65 0.20
5	Mean SD	$0.04 \\ 0.01$	3.46 1.89	$0.81 \\ 0.64$	$1.11 \\ 0.45$	4.73 1.47	1.85 0.44	2.04 0.64	0.80 0.26	0.59 0.18
9	Mean SD	$0.04 \\ 0.01$	3.18 1.23	0.97 0.43	$0.72 \\ 0.18$	1.85 0.32	2.35 0.67	$1.57 \\ 0.50$	$0.91 \\ 0.33$	0.58
7	Mean SD	0.10 0.03	5.01 2.60	1.39 0.31	$0.97 \\ 0.45$	3.10 1.35	1.58 0.50	$1.07 \\ 0.28$	0.91 0.28	0.63 0.11
8	Mean SD	0.05 0.03	7.55 4.56	2.95 0.84	2.67 1.16	6.01 2.14	2.02 0.64	3.59 0.66	1.85 0.49	$1.15 \\ 0.32$
6	Mean SD	$0.02 \\ 0.01$	6.90 2.28	6.16 3.66	3.82 1.34	10.54 4.02	$2.91 \\ 1.72$	$1.45 \\ 0.97$	$2.37 \\ 0.61$	2.33 1.27
10	Mean SD	0.02 0.02	$\begin{array}{c} 10.46\\ 2.84\end{array}$	3.03 1.56	2.55 1.22	4.61 1.33	$1.26 \\ 0.41$	3.02 0.80	$1.70 \\ 0.67$	$1.49 \\ 0.51$
Genera	ll Mean SD	0.06 0.04	7.52 5.05	1.75 2.18	2.08 1.32	5.76 3.07	2.86 1.59	2.40 1.20	1.45 0.68	0.95

order of abundance of the exchangeable fraction of the metal, in absolute quantities was,

 $\texttt{Mn} \ensuremath{\:>} \texttt{Fe} \ensuremath{\:>} \texttt{Zn} \ensuremath{\:>} \texttt{Pb} \ensuremath{\:>} \texttt{Ni} \ensuremath{\:>} \texttt{Co} \ensuremath{\:>} \texttt{Cu} \ensuremath{\:>} \texttt{Cd}$ 

and the order roughly followed the abundance of total metals in the estuary, except for iron. However, when the percentage contributions were considered, the order of abundance was,

Mn > Cd > Pb > Co > Cu > Zn > Ni > Cr > Fe.

Van der Weijden et al. (1977) established an order Cd > Zn > Mn > Ni > Co > Cu > Cr > Fe, Pb

for desorption of metals from suspended sediments of the Rhine river and the order observed for the sediments of the Cochin estuary agreed with the same except for Mn, Pb, Ni, and Zn. While the change in the order of Ni and Zn was not of much significance in the light of the low percentages of these metals found in this fraction and the wide dispersal of the data, the behaviour of Mn and Pb was quite significant. Higher percentages of exchangeable manganese was reported also from the sediments of the Mobile Bay (Brannon et al. 1977), Yamaska and St. Francois rivers (Tessier et al. 1980), Rhine river (Forstner, 1982), Axios river (Samanidou and Fytianos, 1987) and Krka river estuary (Prohic and Kniewald, 1987) - all pointing to a peculiar reactivity of manganese. Mitchell (1964) had established the following empirical affinity sequence of heavy metal towards clay minerals,

Pb > Ni > Cu > Zn .

Soong (1974), who studied the competitive sorption of heavy metals onto clay minerals established that lead was more competitive than copper and zinc. Forstner (1979c) quoted the following affinity series for cation exchange on organic and inorganic substrates,

Pb > Cu > Ni > Co > Zn.

Elliott et al. (1990) also reported the high affinity of Pb for soils. adsorbing sites in Recently Bilinski et al. (1991)reported high affinity of Pb to estuarine particles. All these studies clearly demonstrated that lead had a high affinity for clay minerals and organic substances. While its high affinity to clay minerals could be attributed to its ionic radii (Soong, 1974), its affinity to organic substances might be due to the high stability of lead-organic complexes (Irving and Williams, 1948). Thus the higher ranking obtained for exchangeable lead in the sediments of the Cochin estuary was justifiable in the light of the known adsorption behaviour of the element. Exchangeability of cations depends on their ionic radii, electronegativity, ionic potential etc. When the relative abundance of the trace metals in the exchangeable phase was compared with these parameters (all data from Krauskopf, 1967), it was found that the affinity series established for the exchangeable fraction of trace metals in the sediments of the Cochin estuary corresponded closely with the electronegativity (Fig. 5.2a), ionic potential (Fig. 5.2b), ionic radius (Fig. 5.3 ) and the negative logarithm of the solubility product of the respective metal hydroxide. Thus the data obtained from the ammonium acetate extractant was chemically explainable and the fraction extracted did have a separate chemical identity and meaning.

the metals studied, the percentage extractable For all fraction showed decreasing trends from the riverine regions to near marine regions, on both arms of the estuary. This observation was in agreement with the model presented by Bourg (1983) as well as with the experimental results of Duinker (1983) on particulates of the Rhine estuary. In both the studies, a decrease in the percentage of adsorbed fraction was indicated with an increase in salinity. It might be due to the competitive replacement of trace metals by the alkaline earth metals (calcium and magnesium) which are abundant in saline Bourg (1983) demonstrated that although the increased waters. pH in saline waters ought to have enhanced the adsorbed metals,



Fig. 5.2 Variations in percentage exchangeable fraction as related to (a) electronegativity and (b) ionic potential



Fig. 5.3 Variations in percentage exchangeable fraction as related to (a) ionic radii and (b) negative logarithm of the solubility product of the corresponding metal hydroxide.

the effect of competing cations might have a stronger influence in the mixing zone of estuaries. In the Cochin estuary, the absolute values for exchangeable fractions of metals in sediments exhibited a down estuarine increase because of the fine grained nature of the sediments, but the percentage values for exchangeable trace metals were found to decrease. When the percentage values for exchangeable trace metals were compared with the amount of calcium and magnesium released by the ammonium acetate reagent, significant correlations were obtained for all the metals studied (Fig. 5.4), indicating the influence of calcium and magnesium (and thereby of salinity) in controlling the trace metal accumulation in this phase.

## 5.2.2. CARBONATE PHASE

Several workers (Chester and Hughes, 1967; Gupta and Chen, 1975) have shown that significant trace metal concentrations could be associated with sediment carbonates. This fraction would be highly susceptible to changes in pH. Sodium acetate, buffered at pH 5.0, was generally used for selectively extracting the carbonate fraction from the sediments (Table.1). With this reagent, dissolution of the sediment carbonate fraction was essentially complete (Tessier et al. 1979) and the attack on silicates and sulfides minerals was a minimum. The organic fraction remained unaffected. However, the specifically adsorbed trace metals, unaffected by the previous extractant (ammonium acetate, pH = 7.0) could also be extracted at this stage.

A clear cut spatial variation, related to the salinity gradient, as discussed in the case of exchangeable trace metals was not indicated in the case of the carbonate bound fractions (Tables 5.3 and 5.4). For Mn, Cd, Pb, Cu, Zn and Fe, the percentage of metals released with this reagent was higher than that released by ammonium acetate; the percentages of Cd, Cu, Zn, Pb and Cr liberated were found to be significantly higher in Segments III than in Segment I.





	vi2 Cr2	.28 0.12 .06 0.07	.29 0.08 .12 0.03	.28 0.13 .09 0.06	.26 0.11 .07 0.06	31 0.21 13 0.05	33 0.11 .16 0.03	.15 0.12 .07 0.05	.08 0.10 .03 0.03	.06 0.11 .05 0.03	.13 0.08 .08 0.03	.22 0.12 .14 0.06	
Co2		0.31 0 0.09 0	0.20 · 0 0.05 0	0.12 0 0.11 0	0.17 0 0.10 0	0.22 0 0.10 0	0.16 0 0.03 0	0.23 0 0.12 0	0.09 0.00	0.07 0 0.05 0	0.06 0	0.16 0 0.11 0	
	Pb2	0.41 0.08	0.32 0.14	0.56 0.49	0.37 0.21	$1.03 \\ 0.52$	0.80 0.25	0.45 0.09	0.33 0.23	$0.92 \\ 0.61$	0.32 0.12	0.55 0.42	
	Cd2	0.03 0.01	$0.04 \\ 0.02$	0.03 0.01	$0.07 \\ 0.02$	$0.11 \\ 0.04$	0.13 0.08	$0.12 \\ 0.04$	$0.11 \\ 0.04$	$0.12 \\ 0.08$	0.04 0.02	0.08 0.06	
	Cu2	0.12 0.03	$0.13 \\ 0.05$	$0.13 \\ 0.03$	$0.17 \\ 0.08$	$0.27 \\ 0.09$	$0.15 \\ 0.03$	$\begin{array}{c} 0.17 \\ 0.07 \end{array}$	$0.22 \\ 0.06$	$0.32 \\ 0.32$	$\begin{array}{c} 0.16 \\ 0.04 \end{array}$	0.18 0.13	
	Zn2	0.35 0.08	0.35 0.07	$0.50 \\ 0.24$	$0.63 \\ 0.22$	2.61 2.51	23.52 5.27	26.94 5.41	4.87 3.49	6.36 5.35	$2.08 \\ 0.42$	6.82 9.97	
	Mn2	15.38 6.22	13.69 4.13	5.31 3.22	<b>4.</b> 67 2.26	15.04 5.21	22.29 8.00	7.03 2.84	0.90 0.53	$\begin{array}{c} 0.94 \\ 0.88 \end{array}$	<b>1.84</b> 0.72	8.71 8.16	
	Fe2	1.92 1.43	4.36 1.96	4.92 4.43	5.86 6.19	$2.13 \\ 0.97$	32.47 14.80	9.32 6.10	1.93 1.05	2.23 0.83	$0.88 \\ 0.40$	6.60 10.60	
	u	Mean SD	Mean SD	Mean SD	Mean SD	Mean SD	Mean SD	Mean SD	Mean SD	Mean SD	Mean SD	al Mean SD	
	Static	7	7	ς	4	ນ	9	7	8	6	10	Gener	

able	5.4	Annual mean sediments of	percenta the Coch	ages of Iin estuar	the carl y.	oonate-bou	und frac	fon of	trace me	tals in
tation		Fe2	Mn2	Zn2	Cu2	Cd2	Pb2	Co2	Ni2	Cr2
	Mean SD	0.03 0.02	15.18 1.95	1.45 0.49	2.24 0.64	6.29 1.66	5.65 1.56	<b>4.1</b> 5 2.03	$1.77 \\ 0.75$	0.60 0.32
	Mean SD	0.05 0.03	15.97 2.46	$1.36 \\ 0.29$	1.86 0.67	6.72 2.48	3.72 1.16	2.12 0.35	. 1.52 0.58	$0.32 \\ 0.13$
	Mean SD	0.07 0.05	11.90 3.19	$2.11 \\ 0.37$	$2.18 \\ 0.80$	5.61 3.34	5.29 1.75	$1.82 \\ 0.97$	$1.93 \\ 0.70$	$0.58 \\ 0.14$
	Mean SD	0.08 0.08	7.96 2.60	$2.31 \\ 0.84$	2.33 0.90	11.01 5.78	3.83 1.34	3.11 1.61	1.79 0.43	$0.48 \\ 0.22$
	Mean SD	0.01 0.01	$12.74 \\ 3.55$	2.61 1.98	1.58 0.58	5.06 2.21	<b>4.11</b> 1.00	1.38 0.32	0.76 0.33	0.36 0.15
	Mean SD	0.20 0.08	16.41 $2.63$	11.81 1.98	0.51 0.11	2.22 1.00	2.23 0.56	$0.77 \\ 0.11$	$0.60 \\ 0.22$	$0.16 \\ 0.06$
	Mean SD	0.12 0.07	10.04 2.13	14.32 2.28	0.68 0.30	3.76 1.94	1.95 0.69	1.50 0.73	$0.60 \\ 0.25$	$0.27 \\ 0.08$
	Mean SD	$0.04 \\ 0.02$	5.41 1.34	9.96 4.99	5.53 1.28	14.34 3.91	3.44 1.63	4.08 3.22	$1.14 \\ 0.55$	$0.91 \\ 0.30$
	Mean SD	$0.05 \\ 0.02$	5.46 2.55	10.83 $4.29$	$5.70 \\ 1.93$	12.97 2.96	9.48 3.29	2.43 1.82	$1.77 \\ 1.50$	$1.41 \\ 0.50$
0	Mean SD	0.02 0.01	10.50 2.11	7.51 2.05	4.58 2.63	<b>6.8</b> 0 2.16	3.46 0.79	2.17 1.05	2.14 1.33	1.29 0.77
eneral	l Mean SD	0.07 0.07	11.16 4.59	6.43 5.35	2.72 2.17	7.48 4.84	<b>4</b> .32 2.57	2.35 1.86	$1.40 \\ 0.96$	$0.64 \\ 0.53$

The precipitation of metal carbonates occurs when the corresponding solubility product is exceeded. The solubility data obtained in pure individual system in distilled water represent only a guide to the conditions existing in natural systems, where the solubility product is influenced by the interplay of a variety of forces. In spite of these limitations, a regression line constructed with the percentage carbonate fraction recorded for the sediments of the Cochin estuary against the negative logarithm of the solubility product of the respective metal carbonates (Fig. 5.5) displayed а significant relationship. The comparatively higher percentages of Zn, Cd, Pb and Cu recorded at the northern riverine Stations could be attributed to the industrial effluents. It was quite probable that phosphate contained in the industrial effluents from the fertilizer factory could contribute to the higher content of this fraction. In order to bring out the influence of phosphates in controlling the metals in the M2 phase, the metal concentration recorded in this phase was compared with the phophorus released from the sediments with the same reagent (sodium acetate, pH=5.0). Significant relationships were recorded for the metals, zinc, nickel, manganese and iron (Fig. 5.6). Church (1986) observed a simultaneous removal of iron and phosphates by flocculation at low salinity regions in Delaware bay. From laboratory simulation studies, Bale and Morris (1981) also demonstrated the affinity of iron for phosphates. Windom et al. (1991)observed that many processes involving flocculation, specific adsorption and precipitation of iron and phosphates were possible and in an estuarine system it was likely that several of these occurred simultaneously. These studies indicated that phosphate could be an important accumulative phase for trace metals in estuarine systems and phosphates as well ลร carbonates get extracted by sodium acetate. The order of relative abundance of this fraction of metals in the southern parts of the estuary was,



Fig. 5.5 Variations in percentage carbonate- bound fraction as related to the negative logarithm of the solubility product of the corresponding metal carbonates.





 $\label{eq:mn} Mn \ > \ Cd \ > \ Pb \ > \ Co \ > \ Cu \ > \ Zn \ > \ Ni \ > \ Cr \ > \ Fe$  and in the northern part of the estuary was,

Cd > Zn > Mn > Pb > Cu > Co > Ni > Cr > Fe

The shift in the order of abundance signified the anthropogenic input of Cd and Zn to the northern part of the estuary. These results demonstrated that co-precipitation with carbonates would be an important elimination mechanism for metals such as Zn and Cd, particularly when hydrous iron oxides and organic substances were less abundant, especially in slightly higher salinity regions of an estuary. The relatively higher abundance of manganese, zinc and cadmium in this phase was similar to that reported from several other estuaries (Tessier et al. 1980; Prohic and Kniewald, 1987).

## 5.2.3 EASILY REDUCIBLE PHASE

The hydrous oxides of iron and manganese constitute significant sinks of heavy metals in aquatic systems, particularly under oxidizing conditions. The processes involved are those of adsorption and co-precipitation (Adidiran and Kramer, 1987). Under reducing conditions the sorbed metals are readily mobilized and this phenomenon is the basis of estimation of trace metals in the easily reducible phase. Leaching of sediments with a reducing agent such as hydroxylamine hydrochloride in presence of an acid, which keeps the liberated ions in solution, removes trace metals associated with iron and manganese oxides. Brannon et al. (1977) noted that while the extractant dissolved approximately 85 percent of the manganese oxide, only 5 percent of the iron present was dissolved. Tessier et al. (1979) and Tipping et al. (1985) also observed that manganese oxides were more easily leached than iron oxides. The reagent had minimal effect on sulfides, silicates and organic matter (Tessier et al. 1979).

It has already been pointed out (Chapter 4) that the amount of iron and manganese recovered from this phase was comparatively low in the sediments of the Cochin estuary. While the low levels of manganese in this phase could be attributed to the generally low content of manganese in sediments, the low levels of iron was attributed to the low inputs of soluble iron into the estuary because of the highly oxidizing conditions and turbulence. The low levels of trace metals detected in this phase as compared with values reported from other areas are only a consequence of the characteristic low levels of iron and manganese present in this phase.

The annual mean concentrations as well as percentage contribution of this fraction with respect to total metals, are presented in Tables 5.5 and 5.6 respectively. Significant spatial variations in the percentage distribution of this metal fraction were not detected, except for Cd, Zn, Mn and Pb, which showed significantly higher percentages in the northern part of the estuary. Incidentally, these metals exhibited relatively higher percentages in this phase.

Highly significant correlations were obtained between the pairs Fe3- M3 and Mn3- M3, (M epresents any other metal studied), except for the pairs Fe3- Cu3 and Fe3- Pb3 (Table 5. 11). This behavior was consistent with the assumed composition of the fraction 3 (i.e. Fe- Mn oxide bound) and the known scavenging role of these oxides. In the Cochin estuary, the correlation recorded for the metal pair with Mn was better than that with Fe indicating the higher efficiency of the Mn oxides in scavenging the metals in this estuary. A very high correlation was recorded for the Cd3 - Zn3 pair and thispoints to the similarity in their origin and estuarine behaviour.

The relative abundance of metals determined in this phase was in the order,

 ${\tt Zn} \ {\tt Mn} \ {\tt Cd} \ {\tt Pb} \ {\tt Co} \ {\tt Ni} \ {\tt Fe} \ {\tt Cu} \ {\tt Cr}$ 

Table	ວ <b>ໍ</b> ວ	Annual mean metals in sed	concentr liments o	ration (n f the Cou	ng∕kg) of chin estua)	the fry.	aasily ro	ducible	fraction	of trace
Station		Fe3	Mn3	Zn3	Cu3	Cd3	Pb3	Co3	Ni3	Cr3
	Mean SD	149.08 26.43	7.88 3.71	1.25 0.32	0.05 0.02	0.06 0.03	0.38 0.14	0.38 0.21	0.34 0.08	0.03 0.02
7	Mean SD	149.0250.08	5.13 1.50	$1.59 \\ 0.37$	0.02 0.01	0.07 0.05	0.48 0.26	$0.27 \\ 0.07$	.0.29 0.12	0.03 0.02
e	Mean SD	90.33 15.68	$1.92 \\ 0.79$	$1.72 \\ 0.64$	$0.12 \\ 0.05$	0.05 0.02	0.74 0.44	$0.25 \\ 0.16$	0.34 0.08	0.04 0.02
4	Mean SD	99.63 31.50	3.30 1.79	$2.10 \\ 0.69$	0.05 0.03	0.05 0.02	0.44 0.22	$0.22 \\ 0.12$	0.28 0.08	$0.04 \\ 0.02$
വ	Mean SD	78.88 35.81	13.35 4.15	3.47 1.54	0.03 0.01	0.18 0.03	1.47 0.75	0.45 0.12	0.82 0.13	$0.11 \\ 0.02$
9	Mean SD	189.75 35.13	13.79 4.78	25.06 3.66	0.03 0.01	$0.15 \\ 0.04$	$1.20 \\ 0.42$	0.41 0.15	0.59 0.29	0.06
7	Mean SD	165.77 43.57	8.52 3.60	30.93 8.41	0.02 0.01	$0.34 \\ 0.24$	0.72 0.38	0.45 0.09	0.42 0.11	0.06 0.03
8	Mean SD	44.55 17.03	$1.02 \\ 0.28$	$5.10 \\ 2.57$	$0.02 \\ 0.02$	0.06 0.02	0.47 0.36	0.13 0.05	$0.11 \\ 0.09$	0.02 0.01
6	Mean SD	30.05 9.59	1.71 1.70	5.71 5.16	$0.06 \\ 0.02$	0.08 0.05	1.21 1.53	$0.17 \\ 0.07$	0.15 0.11	0.02 0.01
10	Mean SD	35.19 8.42	1.95 0.36	4.29 1.18	0.02 0.01	0.06 0.02	0.37 0.20	$0.13 \\ 0.02$	0.16 0.06	0.01
Genera	nl Mear SD	1 103.23 62.54	5.86 5.33	8.12 10.71	0.04 0.04	$0.11 \\ 0.12$	$\begin{array}{c} 0.75 \\ 0.72 \end{array}$	$\begin{array}{c} 0.29\\ 0.17\end{array}$	0.35 0.24	0.04 0.03

Table	5.6	Annual mean sediments of	percenta the Coch	ages of nin estua	the eastl ry.	y reduc	ible frac	tion of	trace	metals ir
Station		Fe3	Mn3	Zn3	Cu3	Cd3	Pb3	Co3	N13	Cr3
1	Mean SD	1.96 0.52	7.61 1.58	5.14 1.97	0.89 0.32	11.09 6.95	5.15 1.87	4.56 2.06	2.00 0.54	0.13 0.08
2	Mean SD	$\begin{array}{c} 1.77\\ 0.44 \end{array}$	6.13 1.36	6.08 1.14	0.29 0.13	9.79 5.70	5.67 2.83	2.82 0.62	· 1.53 0.57	$0.12 \\ 0.07$
e	Mean SD	1.40 0.30	4.85 1.33	$7.60 \\ 1.34$	2.13 1.21	6.41 2.37	7.27 2.40	<b>4</b> .21 2.00	2.47 0.94	$0.17 \\ 0.09$
4	Mean SD	1.45 0.44	6.14 2.99	7.59 1.63	$0.73 \\ 0.54$	$6.42 \\ 2.46$	4.82 1.95	3.96 2.60	$1.90 \\ 0.47$	0.18 0.11
വ	Mean SD	$0.53 \\ 0.20$	11.24 2.90	3.86 0.73	0.17 0.13	9.33 4.64	6.16 2.17	3.48 2.12	2.18 1.14	0.21 0.13
9	Mean SD	$1.21 \\ 0.26$	10.46 2.25	12.64 $1.16$	0.11 0.06	2.95 0.74	3.36 1.02	$2.14 \\ 1.10$	$1.10 \\ 0.48$	0.09 0.04
٢	Mean SD	$2.19 \\ 0.61$	12.99 5.44	16.14 2.30	$0.11 \\ 0.07$	8.64 4.44	2.68 0.61	$3.02 \\ 0.94$	1.72 0.56	$0.13 \\ 0.06$
8	Mean SD	0.89 0.33	7.33 2.84	10.36 2.28	$0.42 \\ 0.46$	8.23 2.68	$7.12 \\ 6.96$	5.03 2.04	$1.43 \\ 0.92$	0.14 0.08
6	Mean SD	0.66 0.17	10.36 8.10	9.68 3.49	1.81 1.22	8.62 1.80	9.36 5.75	6.43 3.72	2.63 1.00	0.21 0.08
10	Mean SD	0.65 0.21	$11.96 \\ 3.81$	<b>14.64</b> 2.15	$\begin{array}{c} 0.71 \\ 0.49 \end{array}$	9.87 2.01	5.22 4.86	5.25 1.43	2.83 1.11	0.18 0.07
Genera	l Mear SD	1.27 0.67	8.91 4.67	9.37 4.35	$0.74 \\ 0.91$	8.13 4.45	5.68 4.08	<b>4</b> .09 2.39	1.98 0.97	$0.16 \\ 0.09$

and this order agreed quite well with that reported by Tessier et al. (1980) from Yamaska and St. Francois river sediments, except for Cd and Cu. For amorphous oxides of iron in soils, the affinity series established (Elliot et al. 1986) was

$$Pb \rightarrow Cu \rightarrow Zn \rightarrow Cd.$$

The low affinity of Cu, Co, Pb, Ni and Zn for this phase has been indicated for river sediments (Tessier et al. 1980). The higher levels of Zn and Cd observed in this phase in sediments of the Cochin estuary, could only be a reflection of the anthropogenic inputs. Fiq. 5.7 depicts the relation between the negative logarithm of the solubility product of the respective metal hydroxide and the percentage abundance of the metals in the M3 phase. The relationship indicated was remarkably significant, suggesting that the metal fractions determined truly represented the hydroxide precipitation behaviour in this tropical estuary.

#### 5.2.4. ORGANIC/SULFIDE PHASE

The trace metals solubilised after the destruction of organic matter constituted the organic phase. The reagent used was hot acidified hydrogen peroxide, originally suggested by Gupta and Chen (1975).Sulfide minerals also could be extracted with this reagent. In the Cochin estuary, during the survey the bottom water was never been found to be depleted of dissolved oxygen, so as to cause sulfide precipitation, and hence, the metals extracted at this stage could be regarded as good a estimate of the metals bound to the organic fractions of the sediments. In sediments, the metals may be bound to various forms of organic matter, such as living organisms, detritus, organic coating on mineral particles, etc.



Fig. 5.7 Variations in the percentage of easily reducible fractions with (a) negative logarithm of the solubility product of corresponding metal hydroxide and (b) ionic potential

Data on the spatial distribution of organically bound trace metals in the sediments of Cochin estuary are presented in Tables 5.7 (absolute values) and 5.8 (percentage values). For all the metals, the highest concentration in this fraction was detected in Segment II signifying the influence of estuarine controlling the spatial processes in distribution of organically bound metals. As the river water carrying clays and organic matter entered the brackish water region in the estuary, the changed chemical environment caused coagulation and flocculation of these colloidal particles. However, а significant spatial variation in percentage-wise distribution was noted only in the case of Zn and Cr, both having higher percentages in Segment II. The observed minimum spatial variation in percentage values indicated that the variations in the total content of metals were always associated with a variation in the organic phase of the metal and that both were related to the estuarine processes.

The order of relative abundance of trace metals in this fraction, in the sediments of the Cochin estuary was,

Cd > Mn > Zn > Co > Cu > Pb > Ni > Cr > Fe.

The relative concentration of each metal in this fraction depended upon the free metal ion and the equilibrium constant for metal organic complexes, the ligand concentration being the same for all metals. Based on the model suggested by Tessier et al. (1980), the comparative stability constants of the metal organic complexes (with respect to manganese complexes), in estuarine sediments were calculated and the same is presented below.

		Fe	Zn	Cu	Cd	Pb	Co	Ni	Cr
Segment	I	-0.23	0.31	0.70	0.36	0.12	0.46	0.42	0.92
Segment	II	0.14	0.11	1.06	0.58	-0.32	0.55	0.65	1.28
Segment	III	0.05	0.08	0.56	0.13	-0.17	0.37	0.25	0.48

fable.	5.7	Annual mean metals in se	concentr diments c	ation (mg of the Coc	/kg) of chin estu	the organ ary.	ic/sulfide	punoq	fraction	of trace
Station		Fe4	Mn4	Zn4	Cu4	Cd4	Pb4	Co4	Ni4	Cr4
	Mean SD	55.08 23.60	23.52 5.30	2.54 0.78	$\begin{array}{c} 1.76\\ 0.53 \end{array}$	0.17 0.04	1.35 0.54	$1.80 \\ 0.52$	1.58 0.36	2.20 0.41
2	Mean SD	66.35 30.05	22.68 4.78	2.97 0.67	1.35 0.51	0.18 0.05	$1.33 \\ 0.72$	$1.88 \\ 0.34$	. 1.53 0.32	$1.80 \\ 0.38$
e	Mean SD	68.27 40.11	12.79 10.04	<b>4.89</b> 2.25	$1.23 \\ 0.72$	0.25 0.10	<b>1.41</b> 0.79	$1.57 \\ 1.27$	2.21 1.12	2.19 1.33
4	Mean SD	67.13 36.94	15.56 7.10	<b>4.89</b> 2.03	1.35 0.47	0.23 0.09	1.56 0.80	$1.24 \\ 0.30$	1.86 0.53	$1.75 \\ 0.55$
ß	Mean SD	202.38 88.12	36.03 13.21	28.53 8.25	2.14 1.01	1.21 1.18	$1.04 \\ 0.52$	2.73 1.21	6.52 3.11	6.21 3.51
9	Mean SD	318.43 38.08	38.70 12.19	$67.21 \\ 6.29$	<b>4.61</b> 0.59	2.47 1.09	1.26 0.45	3.88 1.02	8.21 1.73	16.34 4.13
7	Mean SD	185.82 34.03	$19.27 \\ 6.32$	61.15 12.65	8.77 4.92	1.59 0.89	1.33 0.32	3.25 1.49	2.32 0.81	10.62 4.18
8	Mean SD	56.28 17.06	4.53 1.19	13.97 5.60	0.76 0.53	0.25 0.10	0.84 0.46	0.46 0.14	0.53 0.14	0.69 0.39
6	Mean SD	32.18 23.87	2.73 1.10	$14.90 \\ 9.20$	2.26 2.77	0.28 0.17	1.06 0.67	0.82 0.50	0.38 0.27	0.56 0.38
10	Mean SD	23.20 10.62	3.26 0.67	6.02 2.13	0.81 0.35	0.19 0.06	0.46 0.17	$0.44 \\ 0.09$	0.62 0.17	0.65 0.43
Genera	rl Mear SD	n 107.51 99.21	$17.91 \\ 14.29$	20.71 23.87	2.50 2.99	0.68 0.96	1.16 0.65	1.81 1.39	2.58 2.79	<b>4.</b> 30 5.48

Table	5.8	Annual mean sediments of	percenta the Cocl	iges of tr hin estuar	ie organi ry.	c/sulfide	punoq	fraction o	f trace m	letals in
Station		Fe4	Mn4	Zn4	Cu4	Cd4	Pb4	Co4	Ni4	Cr4
1	Mean SD	0.69 0.24	25.09 5.59	9.83 2.14	32.07 10.65	33.45 6.28	18.02 7.27	22.01 4.40	9.19 1.66	10.87 1.02
2	Mean SD	0.81 0.37	27.03 3.48	11.46 2.69	$19.62 \\ 6.35$	28.87 7.32	15.44 5.77	20.11 2.48	. 8.26 1.58	7.35 1.76
e	Mean SD	1.01 0.53	26.54 4.20	20.65 3.12	17.29 7.32	35.88 6.95	14.07 2.76	21.89 4.55	14.11 2.98	9.87 2.64
4	Mean SD	0.96 0.48	27.03 6.06	16.81 3.18	17.75 3.97	33.00 4.65	16.14 5.34	21.57 4.31	12.73 2.45	8.03 1.87
5	Mean SD	1.37 0.44	28.55 4.51	32.65 1.65	11.28 $2.84$	35.93 10.32	<b>4.58</b> 2.23	16.44 2.44	13.56 1.93	8.57 2.84
9	Mean SD	2.04 0.40	28.94 3.68	34.05 2.35	15.34 1.97	44.88 12.65	3.60 1.39	<b>18.98</b> 3.27	$15.77 \\ 2.75$	$21.99 \\ 2.87$
7	Mean SD	2.48 0.60	28.07 4.50	32.17 3.26	30.55 4.56	40.92 8.74	5.62 1.66	<b>19.6</b> 7 3.30	8.86 1.65	23.52 3.25
8	Mean SD	1.15 0.38	29.98 3.20	30.44 6.86	16.64 5.43	33.40 5.85	9.79 2.42	17.18 3.99	7.20 1.21	5.31 1.86
6	Mean SD	$0.72 \\ 0.53$	20.07 7.36	33.37 6.29	25.03 10.37	30.69 5.39	11.10 5.64	23.45 4.76	8.85 4.12	6.46 3.61
10	Mean SD	0.41 0.16	$19.81 \\ 5.40$	21.39 8.16	17.83 3.81	31.97 4.71	5.32 1.91	16.83 3.38	10.85 3.81	8.15 3.89
General	l Mean SD	1.16 0.75	$26.11 \\ 5.98$	24.28 10.01	20.34 9.03	34.90 8.94	10.37 6.58	$19.81 \\ 4.42$	10.94 3.79	$11.01\\6.64$

The stability ratios so worked out followed the lrwing William order for the complexation of divalent metals (Irving and Williams, 1948), namely,

Cu> Ni> Zn> Cd> Fe >Mn.

#### 5.2.5 RESIDUAL PHASE

This phase contains primary as well as secondary weathered minerals which are for most part a very stable portion of the elemental constituents (Brannon et al. 1977). Since metals found in this phase were expected to be chemically stable and biologically inactive, strong acid digestion was required to dissolve the metals out of this phase. Hot nitric acid utilized in the scheme was expected to bring about a near total digestion, with the reaction being vigorous enough to dissolve all the heavy metals except that bound to silicate phases (Ure, 1990). The heavy metals considered in the present study were not likely to be associated to any appreciable degree with silicate minerals.

Fraction 5 represented an essentially non-labile component, and the results confirmed that this form, which was usually designated as residual was an important sink for trace metals in sediments. Without exception, trace metal concentrations found in this fraction (Table 5.9 and 5.10) were higher than those observed in any of the other fractions. For iron, more than 95% was detected in this phase; chromium and nickel also followed the same trend, with the percentages recorded above 75 at all Stations. The residual component in the distribution of cobalt, lead and copper in sediments of the Cochin estuary was percent, the percentage values between 60 and 90 being comparatively lower in the industrial and sewage effluent receiving Stations in the northern parts of the estuary. Of the remaining metals, zinc exhibited significant spatial variations, showing higher percentage contribution (above 70 the southern parts lower percentage percent) in and contribution (less than 50 percent) in the northern parts, and

Table	5.0	Annual mean sediments of	r concentu f the Coc	ration (m hin estua	g/Kg) of ry.	the res	idual fra	action of	trace m	etals in
Statio	ų.	Fe5	Mn5	Zn5	Cu5	Cd5	Pb5	Co5	N15	Cr5
1	Mean SD	7686 1640	35.57 10.94	21.63 5.38	3.57 1.15	0.22 0.06	5.15 1.30	5.77 2.09	15.51 5.65	17.65 2.47
2	Mean SD	8078 816	34.80 5.81	21.03 2.20	$5.24 \\ 0.75$	0.30 0.06	5.77 0.82	6.84 $0.93$	·16.30 1.59	22.67 2.26
e	Mean SD	6416 907	21.27 12.29	16.16 5.82	6.86 6.87	0.35 0.22	6.99 4.09	5.31 4.40	13.34 7.63	$18.95 \\ 6.69$
4	Mean SD	6750 929	30.75 14.32	20.86 7.50	5.95 2.00	0.30 0.10	6.47 1.95	$3.94 \\ 0.90$	11.85 1.45	19.87 4.78
S	Mean SD	$14254 \\ 2095$	54.82 18.67	51.46 10.38	$16.69 \\ 6.57$	1.34 1.19	20.31 6.94	$12.50 \\ 4.78$	38.39 15.23	58.84 19.86
9	Mean SD	$15435 \\ 2416$	54.91 18.25	80.00 6.94	25.33 3.83	2.38 0.51	31.68 4.32	15.47 2.22	43.48 10.19	57.29 13.88
7	Mean SD	7332 1090	30.22 9.41	68.22 13.41	19.20 9.85	1.56 0.46	22.85 9.81	$12.02 \\ 4.02$	23.15 8.35	33.12 8.75
œ	Mean SD	4902 812	7.64 2.62	22.18 9.37	3.03 0.94	0.28 0.05	6.75 3.67	$1.97 \\ 0.65$	6.48 1.29	11.30 4.35
6	Mean SD	4590 1452	9.37 5.69	18.46 11.70	3.74 3.24	$0.29 \\ 0.12$	6.58 3.82	$\begin{array}{c} 2.16 \\ 0.95 \end{array}$	<b>4.</b> 39 2.78	7.29 3.04
10	Mean SD	5582 1159	8.37 3.24	15.75 5.01	3.45 1.60	0.28 0.06	7.71 2.11	$2.03 \\ 0.71$	<b>4</b> .82 0.68	6.39 1.88
Gener	al Mean SD	8103 3828	28.77 20.19	33.57 24.14	9.31 8.96	0.73 0.85	12.03 10.05	6.805.34	17.77 14.70	25.34 19.99

Table	5.10	Annual mea the Cochin	n percent estuary.	ages of t	he residu	lal fracti	on of tra	ce metals	in sedi	nents of
Statio	Ę	Fe5	Mn 5	Zn5	Cu5	Cd5	Pb5	Co5	Ni5	Cr5
1	Mean	97.25	36.16	82.98	62.48	42.06	67.27	67.43	85.80	87.75
	SD	0.48	5.87	3.14	10.01	8.17	9.00	6.94	2.87	1.28
2	Mean SD	97.32 0.40	41.91 6.14	80.72 2.17	76.75 6.56	48.71 5.59	70.93 7.63	72.92 2.42	.87.66 1.33	$\begin{array}{c} 91.72 \\ 2.01 \end{array}$
e	Mean	97.44	48.69	69.64	76.41	44.80	69.02	68.40	79.66	88.43
	SD	0.61	3.60	4.29	8.65	9.37	3.03	5.57	4.29	2.65
4	Mean	97.43	53.17	72.64	76.87	43.17	71.15	67.68	81.75	90.66
	SD	0.93	6.22	2.68	4.67	6.74	7.47	5.26	2.67	2.05
5	Mean SD	98.05 0.61	$44.02 \\ 4.20$	60.07 3.50	85.86 3.22	44.94 4.79	83.29 2.00	76.66 3.48	82.70 1.62	90.272.51
9	Mean SD	96.51 0.59	41.00 4.35	40.53 2.41	83.32 2.06	48.10 13.30	88.46 1.94	76.54 2.83	81.62 2.41	$77.18 \\ 2.83$
2	Mean	95.11	43.89	35.98	67.69	43.58	88.17	74.74	87.91	75.46
	SD	1.14	5.13	3.04	4.79	7.95	1.95	3.36	1.76	3.32
8	Mean	97.87	49.73	46.29	74.75	38.02	77.63	70.13	88.36	92.49
	SD	0.67	4.80	3.36	4.69	5.82	6.20	7.29	1.88	1.97
6	Mean	98.54	57.21	39.96	63.65	37.19	67.15	66.24	84.39	89.60
	SD	0.44	14.08	5.99	8.51	11.91	8.77	4.73	5.50	3.54
10	Mean SD	98.90 0.29	47.27 8.64	53.43 8.55	74.34 5.61	46.75 5.38	84.73 5.93	72.73 4.98	82.48 3.96	88.89 3.04
Genera	al Mean	97.44	46.30	58.22	74.21	43.73	76.78	71.35	84.23	87.25
	SD	1.21	9.10	17.19	9.66	9.13	10.24	6.17	4.25	6.22

this signified the anthropogenic input of zinc to the northern parts of the estuary. Cadmium and manganese seemed to be the most loosely bound metals of the environment, since the percentages of residual fraction of these metals detected was the lowest in all the Station in the south (S1 to S4) and was comparable with that of zinc in the northern parts.

The overwhelming importance of residual metals was pointed out in studies on other estuarine sediments too (Brannon et al. 1976; Gibbs, 1977; Tessier et al. 1980; Yetang and Forstner, 1984; Lee, 1985; Martin et al. 1987). The results, thus, clearly illustrated the difficulty of distinguishing between background and anomalous levels of trace metal contamination when only total metal analyses were performed. The order of abundance of metals in the residual phase of the sediments of the Cochin estuary was,

Fe> Zn> Mn > Cr> Ni> Pb> Cu> Co> Cd,

and with the exception of zinc, this represented the order of crustal abundance of these metals. Similar observations, where zinc exceeded manganese in abundance, were reported from the Sepitiba bay (Lacerda et al. 1987), a polluted estuary on the Brazilian coast. However, when the percentage values were considered the order of abundance was,

Fe> Cr> Ni> Pb> Cu> Co> Zn> Mn> Cd

and this signified the high environmental mobility of zinc, manganese and cadmium. Low percentages of zinc, cadmium and manganese in the residual fraction were reported for several other estuaries (Grieve and Fletcher, 1976; Forstner, 1982; Calmano and Forstner, 1983; Lee, 1985; Rosental et al. 1986; Samanidou and Fytianos, 1987; Prohic and Kniewald, 1987; Pardo et al. 1990; Elsokkary and Muller, 1990; Bradley and Cox, 1990). The selective extraction thus showed the existence of a category of transition metals which were relatively depleted in the residual fractions.

nts I							•
oxtracteu	Cr1	0.34	Cr2	0.05	Cr3	0.27	cont d
fferent e	Mn1	0.57 0.05	Mn2	0.44 0.18	Mn3	0.43 0.52	0
ib yd be	Nil	0.37 0.67 0.06	Ni2	0.17 0.35 0.61	Ni3	0.38 0.70 0.77	
s extract	Zn1	0.07 0.20 0.17 0.05	Zn2	0.54 0.05 0.25 0.01	Zn3	$\begin{array}{c} 0.51\\ 0.25\\ 0.42\\ 0.30\end{array}$	
ətal pairs	Pb1	0.36 0.65 0.76 0.30	Pb2	$\begin{array}{c} 0.16 \\ 0.49 \\ 0.26 \\ 0.22 \\ 0.23 \end{array}$	Pb3	0.06 0.30 0.33 0.37 0.28	
ces of me ues +ve)	Co1	0.43 0.57 0.23 0.69 0.17 0.66	Co2	$\begin{array}{c} 0.03\\ 0.44\\ 0.44\\ 0.53\\ 0.07\\ 0.19\end{array}$	Co3	0.50 0.54 0.56 0.50 0.36 0.22	                   
ion matri( ; all val	Cd1	$\begin{array}{c} 0.19\\ 0.58\\ 0.58\\ 0.38\\ 0.38\\ 0.57\\ 0.57\\ 0.28\\ 0.28\end{array}$	Cd2	$\begin{array}{c} 0.10\\ 0.23\\ 0.11\\ 0.06\\ 0.56\\ 0.46\\ 0.04\end{array}$	Cd3	$\begin{array}{c} 0.23\\ 0.41\\ 0.38\\ 0.44\\ 0.74\\ 0.26\\ 0.38\\ 0.38\end{array}$	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
Correlat: (n = 120	Cu1	$\begin{array}{c} 0.13\\ 0.28\\ 0.28\\ 0.21\\ 0.78\\ 0.39\\ 0.16\\ 0.55\end{array}$	Cu2	$\begin{array}{c} 0.15\\ 0.23\\ 0.23\\ 0.13\\ 0.14\\ 0.58\\ 0.69\\ 0.45\\ 0.45\end{array}$	Cu3	$\begin{array}{c} 0.10\\ 0.05\\ 0.21\\ 0.03\\ 0.23\\ 0.04\\ 0.05\\ 0.21\end{array}$	2 1 1 1 1 1 1 1 1 1 1 1 1 1 1
5.11							
Table		Fe1 Cr1 Mn1 Ni1 Zn1 Pb1 Co1 Co1		Fe2 Cr2 Mn2 Ni2 Zn2 Pb2 Co2 Cd2		re3 Cr3 Mn3 Ni3 Zn3 Pb3 Co3 Cd3	         

Table 5.11 c	ontd							
	Cu4	Cd4	Co4	Pb4	Zn4	Ni4	Mn4	Cr4
Fe4 Cr4 Mn4 Ni4 Zn4 Pb4 Co4 Cd4	0.50 0.68 0.32 0.32 0.32 0.75 0.29 0.68	0.79 0.91 0.64 0.75 0.85 0.07 0.81	0.74 0.86 0.79 0.77 0.72 0.33	$\begin{array}{c} 0.11 \\ 0.16 \\ 0.31 \\ 0.14 \\ 0.11 \end{array}$	0.82 0.89 0.49 0.63	0.82 0.80 0.82	0.69 0.68	0.84
	Gu5	Cd5	Co5	Pb 5	Zn5	N15	Mn 5	Cr5
Fe5 Cr5 Mn5 Ni5 Zn5 Pb5 Co5 Cd5	0.69 0.84 0.86 0.86 0.89 0.95 0.92 0.92	0.69 0.80 0.80 0.80 0.88 0.88 0.88	0.75 0.89 0.80 0.93 0.83 0.86	0.71 0.82 0.60 0.82 0.92	0.70 0.78 0.60 0.77	0.82 0.95 0.86	0.75 0.84	0.83
Martin et al. (1987) opined that characterization of the `residual' background on relative basis might be misleading so far as a decrease in residual fraction could only result in the instance of an additional input of an anthropogenic labile material. An operationally defined background level, can however be, presented, on concentration basis, integrating various lithologic and climatic variations. The mean data on residual trace metal concentration recorded at the three segments of the Cochin estuary are presented in Table 5.12. For comparison, the average values for Indian river sediments (Subramanian et al. 1985), world surface soils and earth's crust (Martin and Whitfield 1983) are also presented. Except for manganese and cadmium, the values recorded in Segment TT were comparable to that of crustal abundance indicating that the major source of the trace metals studied was weathered products of earth's crust. In comparison to the average values for Indian sediments, the sediments in Segment II of the Cochin estuary were enriched in zinc and cadmium. In the other two segments the values were comparatively low because of the coarse texture of the sediments, but the order of abundance was the same as recorded in earth's crust. Further, the significant positive correlation (Table 5.11) recorded between different metal pairs determined in this fraction testified that the metals studied had a common source, and this metal phase underwent little chemical transformation within the estuarine environment.

## 5.3 PHASE CONCENTRATION FACTOR

Forstner (1979c) introduced theconcept of phase concentration factor (PCF) to compare the metal enrichment in samples with large differences in metal concentrations. PCF is a relative value, and is not influenced by the total metal The PCF is the ratio between the content. percentage of а certain metal association to the percent content of the respective carrier material. If the computed value is above 1,

Table	5.12	Mean residu estuary (se	al metal gment-wis	content in se e) [mg/kg]	diments of	the Cochin
Metal	Seg I	Seg II	Seg III	Indian Sediments*	World Soils **	Earth Crust **
ъ	7429	12742	5102	29000	40000	35900
Mn	72	108	16	605	1000	720
Zn	26	158	42	16	06	127
Cu	7	26	Ŋ	28	30	32
Cd	0.64	3.96	0.74	NA	0.35	0.20
Pb	6	29	6	NA	35	16
Co	8	18	e	31	8	13
Ni	17	42	9	37	50	49
Сr	22	61	6	87	70	71
NA =	Data not	available				
ری ۱۱ *	sub raman	iam, et al.	1985, *	** = Martin ar	d Whitfield	, 1983

an enrichment in the studied phase is inferred; if the value is less than 1, then , it is indicative of dilution of metal concentration in the sample due to the presence of that phase.

Enrichment of all the metals in the organic phases have been worked out, on the basis of above principles. The organic carbon values were used to quantify the organic phase. The PCF of all the samples studied were worked out, and the averaged, season-wise and segment-wise, values are plotted in Fig. 5.8 All the metals, except iron, showed a factor above 1 indicating that the metal values were enriched by the presence of organically bound metals. This enrichment was a maximum for the sediments of Segment III and a minimum for Segment II. Seasonally, it was during the monsoons, when estuarine riverine environment was dominated by the inputs from terrestrial sources that the influence of organic bound metals on the total pool was a maximum.

## 5.4 ESTUARINE ENRICHMENT

For studying the estuarine behaviour of the various metal phases determined, the metal values recorded at the middle estuary (ME) were compared to that of upper estuary (UE) and lower estuary (LE). Values from Station 1 in the southern end and Station 10 in the northern end were taken as representative of the upper estuarine regions. Station 3 values in the southern arm and Station 7 values in the northern arm represented the middle estuarine region. Metal content at Station 5 (barmouth) was the common factor representing lower estuarine region of both arms of the estuary. To nullify the effects of other sedimentological parameters on chemical partitioning, the ratios of the metal fractions between the two sites were compared to the ratio of total metal between the respective sites and the relationship so obtained is plotted in Fig.5.9 to 5.13.



Fig. 5.8 Phase concentration factor, (a) spatial variation and (b) seasonal variation?

Figure 5.9a on residual metals indicated that, between the upper estuary and middle estuary, in the northern part, the total concentration of all the metals showed enrichment in sediments in the middle estuary. The enrichment was minimum for iron, but was 6 to 7 times for the metals, zinc, chromium, cobalt and copper. Lead, manganese and nickel showed 3 to 5 times enrichment. The residual fractions also exhibited, a similar enrichment and suggested that this fraction had not undergone any appreciable change in this part of the estuary. The enrichment in total metal concentration could be attributed to the difference in particle size distributions whereas the variations in the magnitude of enrichment for various metals could be traced to the segregation of metals by coagulation, as demonstrated by Gibbs (1986). Between the middle estuary and the lower estuary on the northern part (Fig. 5.9b) appreciable enhancement was noted for the metals, iron, manganese, nickel and chromium in the lower estuary, for both total and residual fractions, whereas no significant change was noted for cobalt. On the other hand, the metals, lead, cadmium, copper and zinc showed depletion in the lower estuary. Maximum depletion was in the case of zinc. In the noted southern part of the estuary, (Fig. 5.9c) middle estuarine depletion was noted for manganese whereas for zinc, iron, nickel, cobalt and chromium, appreciable change could not be observed. As observed in the northern parts, enrichment was observed for, lead, cadmium and in both total and residual fraction although copper the enrichment was only 1.5 to 2.0 times. Between the middle estuary and lower estuary in the southern part(Fig. 5.9d), enrichment was noted in the lower estuary, for the total trace metals and the residuals co-varied with totals, signifying the minimal reactivity of residual phase under estuarine conditions.

The estuarine reactivity of organically bound metals is depicted in Fig. 5.10 (a - d). The enrichment of lead, nickel, and cobalt in this phase paralleled that of total metals in the middle estuary (Fig. 5.10a) of the northern arm, indicating





that these metal had undergone little change through organic recycling. For iron and manganese, the enhancement in organic phase was slightly higher (8 and 6 times, respectively) when compared with enrichment (1.5 and 4 times) in the total amounts. Maximum enrichment was recorded for chromium (17)times). Cadmium, zinc and copper were more enriched in the organic phase than the enrichment recorded in total metals. Between the middle estuary and lower estuary in the northern part (Fig. 5.10b), organic phase of copper and chromium showed depletion in the lower estuary, and the depletion was 3 and 2 that recorded for total times higher, respectively, than metals. Organically bound iron also exhibited slight depletion estuary. For the other in the lower metals depletion/ enrichment recorded was parallel to that of total trace metals. As observed for residual manganese, organic bound manganese also was depleted in the middle estuary in the southern arm (Fig 5.10 c). Although total copper was enriched in the middle estuary, corresponding increase in the organic phase was not noted, signifying phase transfer during estuarine processes. For iron, nickel and zinc, the enrichment in organic phase was slightly higher than the corresponding increase in total trace metal levels, but for cobalt, chromium, lead and cadmium, increase in organic phase paralleled that of increase in total metal. Between the lower and middle estuary in the southern part (Fig. 5.10d) enrichment was noted for all the metals (except lead) in the organic phase, as also noted for total metal concentration indicating that dynamic equilibrium was attained in phase transformation through organic cycling.

The behaviour of easily reducible forms of trace metals is depicted in Fig. 5.11 (a- d). The inadequacy of data on copper in this phase has already been pointed out (Chapter 4). For iron, 5 times enrichment was recorded in this phase in the middle estuary (Fig. 5.11a) which was 4 times higher than the enrichment in totals. For zinc and cadmium also slightly high enrichment in this fraction was noted than the increase in totals. For the remaining metals, the enrichment in this phase



was paralleled by an enrichment in totals also, in the northern part of the estuary. However, between middle estuary and lower estuary, in the north, appreciable depletion was noted in the lower estuary for iron, zinc and cadmium in this phase (Fig. 5.11b), whereas, lead recorded enrichment. Chromium, nickel and manganese covaried with totals and cobalt concentrations were unchanged. Between upper and middle estuary in the south (Fig. 5.11c), the depletion/ enrichment in this phase followed the same trend, as was recorded for totals. Between middle and lower parts of the southern estuary (Fig. 5.11d), manganese showed enrichment in this phase; iron and copper exhibited depletion. Other metals did not exhibit any changes between the sites compared.

Except for iron, manganese and zinc, the trace metals in the carbonate bound phase exhibited depletion in the middle estuary, when compared with the status of total trace metals, in the northern arm of the estuary (Fig. 5.12a). While, manganese showed little change, iron fraction showed maximum enrichment (10 times) in the middle estuary, followed by zinc (2 times). Maximum depletion was noted for the carbonate bound iron, and zinc in the lower estuary (Fig. 5.12b) whereas copper and lead in this phase was found to increase. In the southern parts, iron and zinc in this phase (Fig. 5.12c) showed enrichment in the middle estuary when compared to the upper estuary and manganese and cobalt exhibited depletion. No significant changes could be detected for other metals in this phase. Between middle and lower estuary in the south, iron. nickel and chromium showed depletion, while zinc in this phase showed an enhancement in the lower estuary. The carbonate phase of all other metals paralleled the distribution of the total content.

Significant estuarine reactivity was recorded for the metals in the exchangeable phase (Fig. 5.13 a-d). Between middle and upper estuary in the north, the enhancement in the total metals was not followed by a similar enhancement in exchangeable

118





phases, indicating that this phase had undergone rapid transformation between these two segments in the estuary. Iron in this phase was found to get enriched (Fig. 5. 13a) in the middle estuary ; lead also recorded slight enrichment, but, depletion was noted for all other metals. On the other hand, the `totals' of all the metals had recorded enrichment in the middle estuary, and the depletion in the concentration of exchangeable fraction was a convincing evidence for the desorption of trace metals from sediments in lower saline regions of this tropical estuary. This depletion was further aggravated in the lower estuary for all the metals, and the depletion was maximum for zinc (Fig. 5.13b). In the southern parts, however, the exchangeable fraction was found to co-vary with totals between the upper and middle estuary (Fig 5.13c), except for manganese, which exhibited depletion and cobalt, chromium and nickel, which showed enrichment. Between middle and lower estuary in the south, all the metals , excepting zinc, exhibited depletion in this phase. Similar observations were recorded by Fletcher et al. (1983) at the Fraser estuary. The reduction in exchangeable trace metals under the influence of was attributed to desorption. salinity However, an alternative explanation for these observations could include the landward transport of compositionally distinct marine particles in the saltwater intrusion as suggested by Forstner et al. (1990) for the Elbe estuary, or compositional variation reflecting changes in the size distribution of sediments, as pointed out by Nolting et al. (1990) for the Rhine and Meuse estuaries. Textural analysis had shown that the sediments at Station 5 (lower estuarine station) contained relatively higher proportion of coarser particles than that at Station 6, except during the monsoon season (Fig. 3.2). Consequently the relatively low extractable metals observed in this region of salt intrusion might simply reflect the increased proportion of coarse particles. However, a comparison of the reduction in total metals with that of metals in the exchangeable phase showed that desorption was a major process deciding the fate of trace metals in this tropical estuary.

The sequential analysis, thus clearly brings out the estuarine reactivity of trace metals. While the residual fractions had undergone little chemical change, within this estuary, all the other fractions exhibited notable changes. But the direction and magnitude of the change varied widely between the metals studied and also between the two arms of the Cochin estuary.

## 5.5 THE OUTLOOK

Sequential chemical extraction schemes have opened up new vistas for closer examination of the fate of trace metals in the environment. The techniques have become invaluable in studying the terrestrial transport and ultimate removal/ transformation of trace metals. Chemodynamics of trace metals in the continuously changing environment of a tropical estuary, now stand highlighted.

Environmental behaviour of trace metals is receiving increased attention with added emphasis on sources, sinks, contamination levels and bioavailability. This necessitates advancements in developing simpler, rapid reliable andaffordable methodologies. Although, the sequential extraction techniques, employed herein, can be advantageously used in delineating the boundaries and also in predicting the transfer processes between various media in the environment, refinement of techniques, both in analytical terms and conceptual lines, need to be made. Techniques for physical fractionation, sub-phase fractionation, and mathematical systems modeling are potential areas for further exploration. There is also a need to coordinate and integrate the various approaches in chemical fractionation so as to define the bio-geo-physico-chemical factors that influence and regulate the environmental impact of trace metals.

## References

- Abraham, T.J., Salih, K.Y.M. and Chacko, J., 1986. Effect of heavy metals on the filtration rate of bivalve Villorita cyprinoides (Hanley) var.cochinensis. Indian J. Mar. Sci., 15: 195-196
- Ackroyd, D.R., Bale, A.J., Howland, R.J.M., Knox, S., Millward, G.E. and Morris, A.W., 1986. Distributions and behaviour of dissolved Cu, Zn and Mn in the Tamar estuary. *Estuar. Coast. Shelf Sci.*, 23: 621-640
- Ackroyd, D.R., Millward, G.E. and Morris, A.W., 1987. Periodicity in the trace metal content of estuarine sediments. Oceanol. Acta, 10: 161-168
- Adams, D.D., Darby, D.A. and Young, R.J., 1980. Selected analytical techniques for characterising the metal chemistry and geology of fine grained sediments and interstitial waters. In: Baker, R.A. (ed.), *Contaminants* and Sediments, Vol 2, Ann Arbor Sci. Publ. Inc., Ann Arbor. pp 3- 28
- Adediran, S. and Kramer, J.R., 1987. Copper adsorption on clay, iron-manganese oxide and organic fractions along a salinity gradient. Appl. Geochem., 2: 213-216
- Agemian, H. and Chau, A.S.Y., 1976. Evaluation of extraction techniques for the determination of metals in aquatic sediments. *Analyst*, 101, 761-767
- Agemian, H. and Chau, A.S.Y. 1977. A study of different analytical extraction methods for non-detrital heavy metals in aquatic sediments. Arch. Environ. Contam. Toxicol., 6: 69-82

121

Aguilera, N.H. and Jackson, M.L., 1953. Iron oxide removal from soils and clays. Soil Sci. Soc. Amer. Proc., 17: 359 -364

- Allan, R.J., 1990. Estuarine and coastal water contamination: an issue for the 1990s. Sci. Total Environ. 97/98: 1-5
- Allen, J.R.L., Rae, J.E. and Zanin, P.E., 1990. Metal speciation (Cu, Zn, Pb) and organic matter in an oxic salt marsh, Severn estuary, Southwest Britain. Mar. Pollut. Bull., 21: 574-580
- Alloway, B.J., 1990. Cadmium. In: Alloway, B.J. (ed.), *Heavy* Metals in Soils. Blackie and Son, London. pp 100-124
- Amdurer, M., Adler, D. and Santschi, P.H., 1983. Studies of the chemical forms of trace elements in seawater using radiotracers. In: Wong, C.S., Boyle, E., Bruland, K.W., Burton, J.D., and Goldberg, E.D. (eds.), Trace Metals in Sea Water, Plenum Press, New York. pp 537-562
- Ananthakrishnan, R., Parthasarathy, B. and Patham, J.M., 1979. Meteorology of Kerala. Contributions to Marine Sciences. Dedicated to Dr. C. V. Kurian, Univ. Cochin. pp 60-125
- Anderson, P.R. and Christensen, T.H., 1983. Parameters controlling the distribution of Cd, Co, Ni and Zn in soils. *Proc. Int. Conf. on Heavy Metals in the Environment*, Hiedelberg. CEP Consultants. pp 1187-1190
- Anderson, R.A., 1981. Nutritional role of chromium. Sci. Total Environ., 17: 13-29
- Anirudhan, T.S., 1988. Studies on the Nutrient Chemistry of a Tropical Estuary. Ph.D. Thesis. Cochin University.
- Anirudhan, T.S., Balchand, A.N., Nair, S.M. and Nambisan, P.N.K., 1987. Distribution pattern of salinity and silicon and their interrelationship in Cochin backwaters. Proc. Natn. Sem. Estuarine Management, Trivandrum, Kerala. pp 26-31
- Anirudhan, T.S. and Nambisan, P.N.K., 1990. Distribution of salinity and silicon in Cochin estuary. Indian J. Mar. Sci., 19: 137-139
- Anto, A.F., Udaya Varma, P. and Krishna Iyer, H., 1977. Siltation in the outer channels of Cochin harbour. Indian J. Mar. Sci., 6, 76-79

Anwar Batcha, S.M., 1985. Studies on Bottom Fauna of North Vembanad Lake. Ph.D. Thesis, Cochin University.

Arrhenius, G.O.S. and Korkish, J., 1959. Uranium and thorium in marine minerals. Int. Ocean. Congress Am. Assoc. Adv. Sci. (preprints). 497p.

Aston, S.R., 1978. Estuarine Chemistry. In: Riley, J.P. and Chester, R. (eds.), Chemical Oceanography, Vol. 7. Academic Press, London. pp 361-440

Baby, K.V., 1987. Combined Toxicity of Heavy Metals and Petroleum Hydrocarbons on Selected Marine Organisms. Ph.D. Thesis. Cochin University.

Baby, K.V. and Menon, N.R., 1986. Oxygen uptake in the brown mussel *Perna indica* (Kuriakose and Nair) under sublethal stress of Hg, Cd and Zn. *Indian J. Mar. Sci.*, 15: 127-128

Baker, D.E., 1990. Copper. In Alloway, B.J. (ed.) *Heavy* Metals in Soils. Blackie and Son Ltd., London. pp 151-176

Balakrishnan, K.P. and Shynamma, C.S., 1976. Dial variations in hydrographic conditions during different seasons in the Cochin harbour (Cochin backwaters). Indian J. Mar. Sci., 5: 190-195

Balchand, A.N., 1984. Studies on the Dynamics and Water Quality of Muvattupuzha river in relation to effluent discharge. Ph.D. Thesis. Cochin University.

Balchand, A.N., Nair, S.M. and Nambisan, P.N.K., 1990. A survey of the protein content in estuarine waters. *Toxicol. Environ. Chem.*, 27: 285-296

Balchand, A.N. and Nambisan, P.N.K., 1986. Effect of pulp paper effluents on the water quality of Muvattupuzha river emptying into Cochin backwaters. Indian J. Mar. Sci., 15: 253-259

Bale, A.J. and Morris, A.W., 1981. Laboratory simulation of chemical processes induced by estuarine mixing-the behaviour of iron and phosphate in estuaries. *Estuar. Coast. Shelf Sci.*, 13: 1-10

Balistrieri, L.S. and Murray, J.W., 1982. The adsorption of Cu, Pb, Zn and Cd on goethite from major ion sea water. Geochim. Cosochim. Acta, 46: 1253 - 1265

- Bassett, J., Denney, R.C., Jeffery, G.H. and Mendham, J., 1978. Vogel's Textbook of Quantitative Inorganic Analysis. ELBS and Longman, London. 925p
- Belzile. N., Lecomte, P. and Tessier, A., 1989. Testing readsorption of trace elements during partial chemical extractions of bottom sediments. *Environ. Sci. Technol.*, 23: 1015-1020
- Bennet, B.G., 1982. Summary exposure assessment Chromium. In: Exposure Commitment Assessment of Environmental Pollutants, Vol.2. MARC Report No. 28, Monitoring and Assessment Research Center, Chelsea College, University of London. pp 30-42 (cited in Thiessan et al. 1988)
- Bermond, A. and Sommer, G., 1989. Simulation of heavy metal extraction in soil samples compared with experimental results. *Environ. Technol.Lett.*, 10: 989-994
- Bernhard, M., Brinckman, F.E. and Irgolic, K.J., 1986. Why `speciation' .In: Bernhard, M., Brinckman F.E. and Sadler, P.J. (eds.), The Importance of Chemical Speciation in Environmental Processes. Springer- Verlag, Berlin. pp 7-14
- Bero, A.S. and Gibbs, R.J., 1990. Mechanisms of pollutant transport in the Hudson estuary. Sci. Total Environ., 97/98: 9-22
- Bewers, J.M. and Yeats, P.A., 1980. Behaviour of trace metals during estuarine mixing. In: Martin, J.M., Burton, J.D. and Eisma, D. (eds.), *River inputs to ocean systems* UNEP/UNESCO, Paris. pp 103-115
- Biksham, G. and Subramanian, V., 1988. Elemental composition of Godavari sediments (central and southern Indian subcontinent). *Chem. Geol.*, 70: 270-286
- Bilinski, H., Kozar, S., Plavsic, M., Kwokal, Z. and Branica, M., 1991. Trace metal adsorption on inorganic solid phases under estuarine condition. Mar. Chem., 32: 225-233
- Borole, D.V., Sarin, M.M. and Somayajulu, B.L., 1982. Composition of Narmada and Tapti estuarine particulates and adjacent Arabian sea sediments. *Indian. J. Mar. Sci.*, 11: 51-62
- Bourg, A.C.M., 1983. Role of fresh water /sea water mixing on trace metal adsorption. In: Wong. C.S., Boyle, E., Bruland, K.W., Burton, J.D. and Goldberg, E.D. (eds.), Trace Metals in Sea Water. Plenum Press, New York. pp 195-208

- Bowden, K.F. 1980. Physical factors salinity, temperature. circulation and mixing processes. In: Olausson, E. and I. Cato, I. (eds.), Chemistry and Biogeochemistry of estuaries. Wiley-Interscience. pp 37-70
- Bowers, A.R. and Huang, C.P., 1987. Role of Fe(III) in metal complex adsorption by hydrous solids. *Wat. Res.*, 21: 757-764
- Boyle, E.A., Edmond, J.M. and Sholkovitz, E.R., 1977. The mechanism of iron removal in estuaries. *Geochim. Cosmochim. Acta*, **41**: 1313-1324
- Bradley, S.B. and Cox, J.J., 1990. The significance of the floodplain to the cycling of metals in the river Derwent catchment, U.K. Sci. Total Environ., 97/98: 441-454
- Brannon, J.M., Rose, J.R., Engler, R.M. and Smith, I., 1977. The distribution of heavy metals in sediment fractions from Mobile Bay, Alabama. In: Yen, T.F. (ed.), Chemistry of Marine Sediments. Ann Arbor Sci. Publ. Inc. Mich., pp 125-149
- Brook, E.J. and Moore, J.N., 1988. Particle size and chemical control of As, Cd, Cu, Fe, Mn, Ni, Pb and Zn in bed sediment from the Clark Fork river, Montana (USA). Sci. Total Environ., 76: 247-266
- Bruland, K.W., Bertine, K., Koide, M. and Goldberg, E.D., 1974. History of metal pollution in southern California Coastal Zone. Environ. Sci. Technol., 8: 425-432
- Bruemmer, G.W., Gerth, J. and Tiller, K.G., 1988. Reaction kinetic of the adsorption of Ni, Zn and Cd by geothite. 1 Adsorption and diffusion of metals. J. Soil Sci., 39: 37-52
- Burrow, N.J., 1987. The effect of phosphate on zinc sorption by a soil. J. Soil Sci., 38: 453-459
- Burton, J.D., 1978. Behaviour of some trace chemical constituents in estuarine waters. Pure Appl. Chem., 50: 385-393
- Burton, J. D. and Liss, P.S. (eds.) , 1976. Estuarine Chemistry. Academic Press, London. 229p.
- Butler, A.T. and McManus, J. 1979. Sediment sampling and analysis In: Dyer, K.R. (ed.), *Estuarine Hydrography and Sedimentation*. Cambridge University Press. pp 87-131

- Calmano, W. and Forstner, U., 1983. Chemical extraction of heavy metals in polluted river sediments in Central Europe. Sci.Total Environ., 28: 77-90
- Campbell, P.G.C. and Tessier, A., 1989. Geochemistry and bioavailability of trace metals in sediments. In: Boudou, A. and Ribeyre, F. (eds.), Aquatic Ecotoxicology: Fundamental Concepts and Methodologies. CRS Press Inc .Boca Raton. pp 125-148
- Campbell, J.A., Whitelaw, K., Riley, J.P., Head, P.C. and Jones, P.D., 1988. Contrasting behaviour of dissolved and particulate Ni and Zn in a polluted estuary. Sci. Total Environ., 71: 141-155
- Carey, J.H., 1990. The influence of a salt wedge and tidal flow dynamics on contaminant path ways in the Fraser river estuary, British Columbia. In: Michaelis, W. (ed.), *Estuarine Water Quality Management*. Springer-Verlag, Berlin. pp 203-208
- Carruesco, C. and Lapaquellerie, Y., 1985. Heavy metal pollution in the Arcachon Basin (France): Bonding states. *Mar. Pollut. Bull.*, 16: 493-497
- Cary, E.E., Allaway, W.H. and Olsen, O.E., 1977. J. Agric. Food Chem. 25: 305-309 (Cited in McGrath and Smith, 1990)
- Chao, L.L., 1972. Selective dissolution of manganese oxides from soils and sediments with acidified hydroxylamine hydrochloride. Soil. Sci. Soc. Am. Proc., 36: 764-768
- Chapman, H.D., 1965. In Black, C.A. (ed.), Methods of Soil Analysis. American Society of Argonomy, Madison. 891-904
- Cheriyan, P.V., 1967. Hydrographical studies in and around the Cochin harbour. Bull. Dept. Mar. Biol. Oceanogr. Univ. Kerala, 3: 9-17
- Chester, R., 1990. Marine Geochemistry. Unwin Hyman, London. 698p.
- Chester, R. and Hughes, M.J., 1967. A chemical technique for the separation of ferro-manganese minerals, carbonate minerals and adsorbed trace elements from pelagic sediments. *Chem. Geol.*, 2: 249-262
- Chester, R. and Hughes, M.J., 1969. The trace element geochemistry of a North pacific pelagic clay core. *Deep Sea Res.*, 16: 639-654

- Chester, R., Kudoja, W. M., Thomas, A. and Towner, J., 1985. Pollution reconnaissance in stream sediments using non-residual trace metals. *Environ. Pollut.*, 10: 213-238
- Chester, R., Thomas, A., Lin, F.J., Basaham, A.S. and Jacinto, G., 1988. The solid state specification of copper in surface water particulates and oceanic sediments. *Mar. Chem.*, 24: 261-292
- Chisolm, J.J., 1971. Lead poisoning. Sci. Am., 224: 15 23
- Church, T.M., 1986. Biogeochemical factors influencing the residence time of microconstituents in a large tidal estuary, Delaware Bay. Mar. Chem., 18: 393-406
- Clevenger, T. E., 1990. Use of sequential extraction to evaluate the heavy metals in mining wastes. Water Air Soil Pollut., 50: 241-254
- Coonley, L.S., Baker, E.B. and Holland, H.D., 1971. Iron in the Mullica river and in Great Bay, New Jersey. Chem. Geol., 7: 51-63
- Cooper, B.S. and Harris, R.C., 1974. Heavy metals in organic phases of river and estuarine sediment. Mar. Pollut. Bull. 5: 24- 26
- Craig, P.J., 1986. Chemical species in industrial discharges and effluents. In: Bernhard, E., Brinckman, F.E. and Sadler, P.J. (eds.), The Importance of Chemical "Speciation" in Environmental Processes, Springer-Verlag, Berlin. pp 443-464
- Danielsson, L.G., Magnusson, B., Westerlund, S. and Zhang, K., 1983. Trace metals in the Gota river estuary. Estuar. Coast. Shelf Sci., 17: 73- 85
- Davies, B.E., 1990. Lead. In: Alloway, B.J. (ed.), *Heavy* Metals in Soils. Blackie and Son, London. pp 177-196
- Davies-Colley, R.J., Nelson, P.O. and Williamson, K.J., 1984. Copper and cadmium uptake by estuarine sedimentary phases. Environ. Sci. Technol., 18: 491-499
- Davis, J.A. and Leckie, J.O., 1978a. Surface ionization and complexation at the oxide/ water interface. II. Surface properties of amorpho us iron oxyhydroxide and adsorption of metal ions. J.Colloid Interface Sci., 67: 90-107

- Davis, J.A. and Leckie, J.O., 1978b. Effect of adsorbed complexing ligands on trace metal uptake by hydrous oxides. *Environ. Sci. Technol.*, 12: 1309-1315
- de Groot, A.J., Salomons, W. and Allersma, E., 1976. Processes affecting heavy metals in estuarine sediments. In: Burton, J.D. and Liss, P.S. (eds.), Estuarine Chemistry. Academic Press. New York. pp 131-158
- de Groot, A.J., Zschuppe, K.H. and Salomons, W., 1982. Standardisation of methods of analyzing for heavy metals in sediments. Hydrobiologia, 92: 689- 695
- Delfino, J.J. and Lee, G.F., 1971. Variation of manganese, dissolved oxygen and relater chemical parameters in the bottom waters of Lake Mendota, Wisconsin. Wat. Res., 5: 1207-1217
- Delfino, J.J. and Otto, R.G., 1986. Trace metal transport in two tributaries of the Upper Chesapeake Bay : the Susquehanna and Bush rivers. Mar. Chem., 20: 29-44
- de Luca Rebello, A., Haekel, W., Moreira, I., Santelli, R. and Schoeder, F., 1986. The fate of heavy meals in an estuarine tropical system. Mar. Chem., 18: 215-225
- de Mora, S.J. and Harrison, R.M., 1984. Physico chemical speciation of inorganic compounds in environmental media. In: Saxena, J.I. (ed.), Hazard Assessment of Chemicals: Current Developments. Vol. III, Academic Press, Orlando. pp 1- 61
- Dempsey, B.A. and Singer, P.C., 1980. The effects of calcium on the adsorption of zinc by Mn- oxides and Fe- oxides. In: Baker, R.A. (ed.), Contaminants and Sediments, Ann Arbor Sci. Publ., Ann Arbor. MI. pp 333-355
- Deurer, R., Forstner, U. and Schmoll, G., 1978. Selective chemical extraction of carbonate-associated trace metals in recent lacustrine sediments. *Geochim. Cosmochim. Acta*, 42: 425-427
- Dudka, S. and Chlopecka, A., 1990. Effect of solid-phase speciation on metal mobility and phytoavailability in sludge amended soil. *Water Air Soil Pollut.*, 51: 153-160
- Duinker, J.C., 1980. Partition of Fe, Mn, Al, K, Mg, Cu and Zn over particulate organic matter and mineral components and its dependence on total suspended matter concentration in the Southern Bight. *Thalassia Jugosl.*, 16: 249-250

- Duinker, J.C., 1983. Effects of particle size and density on the transport of metals to the oceans. In: Wong, C.S., Boyle, E., Bruland, K.W., Burton, J.D. and Goldberg, E.D. (eds.), Trace Metals in Sea Water. Plenum Press, New York. pp 209-226
- Duinker, J.C., 1985. Estuarine processes and riverborne pollutants. In: Nurnberg, H.W. (ed.), Pollutants and Their Ecotoxicological Significance. Wiley, New York. pp 227-238
- Duinker, J.C., 1986. Formation and transformation of element species in estuaries. In: Bernhard, M., Brinckman, F.E. and Sadler, P.J. (eds.), The Importance of Chemical "Speciation" in Environmental Processes. Springer- Verlag, Berlin. pp 365-384
- Duinker, J.C., Hillebrand, M.T.J. and Nolting, R.F.,1985. The River Ems: processes affecting the behaviour of metals and organochlorines during estuarine mixing. *Neth. J. Sea Res.*, 19: 19- 29
- Duinker, J.C., Hillebrand, M.T.J., Nolting, R.F. and Wellershaus, S. 1982. The river Elbe: processes affecting the behaviour of metals and organochlorines during estuarine mixing. Neth. J. Sea Res., 15: 141-169
- Duinker, J.C. and Nolting, R.F., 1982. Dissolved copper, zinc and cadmium in the southern bight of the North Sea. Mar. Pollut. Bull., 13: 93-96
- Duinker, J.C., Nolting, R.F, Michael, D., 1982. Effects of salinity, pH and redox conditions on the behaviour of Cd, Zn, Ni and Mn in the Scheldt estuary. Thalassia Jugosl., 18: 191-202
- Dyer, K.R., 1988. Fine sediment particle transport in estuaries. In: Dronkers, J. and Van Leussen, W. (eds.), *Physical processes in Estuaries.* Springer-Verlag, Berlin. pp 295-310
- Dyer, K.R., 1991. Circulation and mixing in stratified estuaries. Mar. Chem., 32: 111-120
- Dyrssen, D. and Wedborg, M., 1980. Major and minor elementschemical speciation in estuarine waters. In: Olausson, E. and Cato., I. (eds.), Chemistry and Biogeochemistry of Estuaries. J.Wiley, New York. pp 71-119
- Eaton, A., 1979a. Removal of soluble iron in the Potomac river estuary. *Estuar. Coast. Mar. Sci.*, 9: 41-49

Eaton, A., 1979b. The impact of anoxia on Mn fluxes in the Chesapeake Bay, Geochim. Cosmochim. Acta, 43: 429-432

Eaton, A., 1979c. Observations on the geochemistry of soluble copper, iron, nickel and zinc in the San Francisco Bay estuary. *Environ. Sci. Technol.*, 13: 425-431

- Ehrlich, H.L., 1980. Different forms of microbial manganese oxidation and reduction and their environmental significance. In: Trudinger, P.A., Walter, M.R. and Ralph, B.J. (eds.), Biogeochemistry of Ancient and Modern Environments. Springer-Verlag, Berlin. pp 327-332
- Elbaz-Poulichet, F., Huang, W.W., Jednacok-Biscan, J., Martin, J.M. and Thomas, A.J., 1982. Trace metal behaviour in the Gironde estuary: the problem revisited. *Thalassia Jugosl.*, 18: 61- 95
- Elbaz-Poulichet, F., Holliger, P., Wen Huang, W. and Martin, J.M., 1984. Lead cycling in estuaries illustrated by the Gironde Estuary, France. *Nature*, 308: 409-414
- Elbaz Poulichet, F., Guan, D.M. and Martin, J.M., 1991. Trace metal behaviour in a highly stratified Mediterranean estuary: The Krka (Yugoslavia). Mar. Chem., 32: 211 - 224
- Elbaz-Poulichet, F., Martin, J. M., Huang, W.W. and Zhu, J.X., 1987. Dissolved Cd bahaviour in some selected French and Chinese estuaries. Consequences on Cd supply to ocean. *Mar. Chem.*, 22: 125-136
- Elderfield, H., 1978. Chemical variability in estuaries. In: Bio-geochemistry of Estuarine Sediments. Unesco, Paris. pp 171-178
- Elderfied, H., Hepworth, A., Edwards, P.N. and Holliday, L.M., 1979. Zinc in the Conway river and estuary. *Estuar. Coast. Mar. Sci.*, 9: 203-208
- Elliot, H.A., Dempsey, B.A. and Maille, P.J., 1990. Content and fractionation of heavy metals in water treatment sludges. J. Environ. Qual., 19: 330-334
- Elliot, H.A., Liberati, M.R. and Huang, C.P., 1965. Competitive adsorption of heavy metals by soils. J. Environ. Qual., 15: 214-219
- Elsokkary, I.H. and Muller, G., 1990. Assessment and speciation of chromium, nickel, lead and cadmium in the sediments of the river Nile, Egypt. Sci. Total Environ., 97/98: 455-463

- El Wakeel, S.K. and Riley, J.P., 1957. The determination of organic carbon in marine muds. J. Council Intern. Power Explor. Mer. 22: 180-183
- Engler, R.M., Brannon, J.M., Rose, J. and Bigham, G., 1977. A practical selective extraction procedure for sediment characterisation. In: Yen, T.F. (ed.), *Chemistry of Marine Sediments*, Ann Arbor Sci. Publ. Inc. Michigan. pp 163-180
- Etcheber, H. and Bourg, A., 1983. Critical aspects of selective extractions of trace metals from estuarine suspended matter: Fe and Mn hydroxides and organic matter intercation. Proc. Int. Conf. Heavy Metals in the Environment. Heidelberg, CEP Consultants, Edinburgh. pp 1200-1203
- Evans, D.W., Cutshall, N.H., Cross, F.A. and Wolfe, D.A., 1977. Manganese cycling in the Newport river estuary, North Carolina. *Estuar. Coast. Mar. Sci.*, 5: 71-80
- Fairbridge, R.W., 1980. The estuary: its definition and geodynamic cycle. In: Olausson, E. and Cato, I. (eds.), Chemistry and Biogeochemistry of Estuaries. John Wiley and Sons, New York. pp 1-35
- Feely, R.A., Chester, A. J., Paulson, A.J. and Larrance, J.D., 1982. Relationship between organically bound Cu and Mn in settling particulate matter and biological processes in a subarctic estuary. *Estuaries*, 5: 74-80
- Feely, R.A., Massoth, G.J. and Lamb, M.F., 1983. The effect of sewage effluents on the flocculation of major and trace elements in a stratified estuary. In: Wong, C.S., Boyle, E., Bruland, K.W., Burton, J.D., and Goldberg, E.D. (eds.), *Trace Metals in Sea Water*. Plenum Press, New York. pp 227-244
- Feely, R.A., Massoth, G.J., Baker, E.T., Gendron, J.F., Paulson, A.J. and Crecelius, E.A., 1986. Seasonal and vertical variations in the elemental composition in the suspended and settling particulate matter in Puget sound, Washington. Estuar. Coast. Shelf Sci., 22: 215-239
- Feijtel, T.C., DeLaune, R.D. and Patrick, W.H.(Jr.), 1988. Biogeochemical control on metal distribution and accumulation in Louisiana sediments. J. Environ. Qual., 179: 88-94
- Flemming, C.A. and Trevors, J.T., 1989. Copper toxicity and chemistry in the environment: A review. Water Air Soil Pollut., 44: 143-158

- Fletcher, W.K., Holmes, G.S. and Lewis, A.G., 1983. Geochemistry and biological availability of iron and trace elements in the upper Fraser River estuary. *Mar. Chem.*, 12: 195-217
- Forstner, U., 1979a. Metal concentrations in river, lake and ocean waters. In: Forstner, U. and Wittmann, G.T.W. (eds.), Metal Pollution in the Aquatic Environment. Springer-Verlag, Berlin. pp 71-109
- Forstner, U., 1979b. Metal pollution assessment from sediment analysis. In: Forstner, U. and Wittmann, G.T.W. (eds.), Metal Pollution in the Aquatic Environment. Springer-Verlag, Berlin. pp 110-196
- Forstner, U. 1979c. Metal transfer between solid and aqueous phases. In: Forstner, U. and Wittmann, G.T.W. (eds.), Metal Pollution in the Aquatic Environment. Springer-Verlag, Berlin. pp 110-196
- Forstner, U., 1980. Cadmium in polluted sediments. In: Nriagu, J.O. (ed.), *Cadmium in the Environment*. Part I. John Wiley and Sons, New York, pp 305-364
- Forstner, U., 1982. Accumulative phases for heavy metals in limnic sediments. *Hydrobiologia*, 91: 269-284
- Forstner, U., 1985. Chemical forms and reactivities of metals in sediments. In: Leschber, R., Davis, R.D. and Hermite, P.L. (eds.), Chemical Methods for Assessing Bioavailable Metals in Sludges and Soils. Elsevier, London. pp 1- 31
- Forstner, U. and Patchineelam, S.R., 1976. Bindung und Mobilisation von Schwermetallen in fluviatilen Sedimenten. Chem. Z., 100: 49- 57 (cited in Forstner, 1979)
- Forstner, U., Schoer, J. and Knauth, H.D., 1990. Metal pollution in the tidal Elbe river. Sci. Total Environ., 97/98: 347- 348
- Forstner, U. and Stoffers, P., 1981. Chemical fractionation of transition elements in Pacific pelagic sediments. *Geochim. Cosmochim. Acta*, 45: 1141-1146
- Forstner, U. and Wittmann, G.T.W. (eds.), 1979. Metal Pollution in the Aquatic Environment. Springer-Verlag, Berlin. 486p.
- Fowler, S.W., 1990. Critical review of selected heavy metal and chlorinated Hydrocarbon concentrations in the marine environment. *Mar. Environ. Res.*, 29: 1-64

- Fukai, R. and Huynh-Ngoc, L., 1968. Studies on the chemical behaviour of radionuclides in seawater. 1 General considerations and study of precipitations of trace amounts of Cr, Mn, Fe, Co, Zn and Ce. In: Radio activity in the Sea. Int. Atomic Energy Agency, (pub. No.22) Vienna.
- Geering, H.R. and Hodgson, J.F., 1969. Soil Sci. Am. Proc. 30: 54-59 (cited in Kiekens, 1990)
- George, M.J. and Kartha, K.N.K., 1963. Surface salinity of Cochin backwaters with reference to tide. J. of Mar. Biol. Assoc. India, 5: 178-184
- Gibbs, R.J., 1973. Mechanisms of trace metals transport in rivers. Science, 130: 71-73
- Gibbs, R.J., 1977. Transport phases of transition metals in the Amazon and Yukon rivers. *Geol. Soc. Am. Bull.*, 88: 829-843
- Gibbs, R.J., 1986. Segregation of metals by coagulation in estuaries., Mar. Chem., 20: 149-159
- Gibson, J.J. and Farmer, J.G., 1986. Multistep sequential chemical extraction of heavy metals from urban soils. *Environ. Pollut.* (Ser.B), 11: 117-135
- Gilkes, R.J. and McKenzie, R.M. 1988. In: Graham, R.D., Hannam, R.J. and Uren, N.C. (eds.), Manganese in Soils and Plants. Kluwer, Dordrecht (cited in Smith, 1990)
- Goldberg, E.D. and Arrhenius, G.O.S., 1958. Chemistry of Pacific pelagic sediments. *Geochim. Cosmochim. Acta*, 13: 153-212
- Gole, C.V. and Tarapore, Z.S., 1966. Radioactive tracer studies at Cochin. I. I. O. E. Newsletter, 4: 28
- Goldwater, L., 1971. Mercury in the environment. Sci. Am. 224: 15-21
- Gopalakrishna Pillai, N.G., 1979. Microbenthos of a Tropical Estuary. Ph.D. Thesis, Cochin University, Cochin.
- Gopalan, U.K., Meenakshikunjamma, P.P. and Vengayil, D.T. 1987. Macrobenthos of Vembanad Estuary in relation to the deposition of degraded water fern Salvinia and other macrophytes. Proc. Natn. Semi. Estuarine Management, Trivandrum. pp 410- 418

- Gopalan, U.K., Vengayil, D.T, Udaya Varmma, P. and Krishnankutty, M., 1983. The shrinking backwaters of Kerala. J. Mar. Biol. Assoc. India, 25: 131-141
- Gopinathan, C.K. and Qasim, S.Z., 1971. Silting in navigational channels of the Cochin harbour area. J. Mar. Biol. Assoc. India, 13: 14-26
- Graham, W.F., Bender, M.L. and Klinkhammer, G.P., 1976. Manganese in Narrangansett Bay. Limnol. Oceanol., 21: 665-673
- Grieve, D.A. and Fletcher, W.K., 1976. Heavy metals in deltaic sediments of the Fraser river, British Columbia. *Can. J. Earth Sci.*, 13: 1683-1693
- Grieve, D.A. and Fletcher, W.K., 1977. Interactions between zinc and suspended sediments in the Fraser River Estuary, British Columbia. *Estuar. Coast. Mar. Sci.*, 5: 415-419
- Grigg, N.S., 1990. Estuarine water quality management: planning, organizing, monitoring and modelling. In: Michaelis, W. (ed.), Estuarine Water Quality Management. Springer-Verlag, Berlin. pp 3- 10
- Gunn, A.M., Hunt, D.T.E. and Winnard, D.A., 1989. The effect of heavy metal speciation in sediment on bioavailability to tubificid worms. *Hydrobiologia*, 188/189: 487 - 496
- Gupta, S.K. and Chen K.Y., 1975. Partitioning of trace metals in selective chemical fractions of nearshore sediments. *Environ. Lett.*, 10: 129-158
- Hakanson, L. and Jansson, M., 1983. Principles of Lake Sedimentology. Springer-Verlag, New York. 316p
- Haridas, P., Madhu Pratap, P. M. and Rao, T.S.S., 1973. Salinity, temperature, oxygen and zooplankton biomass of the backwaters from Cochin to Alleppey. Indian J. Mar. Sci., 2: 94-102
- Harrison, R.M., 1987. Physico-chemical speciation and chemical transformations of toxic metals in the environment. In Coughtrey, P.J., Martin, M.H. and Unsworth, M.H. (eds.), Pollutant Transport and Fate in Ecosystems. Blackwell Sci. Publ., Oxford. 239-247
- Harrison, R.M. and Wilson S.J., 1982. Physico-chemical speciation of trace metals in environmental samples. In: Albaiges, J. (ed.), Analytical Techniques in Environmental Chemistry Part 2. Pergamon Press, Oxford. pp 301-314

- Harrison, R.M., Laxen, D.P.H. and Wilson, S.J., 1981. Chemical associations of lead cadmium, copper and zinc in street dusts and roadside soils. *Environ. Sci. Technol.*, 15: 1378-1383
- Hart, B.T., 1982. Uptake of trace metals by sediments and suspended particulates: a review. Hydrobiologia, 91: 299-313
- Hart, B.T and Davies, S.H.R., 1981. Trace metal speciation in the freshwater and estuarine regions of the Yarra River, Victoria. Estuar. Coast. Shelf Sci., 12: 353-37
- Hawkes, H.E. and Webb, J.S., 1962. Geochemistry in Mineral Exploration. Harper and Row, New York. 415p
- Heath, G.R. and Dymond, J., 1977. Genesis and transformation of metalliferous sediments from the East Pacific Rise, Bauer Deep and Central Basin, Northwest Naszca plate. Geol. Soc. Am. Bull., 88: 723-733
- Helz,G.R. and Sinex, S.A., 1986. Influence of infrequent floods on the trace metal composition of estuarine sediments. Mar. Chem., 20: 1 - 11
- Hines, M.E., Lyons, W.B., Armstrong, P.B., Orem, W.H., Spencer, M.J., Gaudette, H.E. and Jones, G.E., 1984. Seasonal metal remobilisation in the sediments of Great Bay, New Hamphire . Mar. Chem., 15: 173-187
- Hirose, K., 1990. Chemical speciation of trace metals in seawater - implication of particulate trace metals. Mar. Chem., 28: 267-274
- Hodgson, J.F., Lindsay, W.I. and Trierweiler, J.F., 1966. Micronutrient cation complexing in soil solution: II. Complexing of Zn and Cu in displaced solution from calcareous soils. Soil Sci. Soc. Am. Proc., 30: 723-730
- Hohl, H. and Stumm, W., 1976. Interaction of Pb with hydrous Al- oxide. J. Colloid Interface Sci., 55: 281-288
- Holliday, L. M. and Liss, P.S., 1976. The behaviour of dissolved iron, manganese and zinc in the Beaulieu estuary, S. England. Estuar. Coast. Mar. Sci., 4: 349-353
- Holmes, C.W., 1986. Trace metal seasonal variations in Texas marine sediments. *Mar. Chem.*, 20: 13-27
- Holmgren, G.S., 1967. A rapid citrate- dithionite extractable iron procedure. Soil Sci. Soc. Am. Proc., 31: 210-211

- Hong Y.T. and Forstner, U., 1983. Speciation of heavy metals in yellow river sediments. In: Proc. Int. Conf. Heavy metals in the Environment, CEP Consultants, Edinburgh. pp 872-875
- Hong, G.H., Park, Y.A. and Lee, K.M., 1983. Partitioning of heavy metals in sediments from Jinhae Bay Korea. J. Oceanol. Soc. Korea, 18: 180 - 184
- Hoshika, A., Shiozawa, T., Kawana, K. and Tanimoto, T., 1991. Heavy metal pollution in sediment from the Seto Inland Sea, Japan. Mar. Pollut. Bull., 23: 101-105
- Hungspreugs, M., Dharmvanij, S., Utoomprukporn, W. and Windom, H.L., 1990. A Comparative study of the trace metal fluxes of the Bang Pakong and the Mae Klong rivers, Thailand . Sci. Total Environ., 97/98: 89-102
- Hunter, K.A., 1983. On the estuarine mixing of dissolved substances in relation to colloid stability and surface properties. *Geochim. Cosmochim. Acta*, 47: 467-475
- Huynh-Ngoc, L., Whitehead, N.E. and Oregioni, B., 1988. Cadmium in the Rhone River. Wat. Res., 22: 571 - 576
- Irving, H., Williams, R.J.P., 1948. Order of stability of metal complexes. Nature, 162: 746- 747
- Jackson, M.L., 1958. Soil Chemical analysis. Prentice-Hall Englewood Cliffs, N.J. 495p.
- James, R. O. and Healy, T.W., 1972. Adsorption of hydrolyzable metal ions at the oxide - water interface. J. Colloid Interface Sci. 40: 65-81
- Jayalakshmy, K.V. and Sankaranarayanan V.N., 1983. Auto correlograms and auto regressive models of trace metal distribution in Cochin backwaters. Indian. J. Mar. Sci., 12: 236-238
- Jayapalan, S.P., Sasidharan, K.M. and Achuthan Nair, K.S., 1976. Some aspects of physico chemical and biological variations of Periyar due to the effluent discharges from FACT. Bull. Dept. Fisheries, Kerala, 1: 47-59
- Jenne, E.A., 1968. Controls on Mn, Fe, Co, Ni, Cu and Zn concentrations in soils and water: the significant role of hydrous Mn- and Fe- oxides . In: Baker, R.A. (ed.), Trace Inorganics in Water. Adv. Chem. Ser.73, Am. Chem. Soc., pp 337-387

- Jenne, E.A., Baccini, P., Bauld, J., Brummer, G.W., Chau, Y.K. Fremmel, F.W., Gamble, D.S., Kabata-Pendias, A., Kane, P.F., Leckie, J.O., Muller, G.K., Prosi, F.H. and Tobschall, H.J., 1986. Chemical species in fresh water and terrestrial systems. In: Bernhard, M., Brinckman, F.E. and Sadler, P.J. (eds.), The importance of chemical speciation in Environmental Processes. Springer-Verlag, Berlin. pp 121-147
- Jenne,E.A., Kennedy, V.C., Burchard, J.M. and Ball, J.W., 1980. Sediment collection and processing for selective extraction and for total trace metal analysis. In Baker, R.A. (ed.), Contaminants and Sediments, Vol.2, Ann Arbor Sci.Publ., Michigan, pp 169-193
- Jenne, E.A. and Luoma, S.N., 1977. Forms of trace elements in soils, sediments and associated waters: an over view of their determination and biological availability. In: Wildung, R. and Drucker, H. (eds.) Biological Implications of Metals in the Environment. ERDA Symp. Ser.42, Energy Research and Development Administration, Oak Ridge, TN. pp 110-143
- Josanto, V., 1971. The bottom salinity characteristics and the factors that influence the salt water penetration in the Vembanad lake. *Bull. Dept. Mar. Biol. Oceanogr.* Univ. Cochin, 5: 1-16
- Joseph, K.J.. 1989. Studies on some Aspects of Phytoplankton. Ph.D. Thesis . Cochin University of Science and Technology, Cochin.
- Joseph, K.J. and Pillai, V.K., 1975. Seasonal and spatial distribution of phytoplankters in Cochin backwaters. Bull. Dept. Mar. Sci. Univ. Cochin, 7: 171-180
- Joseph, K.J. and Nair, P.V.R., 1975. Growth characteristics of certain estuarine phytoplankters. Bull. Dept. Mar. Sci. Univ. Cochin, 7: 151-159
- Joseph, K.J., Lakshmanan, P.T., Shynamma, C.S. and Nambisan, P.N.K., 1988. Studies on phytoplankton in polluted waters. J. Mar. Biol. Assoc. India, 26: 42-46
- Jouanneau, J.M., Boutier, B., Chiffoleau, J.F., Latouche, C and Philipps, I., 1990. Cadmium in the Gironde Fluvioestuarine systems: behaviour and flow. Sci. Total Environ., 97/98: 465-479

- Jouanneau, J.M., Etcheber, H. and Latouche, C., 1983. Imporishment and decrease of metallic elements associated with suspended matter in the Gironde estuary. In: Wong, C.S., Boyle, E., Bruland, K.W., Burton, J.D. and Goldberg, E.D. (eds.), Trace Metals in Sea Water, Plenum Press, New York. pp 245-263
- Joy, C.M., Balakrishnan, K.P. and Ammini Joseph, 1990. Effect of industrial discharges on the ecology of phytoplankton production in the river Periyar (India). Wat. Res., 24: 787-796
- Juo, A.S.R., 1981. Chemical characteristics. In: Greenland, D.J. (ed.), Characterisation of Soils. Clarendon Press, Oxford. pp 51- 79
- Kamat, S.B. and Sankaranarayanan, V.N., 1975. Distribution of iron in estuarine and nearshore waters of Goa. Indian J. Mar. Sci., 4: 30-33
- Kappenberg, J., Fanger, H.U., Manning, V and Prange, A. 1990. Suspended matter and heavy metal transport in the lower Elbe river under different flow conditions. In: Michaelis, W. (ed.), Estuarine Water Quality Management. Springer-Verlag, Berlin. pp 147-152
- Keeney-Kennicutt, W.L. and Presley, B.J., 1986. The geochemistry of trace metals in the Brazos River Estuary. *Estuar. Coast. Shelf Sci.*, 22: 459-477
- Kester, D.R. (rptr.), 1986. Chemical species in marine and estuarine systems- group report. In: Bernhard, M., Brinckman F.E. and Sadler, P.J. (eds.), The Importance of Chemical Speciation in Environmental Processes. Springer-Verlag, Berlin. pp 275 - 299
- Kersten, M. and Forstner, U., 1987. Effect of sample pretreatment on the reliability of solid speciation data of heavy metals - implications for the study of early diagenetic processes. Mar. Chem., 22: 299-312
- Khalid, R.A., 1980. Chemical mobility of cadmium in sediment-water systems. In: Nriagu, J.O. (ed.), Cadmium in the Environment. John Wiley Sons, New York. pp 257-304
- Kheboian, C. and Bauer, C.F., 1987. Accuracy of selective extraction procedures for metal speciation in model aquatic sediments. Anal. Chem., 59: 1417-1423
- Kiekens, L., (1990). Zinc. In: Alloway, J.B.(ed.), Heavy Metals in Soils. Blackie and Sons, London. pp 261- 279

King, L.D., 1988. Retention of metals by several soils of the Southeastern United States. J. Environ. Qual., 17: 239-246

Knudtsen, K. and O`Connor, G.A., 1987. Characterization of iron and zinc in Albuquerque sewage sludge. J. Environ. Qual., 16: 85- 90

Koske, P.H., 1990. Suspended matter in the outer Elbe estuary and in the Wadden Sea: distribution and transport. In: Michaelis, W. (ed.), Estuarine Water Quality Management. Springer-Verlag, Berlin. pp 158-164

Krauskopf, K.B., 1967. Introduction to Geochemistry. McGraw-Hill, New York. 721p.

Krishnakumar, P.K., 1987. Physiological effects of some heavy metals on Perna viridis (Linnaeus). Ph.D. Thesis. Cochin University of Science and Technology, Cochin.

Krishnakumar, P.K., Damodaran, R. and Nambisan, P.N.K., 1990. Accumulation, distribution and depuration of mercury in the green mussel Perna viridis (Linnaeus). Proc. Indian Acad. Sci. (Anim. Sci.), 99: 345-352

Krumbein, W.C. and Pettijohn, F.J., 1938. Manual of Sedimentary Petrography. Appleton Century Crofts, Inc. New York. 549p.

Krumgalz, B. S. and Fainshtein, G., 1989. Trace metal contents in certified reference sediments determined by nitric acid digestion and atomic spectrometry. Anal. Chim. Acta, 218: 335-340

Kumar, U.V. and Roonwal, G.S., 1987. Geochemical characteristics and selective leaching of transition elements in sediments from the Central Pacific Ocean. Indian. J. Mar. Sci., 16: 77-85

Kurian, C.V., Damodaran, R. and Antony, A. 1975. Bottom fauna of the Vembanad lake. Bull. Dept. Mar. Sci. Univ. Cochin, 7:987-994

Kuwabara, J.S., Chang, C.C.Y., Cloern, J.E., Fries, T.L., Davis, J.A. and Luoma, S.N. 1989. Trace metál associations in water column of south San Francisco Bay, California. *Estuar. Coast.Shelf Sci.*, 28: 307-325

Lacerda, L.D., De Paula, F.C.F., Ovalle, A.R.C., Pfeiffer, W.C. and Malm, O., 1990. Trace metals in fluvial sediments of the Madeira river watershed, Amazon, Brazil. Sci. Total Environ., 97/98: 525- 530

- Lacerda, L.D., Pfeiffer, W.C. and Fiszman, M., 1987. Heavy metal distribution, availability and fate in Sepetiba bay, S.E. Brazil. Sci. Total Environ., 65: 163-173
- Lake, D.L., Kirk, P.W.W. and Lester, J.N., 1984. Fractionation, characterisation and speciation of heavy metals in sewage sludge and sludge amended soil: a review. J. Environ. Qual., 13: 175-183
- Lake, D.L, Kirk, P.W.W. and Lester, J.N., 1989. Heavy metal solids association in sewage sludges. Wat. Res., 23: 285-291
- Lakshmanan, P.T., 1982. Investigations on the chemical constituents and trace metal interactions in some bivalve molluscs of the Cochin backwaters. Ph.D. Thesis. Cochin University, Cochin.
- Lakshmanan, P.T. and Nambisan, P.N.K., 1977. Toxicity of Copper on the bivalve Villorita cyprinoides var. cochinensis. Indian. J. Mar. Sci., 6: 83-85
- Lakshmanan, P.T. and Nambisan, P.N.K., 1979. Accumulation of mercury by the mussel Perna viridis (L.). Curr. Sci., 48: 672-674
- Lakshmanan, P.T. and Nambisan, P.N.K., 1980. Biochemical composition of the bivalve molluscs Villorita cyprinoides var. cochinensis (Hanley) and Meretrix casta (Chemnitz). Indian J. Mar. Sci., 9: 65-67
- Lakshmanan, P.T. and Nambisan, P.N.K., 1983. Seasonal variations in trace metal content in bivalve molluscs Villorita cyprinoides var. cochinensis (Hanley), Meretrix casta (Chemnitz) and Perna viridis (Linnaeus). Indian J. Mar. Sci., 12: 100 - 103
- Lakshmanan, P.T. and Nambisan, P.N.K., 1985a. Tissue lactic acid and glycogen level of molluscs exposed to Cu and Hg. *Curr. Sci.*, 54: 478-479
- Lakshmanan, P.T. and Nambisan, P.N.K., 1985b. Uptake and loss of mercury in three bivalve molluscs, Perna viridis (Linnaeus), Villorita cyprinoides var. cochinensis and Meretrix casta (Chemnitz). In: Harvest and Post harvest Technology of Fish. Soc. Fisher. Technologists (India), Matsyapuri, Cochin. pp 419- 423

- Lakshmanan, P.T. and Nambisan, P.N.K. (1989). Bioaccumulation and depuration of some trace metals in the mussel *Perna viridis* (L.). *Bull. Environ. Contam. Toxicol.*, 43: 131 -138
- Lakshmanan, P.T., Shynamma C.S., Balchand, A.N., Kurup, P.G. and Nambisan, P.N.K., 1982. Distribution and seasonal variation of temperature and salinity in Cochin backwaters. Indian. J. Mar. Sci., 11: 170-172
- Lakshmanan, P.T., Shynamma, C.S., Balchand, A.N. and Nambisan, P.N.K., 1987. Distribution and variability of nutrients in Cochin backwaters, south west coast of India. *Indian J. Mar. Sci.*, 16: 99-102
- Langston, W.J. and Bryan, G.W., 1984. The relationship between metal speciation in the environment and bioaccumulation in aquatic organisms. In: Cramer, C.J.M. and Duinker, J.C.(eds.), Complexation of Trace Metals in Natural Waters. Martin Nijhoff/ Dr. W.Junk Publishers. pp 375-392
- Latha Thampuran., 1987. Physiological Effects of Copper (II) on Sunetta scripta. Ph.D. Thesis. Cochin University.
- Lawson, P.S., Sterritt, R.M. and Lester, J.N., 1984. The speciation of metals in sewage and activated sludge effluent. *Water Air Soil Pollut.*, 21: 387-402
- Lee, C.B. 1985. Sedimentary processes of fine sediments and the behaviour of associated metals in the Keum Estuary, Korea. In: Singleo, A.C. and Hattori, A. (eds.), Marine and Estuarine Geochemistry. Lewis Publishers Inc. Michigan. pp 209-225
- Leita, L. and de Nobili. M., 1991. Heavy metals in the environment: water soluble fractions of heavy metals during composting of municipal solid wastes. J. Environ. Qual., 20: 73- 78
- Le Mare, P.H., 1981. Phosphorus sorption and release. In: Greenland, D.J. (ed.), *Characterisation of Soils*. Clarendon Press, Oxford. pp 95-134
- Li, Y.H.L., Burkhardt, L. and Teraoka, H., 1984. Desorption and coagulation of trace elements during estuarine mixing. *Geochim. Cosmochim. Acta*, 48: 1879-1884
- Liang, J., Stewart, J.W.B., and Karamanos, R.E., 1990. Distribution of zinc fractions in prairie soils. Can. J. Soil Sci., 70: 335- 342

- Lion, L.W., Altmann, R.S. and Leckie, J.O., 1982. Trace metal adsorption characteristics of estuarine particulate matter: evaluation of contribution of Fe/Mn oxide and organic surface coatings. *Environ. Sci. Technol.*, 16: 660-666
- Lithnor, G. 1975. Pretreatment of samples (organic matter, sediments and water) for subsequent determination of heavy metals by atomic absorption spectroscopy. In: Manual of Methods in Aquatic Environmental Research. Part I. Methods for Detection, Measurement and Monitoring of Water Pollution, FAO, Rome. pp 41-46
- Loring, D.H., 1976. Distribution and partition of cobalt, nickel chromium and vanadium in the sediments of the Saguenay Fjord. Can. J. Earth Sci., 13: 1706-1718
- Lu, J.C.S. and Chen, K.Y., 1977. Migration of chemical constituents in sediment seawater interfaces. In: Yen, T.F. (ed.), *Chemistry of Marine Sediments*. Ann Arbor Sci. Publ. Inc. Michigan. pp 181-221
- Lum, K.R., Betteridge, J.S. and MacDonald, R.R., 1982. The potential availability of P, Al, Cd, Co, Cr, Cu, Fe, Mn, Pb, and Zn in urban particulate matter. Environ. Technol. Lett., 3: 57-62
- Luoma, S.N., 1983. Bioavailability of trace metals to aquatic organisms: a review. Sci. Total Environ., 28: 1-22
- Luoma, S.N., 1986. A comparison of two methods for determining copper partitioning in oxidised sediments. Mar. Chem., 20: 45-59
- Luoma, S.N., 1989. Can we determine the biological availability of sediment bound trace elements?, Hydrobiologia, 176/177: 379-396
- Luoma, S.N., 1990. Processes affecting metal concentrations in estuarine and coastal marine sediments In: Rainbow, P. and Furness, R. (eds.), Heavy Metals in the Marine Environment. CRC Press, Uniscience. pp 51-66
- Luoma, S.N. and Bryan, G.W., 1978. Factors controlling the availability of sediment bound lead to the estuarine bivalve Scrobicularia plana. J. Mar. Biol. Ass. U. K., 58: 793-802
- Luoma, S.N. and Bryan, G.W., 1979. Trace metal bioavailability: modeling chemical and biological interactions in sediment bound zinc. In: Jenne, E.A. (ed.), Chemical Modeling in Aqueous Systems. ACS Symp. Ser.93. pp 577-604
- Luoma, S.N. and Bryan, S.W., 1981. A statistical assessment of the form of trace metals in oxidised estuarine sediments employing chemical extractants. *Sci. Total Environ.*, 17: 165-196
- Luoma, N. and Bryan, G.W., 1982. A statistical study of environmental factors controlling concentrations of heavy metal in the burrowing bivalve Scrobicularia plana and the polichaete Nereis diversicolor. Estuar. Coast. Shelf Sci., 15: 95-108
- Luoma, S.N. and Davis, J.A., 1983. Requirements for modeling trace metal partitioning in oxidised estuarine sediments. Mar. Chem., 12: 159-181
- Luoma, S.N. and Jenne, E.A., 1976. Estimating bioavailability of sediment-bound metals with chemical extractants. In: Hemphill, D.D. (ed.), *Trace Substances in Environmental Health*, X. Univ. Missouri Press, Missouri. pp 343-353
- Luoma, S.N. and Phillips, D.J.H., 1988. Distribution, variability and impacts of trace elements in San Francisco Bay. *Mar. Pollut. Bull.*, 19: 413-425
- Madhusoodana Kurup, B., 1983. Studies on the Systematics and Biology of the Fishes of the Vembanad Lake. Ph.D. Thesis. Cochin University.
- Mahajan, B.A., Srinivasan, M. and Mahapatra, S.P., 1987. Anthropogenic influence on the metal pollution in the estuarine region of Bombay island. Proc. Natn. Sem. Estuarine Management, Trivandrum. pp 163-167
- Makar V.M., Ganapathy, S. and Pillai, K.C., 1981. Distribution of Zn, Cu, Mn, and Fe in Bombay Harbour Bay. Indian J. Mar. Sci., 10: 35-40
- Mallik, T.K., and Suchindan G.K., 1984. Some sedimentological aspects of Vembanad lake, Kerala, West Coast of India. Indian. J. Mar. Sci., 13: 159-163
- Malo, B.A., 1977. Partial extraction of metals from aquatic sediments. Environ. Sci.Technol., 11: 277-282

- Manikoth, S. and Salih, K.Y.M., 1974. Distribution characteristics of nutrients in the estuarine complex of Cochin. Indian J. Mar. Sci., 3: 125-130
- Martin, J.M., Nirel, P. and Thomas A.J., 1987. Sequential extraction techniques : promises and problems. Mar. Chem., 22: 313- 341
- Martin, J.M. and Meybeck, M., 1979. Elemental mass balance of material carried by major world rivers. Mar. Chem., 7: 173-206
- Martin, J.M. and Whitfield, M., 1983. The significance of the river input of chemical elements to the ocean. In: Wong, C.S., Boyle, E., Bruland, K.W., Burton, J.D. and Goldberg, E.D. (eds.), Trace Metals in Sea Water. Plenum Press, New York. 265-296
- Mayer, L.M., 1982. Aggregation of colloidal iron during estuarine mixing: kinetics, mechanism and seasonality. Geochim. Cosmochim. Acta, 46 2527-2535
- McCaffrey, R.J., Myers, A.C., Bender, M., Luedtke, N., Cullen, D., Froelich, P. and Klinkhammer, G., 1981. The relation between pore water chemistry and benthic fluxes of nutrients and manganese in Narragansett bay, Rhode island. *Limnol. Oceanogr.*, 25: 31-44
- McGrath, S.P. and Smith, S., 1990. Chromium and nickel. In: Alloway, B.J. (ed.), *Heavy Metals in Soils*. Blackie and Son, London. pp 125-150
- Menon, N.R., 1986. Combined toxicity of silver and copper on the sublethal response of Perna viridis. Proc. Natn. Sem. on Mussel Watch. University of Cochin. pp 13-14
- Michaelis, W., 1990. The bilex concept- research supporting public water quality surveillance in estuaries. In: Michaelis, W. (ed.), Estuarine water Quality Management, Springer-Verlag. Berlin. pp 79- 88
- Miller, W.P., Martens, D.C., and Zelazny, L.N., 1986. Effect of sequence in extraction of trace metals from soils. Soil. Sci. Soc. Am. J., 50: 598-601
- Mitchell, R.L., 1964. Trace elements in soil. In Bear, F.E. (ed.), Chemistry of the Soil . Reinhold, pp 320-368
- Morgan, J.J. and Stumm, W., 1964. Colloid-chemical properties of manganese dioxide. J. Coll. Sci., 19: 347-353

Morris, A.W., 1990. Kinetic and equilibrium approaches to estuarine chemistry. Sci. Total. Environ., 97/98: 253-266

Morris, A.W., 1986. Removal of trace metals in the very low salinity regions of the Tamar estuary, England. Sci. Total. Environ., 49: 297-304

Morris, A.W. and Bale, A.J., 1979. Effects of rapid precipitation of dissolved Mn in river water on estuarine Mn distributions. *Nature*, 279: 318-319

Morris, A.W., Bale, A.J. and Howland, R.J.M., 1982. Chemical variability in the Tamar Estuary, Southwest England. *Estuar. Coast. Shelf Sci.*, 14: 649-661

Morris, A.W., Bale, A.J., Howland, R.J.M, Millward, G.E., Ackroyd, D.R., Loring, D.H., Rantala R.T., 1986. Sediment mobility and its contribution to trace metal cycling and retention in a macrotidal estuary. *Wat. Sci. Technol.*, 18: 111-119

Murray, J.W., 1975a. The interaction of metal ions at the manganese dioxide- solution interface. *Geochim. Cosmochim.* Acta, 39: 505-519

Murray, J.W., 1975b. The interaction of cobalt with hydrous manganese dioxide. *Geochim. Cosmochim. Acta.* 39: 635-647.

Murray, J.W. and Gill, G., 1978. The geochemistry of iron in Puget Sound. Geochim. Cosmochim. Acta, 42: 9-19

Murty, P.S.N., Ch. M. Rao and Reddy, C.V.G., 1973. Partition patterns of iron, manganese, nickel and cobalt in the shelf sediments of west coast of India. *Indian J. Mar. Sci.*, 2: 6-12

Murty, P.S.N., Rao, M., Veerayya, M. and Reddy, C.V.G., 1977. Partition patterns of Al, Fe, Mn, Ni, Co and Cu in the sediments of the deep sea drilling project, site 219. Indian. J. Mar. Sci., 6: 64-71

Murty, P.S.N., Rao, M., Paropkari, A.L. and Topgi, R.S., 1980. Distribution of Mn, Ni, Co and Cu in non-lithogenous fractions of sediments of western continental India. *Indian* J. Mar. Sci., 9: 56-61

Murty, P.S.N., and Veerayya, M., 1972. Studies on the sediments of Vembanad lake, Kerala State. Part II Distribution of phosphorus. Indian. J. Mar. Sci., 1: 106-115

- Murty, P.S.N. and Veerayya, M., 1981. Studies on the sediments of Vembanad lake, Kerala state. Part IV Distribution of trace elements. *Indian J. Mar. Sci.*, 10: 165-172
- Nair, C.K., Balchand, A.N. and Nambisan, P.N.K., 1991. Heavy metal speciation in sediments of Cochin estuary determined using chemical extraction techniques. Sci. Total Environ., 102: 113-128
- Nair, N.B., Abdul Aziz, P.K., Suryanarayanan, H., Arunachalam
  M., Krishnakumar, K. and Tresa, V.F., 1987. Distribution of heavy metals in the sediments of the Ashtamudi estuary, S.W. Coast of India. In: Contributions in Marine Sciences. Dr. S.Z Qasim Sastyabdapurti Felicitation Volume, National Institute of Oceanography, Dona Paula, India. pp 269-289
- Nair, P.V.R., Joseph K.J., Balachandran, V.K. and Pillai, V.K., 1975. A study on the primary production in the Vembanad Lake. Bull. Dept. Mar. Sci. Univ. Cochin, 7: 161-170
- Nair, S.M., Balchand, A.N. and Nambisan, P.N.K., 1989. On the determination and distribution of hydroxylated aromatic compounds in estuarine waters. *Toxicol. Environ. Chem.*, 22: 203-213
- Nair, S.M., Balchand, A.N. and Nambisan, P.N.K., 1990. Metal concentrations in recently deposited sediments of Cochin backwaters, India. Sci. Total Environ., 97/98: 507-524
- Nambisan, P.N.K., Lakshmanan, P.T. and Salih, K.Y.M., 1977. On the uptake of copper (II) by Meretrix casta (Chemnitz), an indicator species of metal pollution. *Curr. Sci.*, 46: 437-440
- Narayana Pillai, V., Cherian, T., George, M.D. and Anto, A.F., 1973. The effects of tidal currents on the hydrography of the back-waters around Cochin barmouth during post and pre-monsoon seasons. Proc. Indian Geophys. Union, 10: 231-239
- Nelson, S.K. and Lund, W., 1982. The determination of weakly and strongly bound copper, lead and cadmium in Oslo fjord samples. Mar. Chem., 11: 223-233
- Nichols, F.H., Cloern, J.E., Luoma, S.N. and Peterson, D.H., 1986. The modification of an estuary. *Science*, 231: 567-573
- Nirel, P.M.V. and Morel, F.M.M., 1990. Pitfalls of sequential extractions. *Wat. Res.*, 24: 1055-1056

- Nissenbaum, A., 1972. Distribution of several metals in chemical fractions of sediment core from the Sea of Okhotsk. Isr. J. Earth. Sci., 21: 143-154
- Nolting, R.F., Sundby, B. and Duinker, J.C., 1990. Behaviour of minor and major elements in suspended matter in the Rhine and Meuse rivers and Estuary. *Sci. Total Environ.*, 97/98: 169-183
- Noriki, S., Ishimori, N., Harada, K and Tsunogai, S., 1985. Removal of trace metals from sea water during a phytoplankton bloom as studied with sediment traps in Funka Bay, Japan. *Mar. Chem.*, 17: 75-89
- Nriagu, J. O. (ed.), 1980. Nickel in the Environment. John Wiley, New York. 833p.
- Nriagu, J. O. (ed.), 1980. Zinc in the Environment. Part 1. Ecological Cycling. John Wiley, New York. 453p
- Nriagu, J. O. (ed.), 1980. Cadmium in the Environment. Part 1. Ecological Cycling. John Wiley, New York. 693p
- Nriagu, J.O. and Pacyna, J.M., 1988. Quantitative assessment of world-wide contamination of air, water and soils by trace metals. *Nature*, 333: 134-139
- Nyffeler, U.P., Li, Y.H. and Santschi, P.H., 1984. A Kinetic approach to describe trace element distribution between particulates and solution in natural systems. *Geochim. Cosmochim. Acta*, 48: 1513-1522
- Oakley, S.M., Nelson, P.O. and Williamson K.J., 1981. Model of trace metal partitioning in marine sediments. *Environ. Sci. Technol.*, 15: 474-480
- Olsen, C.R., Cutshall, N.H.and Larsen, I.L., 1982. Pollutant particle associations and dynamics in coastal marine environments: a review. Mar. Chem., 11: 501-533
- Ouseph, P.P., 1987. Heavy metal pollution in the sediments of Cochin estuarine system. Proc. Natn. Sem. Estuarine Management, Trivandrum. pp 123-127
- Ouseph, P.P. 1990. Dissolved, particulate and sedimentary mercury in the Cochin estuary, south west coast of India. In: Michaelis, W. (ed.) Estuarine Water Quality Management, Springer-Verlag, Berlin. pp 461-465

- Pardo, R., Barrado, E., Perez, L. and Vega, M., 1990. Determination and speciation of heavy metals in sediments of the Pisuerga river. Wat. Res., 24: 373-379
- Patchineelam, S.R., 1975. Untersuchungen uber die Hauptbindung sarten und die Mobilisierbarkeit von Schwermetallen in fluviatilen Sedimenten. Diss. Univ. Heidelberg, 136p. (cited in Forstner and Wittmann, 1979)
- Patchineelam and Forstner, U., 1983. Sequential chemical extractions on polluted sediments from Subae river, Brazil. In: Proc. Symp. Heavy metals in the Environment, Heidelberg, CEP Consultants, Edinburgh. pp 860-863.
- Patel, B., Vasanti, S.B., Shakunt Patel and Balani, M.C.,1985. Heavy metals in Bombay Harbour area. Mar. Pollut. Bull., 16: 22-28
- Paul, A.C. and Pillai, K.C., 1983a. Trace metals in a tropical river environment: distribution. Water Air Soil Pollut., 19: 63-73
- Paul, A.C. and Pillai, K.C., 1983b. Trace metals in a tropical river environment: speciation and biological transfer. Water Air Soil Pollut., 19: 75-86
- Paulson, A.J., Carl, H.C.(Jr.) and Cockelet, E.D., 1991. Remobilization of Cu from marine particulate organic matter and from sewage. Mar. Chem., 33: 41- 60
- Paulson, A.J., Feely, R.A., Carl, H.C.(Jr.) and Gendron, J.F., 1984. Behaviour of Fe, Mn, Cu and Cd in the Duwamish River estuary down stream of a sewage treatment plant. Wat. Res., 18: 633-641
- Pederson, T.F., Waters, R.D. and McDonald, R.W., 1989. On the natural enrichment of cadmium and molybdenum in the sediments of Ucluelet inlet, British Columbia. Sci. Total Environ., 79: 125-139
- Pillai, J.S., 1989. Studies on siltation in Cochin Harbour -Dynamics of suspensate. Ph.D. Thesis. Cochin University of Science and Technology, Cochin.
- Ponnamperuma, F. N., 1972. The Chemistry of submerged soils. Advances in Agronomy, 24: 29-96
- Ponnamperuma, F.N., 1984. In: Kozlowski, T.T.(ed.), Flooding and Plant growth. Academic Press, Orlando. (cited in Smith, 1990)

- Prabhudeva, K.N., 1988. Toxicity, accumulation and depuration of heavy metals in the brown mussel, Perna indica. Ph.D. Thesis. Cochin University of Science and Technology, Cochin.
- Prabhudeva, K.N. and Menon, N.R., 1987. Toxic effects of zinc-oil mixtures on Perna viridis (L.). Indian J. Mar. Sci. 16: 199-200
- Prasada Reddy, P.R., 1986. Studies on some physico-chemical parameters of harbour and coastal waters of Visakhapatnam (Bay of Bengal). Ph.D. Thesis. Andhra Univesity.
- Presley, B.J., Kolodny, Y., Nissenbaum, A. and Kaplan, I.R., 1972. Early diagenesis in a reducing fjord, Saanich Inlet, British Columbia. II. Trace element distribution in interstitial water and sediment. Geochim. Cosmochim. Acta, 36: 1073-1099
- Pritchard, D.W., 1967. What is an estuary? Physical view point. In: Lauff, G.H. (ed.), Estuaries. Am. Assoc. Adv. Sci., Washington. pp 3-5
- Prohic, E. and Kniewald, G., 1987. Heavy metal distribution in recent sediments of the Krka river estuary - example of sequential extraction analysis. *Mar. Chem.*, 22: 279-297
- Prosi, F., 1989. Factors controlling biological availability and toxic effects of lead in aquatic organisms. Sci. Total Environ., 79: 157-169
- Pylee, A., Udaya Varma, P. and Revichandran N.C., 1990. Some aspects of circulation and mixing in lower reaches of Periyar estuary, West coast of India. *Indian J. Mar. Sci.*, 19: 32-35
- Qasim, S.Z. and Gopinathan, C.K., 1969. Tidal cycle and the environmental features of Cochin backwaters (a tropical estuary). Proc. Indian Acad. Sci., 69: 338-348
- Qasim, S.Z. and Sankaranarayanan, V.N., 1972. Organic detritus of a tropical estuary. *Mar. Biol.*, 15: 193-199
- Qasim, S.Z. and Sen Gupta, R., 1988. Some problems of coastal pollution in India. Mar. Pollut. Bull., 19: 100-106
- Rama Raju, V.S., Udaya Varma, P. and Ahraham Pylee, 1979. Hydrographic characteristics and tidal prism at the Cochin harbour mouth. Indian J. Mar. Sci., 8: 78-84

- Ramesh, R. Subramanian, V. and Van Grieken, R., 1990. Heavy metal distribution in sediments of Krishna River Basin, India. Environ. Geol. Water Sci., 15: 207-216
- Rao, M., Rajamanickam, G.V., Murty, P.S.N., and Reddy, C.V.G., 1976. Partition patterns of iron in the recent sediments of the northern half of the western continental shelf of India. Indian J. Mar. Sci., 5: 46-50
- Rao, M., Veerayya, M., Murty, P.S.N., Paropkari, A.L. and Reddy, C.V.G., 1978. Partitioning studies in the sediments of deep sea drilling project site 220 in the southeastern Arabian sea. Indian J. Mar. Sci., 7: 254-257
- Rapin, F., 1983. Speciation of heavy metals in a sediment core from Baie de Nice (Mediterranean Sea). Proc. Symp. Heavy Metals in the Environment, Heidelberg. CEP Consultants, Edinburgh. pp 1005-1008
- Rapin, F., Tessier, A., Campbell, P.G.C. and Carigman, R., 1986. Potential artifacts in the determination of metal partitioning in sediments by a sequential extraction procedure. *Environ. Sci. Technol.*, 20: 836-839
- Remani, K.N., Sarala Devi, K., Venugopal, P. and Unnithan R.V., 1983. Indicator organisms of pollution in Cochin backwaters of pollution in Cochin backwaters. *Mahasagar*, *Bull. Nat Inst. Oceanogr.*, 16: 199-207
- Remani, K.N., Venugopal, P., Sarala Devi, K. and Unnithan, R.V., 1980. Studies on the sediments of the Cochin backwaters in relation to pollution. *Indian J. Mar. Sci.*, 9: 111-114
- Remani, K.N., Venugopal, P., Sarala Devi, K. and Unnithan, R.V., 1981. Sediments of a retting yard. Indian J. Mar. Sci., 10: 41-45
- Rendell, P.S., Batly, G.E. and Cameron, A.J., 1980. Adsorption as a control of metal concentrations in sediment extracts. *Environ. Sci. Technol.*, 14: 314-318
- Reuber, B., Mackay, D., Paterson, S. and Stokes, P. 1987. A discussion of chemical equilibria and transport at the sediment water interface. *Environ. Toxicol. Chem.*, 6: 731-739
- Rosental, R., Eagle, G.A. and Orren, M.J., 1986. Trace metal distribution in different chemical fractions of nearshore marine sediments. *Estuar. Coast. Shelf Sci.*, 22: 303-324

Sadiq, M., 1989. Nickel sorption and speciation in a marine environment. *Hydrobiologia*, 176/177: 225-232

Salomons, W. and Bril, J., 1990. Adsorption and release processes in estuaries. In: Michaelis, W. (ed.), Estuarine Water Quality Management, Springer- Verlag, Berlin. pp 405-412

Salomons, W. and Eagle, A.M., 1990. Hydrology, sedimentology and the fate and distribution of copper in the mine-related discharges in the Fly River system, Papua New Guinea. Sci. Total Environ., 97/98: 315-334

Salomons, W. and Eysink, W.D., 1981. Pathways of mud and particulate trace metals from rivers to the southern North Sea. Spec. Publ.Int. Assoc. Sedimentol., 5: 429-450

Salomons, W. and Forstner, U., 1980. Trace metal analysis on polluted sediments. Part II Evaluation of environmental impact. Environ. Technol. Lett. 1: 506-517

Salomons, W. and Forstner, U., 1984. Trace metals in estuaries: field investigations. In : Salomons, W. and Forstner, U. (eds.), Metals in the Hydrocycle. Springer-Verlag, Berlin. pp 223-233

Salomons, W. and Mook, W.G., 1980. Biogeochemical processes affecting metal concentrations in lake sediments (IJsselmeer, The Netherlands). Sci. Total Environ., 16: 217-229

Samanidou, V. and Fytianos, K., 1987. Partitioning of heavy metals into selective chemical fractions in sediments from rivers in Northern Greece. Sci. Total Environ., 67: 279-285

Sankaranarayanan, V.N., Joseph, T., Jayalakshmy, K.V. and Balachandran, K.K, 1984. Atypical behaviour of dissolved silicate in the Cochin Backwater and Periyar River. Indian. J. Mar. Sci., 13: 60-63

Sankaranarayanan, V.N. and Punampunnayil, S.U., 1979. Studies on organic carbon, nitrogen and phosphorus in sediments of the Cochin backwater. Indian J. Mar. Sci., 8: 27-30

Sankaranarayanan, V.N., Purushan, K.S. and Rao, T.S.S., 1978. Concentration of some heavy metals in the oyster Crassostrea madrasensis (Preston) from the Cochin region. Indian J. Mar. Sci., 7: 130 - 131

- Sankaranarayanan, V.N. and Qasim, S.Z., 1969. Nutrients of the Cochin backwater in relation to environmental characteristics. Mar. Biol., 3: 236-247
- Sankaranarayanan, V.N. and Rosamma Stephen, 1978. Particulate iron, manganese, copper and zinc in waters of the Cochin backwater. Indian J. Mar. Sci., 7: 201-203
- Sankaranarayanan, V.N., Udaya Varma, P., Balachandran, K.K., Pylee, A. and Joseph, T., 1986. Estuarine characterstics of the lower reaches of the river Periyar (Cochin Backwater). Indian J. Mar. Sci., 15: 166-170
- Santschi, P. H., 1988. Factors controlling the biogeochemical cycles of trace elements in fresh and coastal marine waters as revealed by artificial radioisotopes. Limnol. Oceanogr. 33: 848-866.
- Sarala Devi, K., 1989. Temporal and spatial variations in particulate matter, particulate organic carbon and attenuation coefficient in Cochin backwaters. Indian. J. Mar. Sci., 18: 242-245
- Sarala Devi, K., Sankaranarayanan, V.N. and Venugopal, P., 1991. Distribution of nutrients in the Periyar river estuary. Indian J. Mar. Sci., 20: 49-54
- Sarala Devi, K., Venugopal, P., 1989. Benthos of Cochin backwaters receiving industrial effluents. Indian. J. Mar. Sci., 18: 165-169
- Sarala Devi, K., Venugopal, P., Remani, K.N., Lalitha, S. and Unnikrishnan, K.V., 1979. Hydrographic features and water quality off Cochin backwaters in relation to industrial pollution. Indian J. Mar. Sci., 8: 141-145
- Sathyanathan, B., Nair, S.M., Jacob Chacko, and Nambisan, P.N.K., 1988. Sublethal effects of copper and mercury on some biochemical constituents of the estuarine clam Villorita cyprinoides var. cochinensis (Hanley). Bull. Environ. Contam. Toxicol., 40: 510-516
- Satyanarayana, D., Rao, I.M. and Prasada Reddy, B.R., 1985a. Chemical oceanography of harbour and coastal environment of Visakhapatnam (Bay of Bengal). Part 1. Trace metals in water and particulate matter. Indian J. Mar. Sci., 14: 139-146.

- Satyanarayana, D., Prasada Reddy, B.R., Swamy, A.S.R. and Krishna Rao, G., 1985b. Chemical Oceanography of Harbour and Coastal environment of Visakhapatnam (Bay of Bengal). Part II. Nature and composition of sediments. Indian J. Mar. Sci., 14: 147-150
- Schirmer, M., 1990. Monitoring the bioavailability of heavy metals in relation to the sediment pollution in the Weser estuary (FRG). In: Michaelis, W. (ed.) Estuarine Water Quality Management. Springer-Verlag, Berlin. pp 377- 383
- Schoer, J., 1985. Iron-oxo-hydroxides and their significance to the behaviour of heavy metals in estuaries. In: *Heavy Metals in the Environment*. Proc. Int. Conf. Athens. pp 384-388
- Schwertmann, U., 1964. Differenzierung der Eisenoxide des Bodens durch photochemische Extraktion mit sauerer Ammoniumoxalat-Losung. Z. Pflanzenernahr. Dung. Bodenkde., 105: 194- 202 (quoted in Forstner, 1982)
- Seidemann, D.E., 1991. Metal pollution in sediments of Jamaica bay, New York, USA, an urban estuary. *Environ. Manage.*, 15: 73-82
- Seralathan, P., 1986. Use of textural (cm) pattern for identification of depositional processes and environments of sediments of the Cauvery delta. Bull. Dept. Mar. Sci. Univ. Cochin, 14: 17-26
- Sharp, J.H., Pennock, J.R., Church, T.M., Tramontano, J.M. and Cifuentes, L.A., 1984. The estuarine interactions of nutrients organics and metals: A case study in the Delaware Estuary. In: Kennedy, V.S. (ed.), Estuary as Filter. Academic Press, New York. pp 241-258
- Shepard, F.P., 1954. Nomenclature based on sand-silt-clay ratios. J. Sediment. Petrol., 24: 151-158
- Shibu, M.P., Balchand, A.N. and Nambisan, P.N.K., 1990. Trace metal speciation in a tropical estuary: significance of environmental factors. Sci. Total Environ., 97/98: 267-287
- Sholkovitz, E.R., 1976. Flocculation of dissolved organic and inorganic matter during mixing of river water and sea water. Geochim. Cosmochim. Acta, 40: 831-845
- Sholkovitz, E.R., 1978. The flocculation of dissolved Fe, Mn, Cu, Ni, Co and Cd during estuarine mixing. Earth Planet Sci. Lett., 41: 77- 86

- Sholkovitz, E.R. and Copland, D., 1981. The coagulation, solubility and adsorption properties of Fe, Mn, Cu, Ni, Cd, Co and humic acids in a river water. Geochim. Cosmochim. Acta, 45: 181-189
- Shynamma, C.S. and Balakrishnan, K.P., 1973. Diurnal variations in some physico-chemical factors in Cochin backwater during south west monsoon. J. Mar. Biol. Assoc. India, 15: 391-399
- Sinex, S.A., Cantillo, A.L. and Helz, H.R., 1980. Accuracy of acid extraction methods for trace metals in sediments. *Anal. Chem.*, 52: 2342-2346
- Sinex, S.A. and Wright, D.A., 1988. Distribution of trace metals in the sediments and biota of Chesapeake Bay. Mar. Pollut. Bull., 19: 425- 431
- Singh, S.K. and Subramanian, V., 1984. Hydrous Fe and Mn oxides - scavengers of heavy metals in the aquatic environment. CRC Rev. Environ. Control, 4: 33-90
- Sivadasan, C.R., 1987. Studies on the effects of selected trace metals on Metapenaeus dobsoni (Miers). Ph.D. Thesis. Cochin University of Science and Technology, Cochin.
- Sivadasan, C.R. and Nambisan, P.N.K., 1988. Seasonal variation of mercury, copper and zinc in the prawn Metapenaeus dobsoni (Miers) from Cochin backwaters. Mar. Pollut. Bull., 19: 579-580
- Sivadasan, C.R., Nambisan, P.N.K. and Damodaran, R., 1986. Toxicity of mercury, copper and zinc to the prawn Metapenaeus dobsoni (Miers). Curr. Sci., 55: 337-340
- Slauenwhite, D.E. and Wangersky, P.J., 1991. Behaviour of copper and cadmium during a phytoplankton bloom : a mesocosm experiment. *Mar. Chem.*, 32: 37-50
- Smith, K.A., 1990. Manganese and Cobalt. In: Alloway, B.J. (ed.), Heavy Metals in soils. Blackie and Son, London. pp 197-221
- Smith, R. M. and Martell, A.E. 1976. Critical stability constants. Vol. 4. Inorganic complexes. Plenum Press, New York. 257p
- Snedecor, G.W. and Cochran, W.G., 1968. Statistical Methods. Oxford and IBH Publ. Co., New Delhi. 593p.

- Soong, K.L., 1974. Versuche zur adsorptiven Bindung von Schwer- metall-Ionen an kunstlichen Tongemischen. Unpubl. Diss. Univ. Heidelberg. (cited in Forstner, 1979)
- Steneker, R.C.H., Van der Sloot, H.A. and Das, H.A., 1988. Leaching studies on dredged material in oxidised and reduced state. Sci. Total Environ., 68: 11- 23
- Strickland, J.D.H. and Parsons, T.R., 1977. A Practical hand book of seawater analysis. Fish. Res. Bd. Can. Bull., (No. 125), Ottawa. 203p.
- Stumm, W. and Morgan, J.J., 1970. Aquatic Chemistry. Wiley Interscience, New York. 583p.
- Subramanian, A.N., Ramdhas, V. and Venugopalan, V.K., 1979. Iron precipitation in estuarine environment. Indian J. Mar. Sci., 8: 175-176
- Subramanian, A.N., Subramanian, B.R. and Venugopalan, V.K., 1981. Seasonal variations of iron in plankton and mangrove leaves in relation to environmental concentration. J. Environ. Biol., 2: 29-40
- Subramanian, V., 1987. Environmental geochemistry of Indian river basins : a review. J. Geol. Soc. India., 29: 205-220
- Subramanian, V., Jha, P.K. and Van Grieken, R., 1988. Heavy metals in the Ganges Estuary. Mar. Pollut. Bull., 19: 290- 293
- Subramanian, V., Al. Ramanathan and Vaithyanathan, P., 1989. Distribution and fractionation of heavy metals in the Cauvery estuary, India. Mar. Pollut. Bull., 20: 286-290
- Subramanian, V., Van'T Dack, L., and Van Grieken, R., 1985. Chemical composition of the river sediments from the Indian sub- continent. *Chem. Geol.*, **48**: 271-279
- Sundaresan, J., 1991. Textural distribution of surficial sed ents of the Cochin harbor. Indian J. Mar. Sci., 20: 127 - 129
- Suresh, K. and Mohandas, A., 1987. Hemolymp lactic acid levels in Sunetta scripta exposed to sublethal concentrations of mercury and copper. J. Invert. Pathol., 49: 265-272.
- Teraoka, H. and Nakashima, S., 1990. Mechanissm of enrichment of trace metals on fine sludges collected from filtration plants. *Environ. Geol. Water Sci.*, 16: 143-148

,

- Tessier, A. and Campbell, P.G.C., 1987. Partitioning of trace metals in sediments: Relationships with bioavailability. *Hydrobiologia*, 149: 43-52
- Tessier, A. and Campbell, P.G.C., 1991. Comment on "Pitfalls of sequential extractions" by P.M.V. Nirel and F.M.M. Morel, Wat. Res., 25: 115- 117
- Tessier, A., Campbell, P.G.C., Auclair, J.C. and Bisson, M., 1984. Relationships between the partitioning of trace metals in sediments and their accumulation in the tissues of the freshwater mollusc Elliptio complanata in a mining area. Can. J. Fish. Aquat. Sci., 41: 1463-1472
- Tessier, A., Campbell, P.G.C. and Bisson, M., 1979. Sequential extraction procedures for the speciation of particulate trace metals. *Anal. Chem.*, 51: 844 - 851
- Tessier, A., Campbell, P.G.C. and Bisson, M., 1980. Trace metal speciation in the Yamaska and St. Francois rivers (Quebec). Can. J. Earth Sci., 17: 90-105
- Tessier, A., Rapin, F., and Carignan, R., 1985. Trace metals in oxic lake sediments: possible adsorption onto iron oxyhydroxides, *Geochim. Cosmochim. Acta*, **49**: 183-186
- Thiessen, L.M., Hallez, S., Lenelle, Y and Verduyn, G., 1988. Evaluation of the total Cr-levels in the ambient air in Belgium. Sci. Total Environ., 71: 519-526
- Thompson, E.A., Luoma, S.N., Cain, D.J. and Johnansson, C., 1980. The effects of sample storage on the extraction of Cu, Zn, Fe, Mn and organic material from oxidised estuarine sediments. Water Air Soil Pollut., 14: 215-233
- Thorne, L.T. and Nickless, G., 1981. The relation between heavy metals and particle size fractions within the Severn estuary (U.K.) inter-tidal sediments. *Sci. Total Environ.*, 19: 207-213
- Tiller, K.G., Merry, R.H., Zarcinas, B.A. and Ward, T.J., 1989. Regional geochemistry of metal-contaminated surficial sediments and seagrasses in Upper Spencer Gulf, South Australia. Estuar. Coast. Shelf Sci., 28: 473-493
- Tipping, E., Hetherington, N.B., Hilton, J., Thompson, D.W., Bowles, E. and Taylor, J.H., 1985. Artifacts in the use of selective chemical extraction to determine distributions of metals between oxides of manganese and iron. Anal. Chem., 57: 1944-1946

- Trefry, J.H. and Metz, S., 1984. Selective leaching of trace metals from sediments are a function of pH. Anal. Chem., 56: 745- 749
- Trefry, J.H. and Presley, B.J., 1976. Heavy metals in sediments from San Antonio Bay and the Northwest Gulf of Mexico. *Environ. Geol.*, 1: 283-294
- Trefry, J.H. and Presley, B.J., 1982. Manganese fluxes from Mississippi delta sediments. Geochim. Cosmochim. Acta, 46: 1715-1726
- Turner, A., Millward, G.E., Karbe, L. and Dembinski, M., 1990. Intercomparison of particulate trace metals from four large estuaries. In: Michaelis, W. (ed.), Estuarine Water Quarity Management. Springer-Verlag, Berlin. pp 455-460
- Udaya Varma, P., Pylee, A. and Rama Raju, V.S., 1981. Tidal influence on the seasonal variation in current and salinity around Willingdon island. *Mahasagar*, 14: 225-237
- Udaya Varma, P., Abraham Pylee and Sankaranarayanan, V.N., 1987. Dynamic circulation and salt balance in the upper reaches of Periyar estuary. *Proc. Natn. Sem. Estuarine Management*, Trivan drum. pp 32-37
- Ui, J., 1972. A few coastal pollution problems in Japan. In: Dyrssen and Tanger, D. (eds.), The Changing Chemistry of the Oceans, Almkoist and Wiksell, Stockholm. pp 171-176
- Uncles, R.J., Stephens, J.A. and Woodrow, T.Y., 1988. Seasonal cycling of estuarine sediment and contaminant transport. *Estuaries*, 11: 108-116
- Unnithan, R.V., Vijayan, M., Radhakrishnan, E.V. and Ramani, K.N., 1977. Incidence of fish mortality from industrial pollution in Cochin backwaters. Indian. J. Mar. Sci., 6: 81-83
- Ure, A.M., 1990. Methods of analysis for heavy metals in soils. In: Alloway, B.J. (ed.), Heavy Metals in Soils. Blackie and Sons, London. pp 40- 80
- Vale, C., 1990. Temporal variations of particulate metals in the Tagus river estuary. Sci. Total Environ., 97/98: 137-154
- Van den Berg, C.M.G., Merks, A.G.A. and Duursma, E.K., 1987. Organic complexation and its control of the dissolved concentra tions of copper and zinc in the Scheldt estuary. *Estuar. Coast. Shelf Sci.*, 24: 785-797

- Van der Weijden, C.H., Arnoldus, M.J.H.L. and Meurs, C.J., 1977. Desorption of metals from suspended material in the Rhine estuary. Neth. J. Sea. Res., 11: 130-145
- Van Leussen, W. and Dronkers, J., 1990. Progress in estuarine water quality modelling. In: Michaelis, W. (ed.), Estuarine Water Quality Management. Springer-Verlag, Berlin. pp 13- 24
- Van Valin, R. and Morse, J.W., 1982. An investigation of methods commonly used for the selective removal of trace metals in sediments. *Mar. Chem.*, 11: 535-564
- Veerayya, M. and Murty, P.S.N., 1974. Studies on the sediments of Vembanad lake, Kerala State. Part III Distribution and interpretation of bottom sediments. Indian J. Mar. Sci., 3: 16-27
- Venugopal, P., Ramani, K.N., Sarala Devi, K and Unnithan, R.V., 1980. Fish kill in Chitrapuzha. Seafood Export J. 12: 17- 20
- Venugopal, P., Saraladevi, K., Remani, K.N. and Unnithan, R.V., 1982. Trace metal levels in the sediments of the Cochin backwaters. Mahasagar, Bull. Natn. Inst. Oceanogr., 15: 205- 214
- Vijayan, M., Remani, K.N., and Unnithan, R.V., 1976. Effect of organic pollution on some hydrographic features of Cochin back waters. Indian. J. Mar. Sci., 5: 196-200
- Volkov, I.I. and Fomina, L.S., 1974. Influence of organic materials and processes of sulphide formation on distribution of some trace elements in deepwater sediments of Black Sea. In Degens, E.T. and Ross, D.A. (eds.), The Black Sea: Geology, Chemistry and Biology. Am. Assoc. Petrol. Geol. Mem., 20: 456-476
- Vollmer, M., Hudec, B., Knauth, H.D. and Schwedhelm, E., 1990. Seasonal variations of mineralogical and heavy metal contamination of suspended matter in the river Elbe estuary. In. Michaelis, W. (ed.), Estuarine Water Quality Management. Springer-Verlag, Berlin. pp 153-158
- Waldichuk, M., 1985. Biological availability of metals to marine organisms. *Mar. Pollut. Bull.*, 16: 7-11
- Walkley, A. and Black, I. A., 1934. An examination of the Degtjareff method for determining soil organic matter and a proposed modification of the chromic acid titration method. Soil. Sci., 37: 29-38

- Wangersky, P.J., 1986. Biological control of trace metal residence time and speciation : a review and synthesis. *Mar. Chem.*, 18: 269-297
- Wells, M.L. and Mayer, L.M., 1991. Variations in the chemical lability of iron in estuarine, coastal and shelf waters and its implications for phytoplankton. *Mar. Chem.*, 32: 195-210
- WHO, 1982. Micropollutants in River Sediments. Euro reports and studies, No. 61, WHO, Copenhagen. 85p.
- Wilke, R.J. and Dayal, R., 1982. The behaviour of iron, manganese and silicon in the Peconic river estuary, New York. Estuar. Coast. Shelf Sci., 15: 577-586
- Williams, S.C., Simpson, H.J, Olsen, C.R., and Bopp, R.F., 1978. Sources of heavy metals in sediments of the Hudson river estuary. Mar. Chem., 6: 195-213
- Windom, H.L., 1975. Heavy metal fluxes through salt- marsh estuaries. In: Cronin, L.E. (ed.), *Estuarine Research*, Vol. I. Academic Press, New York. pp 137-152
- Windom, H., Smith. R.(Jr.), Rawlinson, C., Hungspreugs, M., Dharmvanij, S. and Wattayakorn, G., 1988. Trace metal transport in a tropical estuary. Mar. Chem., 24: 293-309
- Windom, H., Byrd, J., Smith, R., Hungspreugs, M., Dharmvanij, S., Thumtrakul, W. and Yeats, P., 1991. Trace metal nutrient relationships in estuaries. Mar. Chem., 32: 177-194
- Windom, H., Schropp, S.J., Calder, F.D., Ryan, J.D., Smith, R.G.(Jr.), Burney, L.C., Lewis, F.G. and Rawlinson, C.H., 1989. Natural trace metal concentrations in estuarine and coastal marine sediments of the south eastern United States. Environ. Sci. Technol., 23: 314-320
- Windom, H. and Smith, R.G., 1985. The geochemistry of lead in rivers, estuaries and the continental shelf of the southern United States. Mar. Chem. 17: 43-56
- Windom, H., Wallace, G., Smith, R., Dudek, N., Dulmage, R. and Storti, F., 1983. Behavior of copper in southeastern United States estuaries. Mar. Chem., 12: 183-193
- Wittmann, G.T.W., 1979. Toxic metals. In: Forstner, U. and Wittman, G.T.M. (eds.), Metal Pollution in the Aquatic Environment. Springer-Verlag, Berlin. pp 3-70

- Wolfe, D.A. and Kjerfve, B., 1986. Estuarine variability : A overview. In: Wolfe, D.A.(ed.), *Estuarine variability*. Academic Press. Inc. New York. pp 3-17
- Wollast, R., Billen, G. and Duinker, J.C., 1979. Manganese in the Rhine and Scheldt Estuaries. 1 Physico-chemical behaviour. Estuar. Coast. Mar. Sci., 9: 161-169.
- Wright, D.A, and Phillips, D.J.H., 1988. Chesapeake and San Francisco Bays: a study in contrasts and parallels. Mar. Pollut. Bull., 19: 405-413
- Yeats, P.A., Bewers, J.M. and Walton, A., 1978. Sensitivity of coastal waters to anthropogenic trace metal emissions. *Mar. Pollut. Bull.*, 9: 264 - 268
- Yeats, P.A., Sundby, B. and Bewers, J.M., 1979. Manganese cycling in coastal waters. *Mar. Chem.*, 8: 43-55
- Yen, T.F. and Tang, J.I.S. 1977. Chemical aspects of marine sediments. In: Yen, T.F. (ed.), Chemistry of Marine Sediments. Ann Arbor Sci.Publ. Inc. Michigan. pp 1-38
- Yetang, H. and Forstner, U., 1984. Chemical forms of some heavy metals in Huang He river sediments (China) and comparison with data from Rhine river sediments (West Germany). *Geochemistry*, 3: 37-44
- Yoshimura, H., Shiba, T. and Watanabe, K., 1988. Identification of pollution sources of heavy metals in marine coastal sediments from patterns of partitioning and diagenetic change. In: Wolfe, D.A. and O'Connor, T.P. (eds.), Urban Wastes in Coastal Marine Environments. Robert E. Krieger Publishing Company, Malabar, Florida. pp 235-243
- Zhang, J., Huang, W.W., and Wang, Q., 1990. Concentration and partitioning of particulate trace metals in the Changjang Yangtze river, China. Water Air Soil Pollut., 52: 57-70
- Zingde, M.A., Rokade, M.A.M., Mandalia, A.V., 1988. Heavy metals in Mindhola river estuary, India. Mar. Pollut. Bull., 19: 538- 540
- Zwolsman, J.J.G. and Van Eck, G.T.M., 1990. The behaviour of dissolved Cd, Cu and Zn in the Scheldt estuary. In: Michaelis, W. (ed.), Estuarine Water Quality Management. Springer-Verlag, Berlin. pp 413- 420